

# The London Gazette

## Published by Authority.

Registered as a newspaper

For Table of Contents see last page

TUESDAY, 5 JULY, 1938

Privy Council Office, 5th July, 1938. UNIVERSITIES OF OXFORD AND CAMBRIDGE ACT, 1923.

A STATUTE made by the Governing Body of Christ Church, in the University of Oxford, on the 25th June, 1938, amending Statute XIII of the College Statutes, has been sub-mitted for the approval of His Majesty in Council, and notice of its having been so submitted is published in accordance with the provisions of " The Universities of Oxford and Cambridge Act, 1923.''

#### Privy Council Office, 5th July, 1938.

#### LOCAL GOVERNMENT ACT, 1933.

Notice is hereby given that a Scheme making provision for the transfer or adjustment of the whole or part of the functions, franchises, property, income, debts, liabilities and expenses of the Urban District Council of Kettering, in the Administrative County of Northampton, and for other purposes, has been settled by a Committee of His Majesty's Most Honourable Privy Council.

Notice is hereby further given that a copy of the said Scheme may be inspected at the Office of the Urban District Council, Huxloe Place, Kettering.

#### Lord Chamberlain's Office, St. James's Palace, S.W.I. · 2nd July, 1938.

The KING has been graciously pleased to appoint the Right Honourable Rowland Thomas, Earl of Cromer, G.C.B., G.C.I.E., G.C.V.O., to be a Permanent Lord in Waiting to His Majesty.

Foreign Office, April 1, 1938.

The KING has been graciously pleased to appoint:-

Thomas James Erskine Haskoll, Esquire, to be His Majesty's Vice-Consul at Marseilles; and

Patrick Murdoch Johnston, Esquire, to be His Majesty's Vice-Consul at Valparaiso.

#### Foreign Office,

April 5, 1938.

The KING has been graciously pleased to appoint:-

John Percival Summerscale, Esquire, to be Commercial Secretary, Second Grade, at His Majesty's Embassy at Washington.

Foreign Office, April 12, 1938. The KING has been graciously pleased to appoint:-

John Eric Maclean Carvell, Esquire, to be His Majesty's Consul-General for Bavaria, with the exception of the Administrative District of Lower Franconia, to reside at Munich.

#### Foreign Office

April 15, 1938.

The KING has been graciously pleased to appoint:-

Sir Norman King, K.C.M.G., to be His Majesty's Consul-General for the Departments of Bouches du Rhône, Pyrénées Orientales, Aude, Hérault, Gard, Var, Vaucluse, Haute Loire, Ardèche, Drôme, Hautes Alpes, Basses Alpes, Lozère, Aveyron and Tarn, to reside at Marseilles.

Foreign Office, April 23, 1938.

The KING has been graciously pleased to appoint:

- Charles William Orde, Esquire, C.M.G., to be His Majesty's Envoy Extraordinary and Minister Plenipotentiary to the Republics of Estonia, Latvia and Lithuania;
- Charles William Baxter, Esquire, M.C., to be an Acting Counsellor of Embassy in His Majesty's Diplomatic Service; and
- Christopher Bramwell, Esquire, to be a First Secretary in His Majesty's Diplomatic Service.

# Foreign Office, April 24, 1938.

The KING has been graciously pleased to appoint:-

Lieutenant-Colonel Geoffrey Lawrence Betham, C.I.E., M.C., to be His Majesty's Envoy Extraordinary and Minister Plenipotentiary in Nepal.

#### Foreign Office, April 27, 1938.

The KING has been graciously pleased to appoint:

Robert Heatlie Scott, Esquire, to be Com-mercial Secretary, Second Grade, at His Majesty's Embassy in China, with effect as from 27th January, 1938.

#### Foreign Office,

May 6, 1938.

The KING has been pleased to approve of :-

- Mr. Jacob Olsen, as Consul of Norway at Tel Aviv for Palestine;
- Mr. Paul Wisness, as Consul of Norway at Newcastle-on-Tyne for Northumberland (with the exception of Amble with Warkworth, Berwick-on-Tweed, and Blyth) Durham (with the exception of West and East Hartlepool, Sunderland and Seaham Harbour), Cumberland and Westmorland;
- Señor Don Alfonso Behrens De Lima, as Consul of Panama at Port of Spain, Trinidad;
- Señor Don Clemente Cerdeira y Fernández, as Consul of Spain at Liverpool, for Liverpool and the counties of Lancashire, Carnarvon, Denbigh, Flint, Merioneth, Montgomery, Derby and Chester, and the isles of Anglesey and Man;
- Señor Don Jesús de Miguel y Lancha, as Consul of Spain at Gibraltar;
- Monsieur Albert Leiber, as Consul of Switzerland at Colombo, with jurisdiction in the island of Ceylon;
- Mr. William E. Chapman, as Consul of the United States of America at Gibraltar; and
- Mr W. Collins, as Vice-Consul of Greece at Immingham.

Foreign Office, May 28, 1938. The KING has been graciously pleased to appoint:

George William Rendel, Esquire, C.M.G., to be His Majesty's Envoy Extraordinary and Minister Plenipotentiary to His Majesty the King of the Bulgarians.

#### Foreign Office,

May 305 1938.

The KING has been pleased to approve of:

- Señor Don Juan B. de Lemoine, as Consul-General of the Argentine Republic at Cape Town, for Southern Rhodesia, Mauritius and Northern Rhodesia;
- Mr. H. M. J. Fein, as Consul-General of the
- Netherlands at Singapore; Mr. A. Merens, as Consul-General of the Netherlands at Calcutta, with jurisdiction also in the island of Ceylon;
- Mr. R. Gellecher, as Consul-General of Turkey at Malta;
- Mr. J. C. Burgess, as Consul of Belgium at Swansea, for the Western part of the county of Glamorgan bounded by the railway line between Briton Ferry and Colbren Junction and the Eastern part of the county of Car-marthen bounded by the railway between Ferryside and Llandyssil;
- Mr. Joseph William Allan, as Consul of Denmark at St. John's, Newfoundland;
- Monsieur Louis François Antoine Lucas, as Consul of France at Haifa, with jurisdiction
- in Tulkarem, Haifa and Acre; Monsieur René Victor Neuville, as Consul of France at Gibraltar;
- Mr. Robert Douglas Stein, as Consul of Ger-many at Port Louis, for the island of Mauritius;
- Mr. Frederick William Harris, as Consul of Greece at Kingston, Jamaica; Don Ugo dei Marchesi Theodoli, as Consul of
- Italy at Nicosia, Cyprus;
- Mr. Toyoji Kaneko, as Consul of Japan at Rangoon;
- Mr. Harold Banks Fearnley, as Consul of Norway at Bradford, for that City;
- Senhor José Peixoto Ponces de Carvalho, as Consul of Portugal at Salisbury, Southern Rhodesia;
- Señor Don Juan Antonio Careaga Andueza, as Consul of Spain at Newcastle for the counties of Northumberland, Westmorland, Durham, Lincolnshire, Nottinghamshire Yorkshire, and Cumberland;
- Mr. Maurice Henry Flash, as Vice-Consul of the Argentine Republic at Cape Town, with jurisdiction in the island of Mauritius;
- Señor Don Hans Schweizer, as Vice-Consul of the Argentine Republic at Singapore, for the Straits Settlements; and
- Mr. Robert Hargreaves, as Vice-Consul of Greece at Sheffield.

#### Foreign Office,

June 18, 1938.

The KING has been pleased to approve of:-

- onsieur F. Jansen, as Consul-General of Belgium at Sydney, with jurisdiction in the Monsieur F. British Possessions in the Pacific situated to the west of 180° longitude (with the exception of Fiji) but including the New Hebrides;
- Doctor Herbert von Dirksen, as Consul-General of Germany in London for the United Kingdom of Great Britain and Northern Ireland;
- Mr. R. N. Roy, as Consul of the Dominican Republic at Calcutta;

- Mr. W. E. Passells, as Consul of the Netherlands at Freetown for Sierra Leone and Gambia:
- Mr. Olaf Kristian Olsen, as Consul of Norway
- at St. John's, Newfoundland; r. Joseph William Allan, as Consul of Portugal at St. John's, Newfoundland; Mr.
- Senhor Felix Borges Medeiros da Horta, as Consul of Portugal at Liverpool; and
- Señor Don Luis Ăntonio Gallardo, as Consul of Salvador at Liverpool.

#### Whitehall, July 1, 1938.

The KING has been pleased, by Letters Patent under the Great Seal of the Realm, bearing date the 28th ultimo, to confer the dignity of a Barony of the United Kingdom upon Sir Josiah Charles Stamp, G.C.B., G.B.E., and the heirs male of his body lawfully begotten, by the name, style and title of BARON STAMP, of Shortlands in the County of Kent.

#### Whitehall, July 1, 1938.

The KING has been pleased, by Letters Patent under the Great Seal of the Realm, bearing date the 29th ultimo, to confer the dignity of a Barony of the United Kingdom upon Vivian Hugh Smith, Esq., and the heirs male of his body lawfully begotten, by the name, style and title of BARON BICESTER, of Tusmore in the County of Oxford.

#### Factory Department, Home Office, 29th June, 1938.

The Chief Inspector of Factories has appointed Dr. H. Broadbent to be Examining Surgeon under the Factories Act for the Bridlington District of the County of York (East Riding).

#### Factory Department, Home Office, 29th June, 1938.

The Chief Inspector appointed Dr. J. D. of Factories has Robertson be to Examining Surgeon under the Factories Act for the Barnoldswick District of the County of York (West Riding).

#### Factory Department, Home Office, 29th June, 1938.

The Chief Inspector of Factories has appointed Dr. R. F. Winckworth to be Examining Surgeon under the Factories Act for the Taunton District of the County of Somerset.

#### Factory Department, Home Office, 30th June, 1938.

Chief Inspector of Factories The has appointed Dr. C. R. McClure to be Examining Surgeon under the Factories Act for the Hampstead District of the County of London.

#### FACTORIES ACT, 1937.

In pursuance of Section I of the Rules Publication Act, 1893, notice is hereby given that the Secretary of State for the Home Department proposes, at the expiration of 40. days from the publication of this notice, to make regulations under Subsection (9) of Section 73 of the Factories Act, 1937, entitled the Factory (Individual Overtime) Regulations, They would enable the occupier of a 1938. factory in which is carried on the manufacture of a variety of light metal articles or of fittings or accessories composed wholly or largely of metal to adopt the provisions of the Regulations allowing overtime employment to be calculated by reference to the individual, in substitution for the provisions of the Section limiting overtime employment by reference to the factory.

Copies of the Regulations may be purchased directly from H.M. Stationery Office at the following addresses:—Adastral House, Kings-way, London, W.C.2; 120, George Street, Edinburgh 2; 26, York Street, Manchester 1; I, St. Andrew's Crescent, Cardiff; or through any bookseller.

Whitehall,

1st July, 1938.

#### FACTORIES ACT, 1937.

The Secretary of State for the Home Department hereby gives notice that on the 30th June, 1938, he made Regulations under the Factories Act, 1937, as indicated below.

Copies of the Regulations may be purchased directly from H.M. Stationery Office at the following addresses:—Adastral House, Kings-way, London, W.C.2; 120, George Street, Edinburgh 2; 26, York Street, Manchester 1; 1, St. Andrew's Crescent, Cardiff; or through any bookseller.

**I.** Factory Overtime (Separation of Different Parts or Sets) Regulations, 1938, prescribing under Section 73 (8) of the Act, conditions subject to which employment in different parts of a factory, or the employment of different sets of persons in different processes, may be treated, for the purpose of reckoning overtime, as if it were employment in different factories.

2. Operations at Unfenced Machinery Regulations, 1938, specifying under Sections 15 and 16 of the Act conditions under which certain operations may be carried out at unfenced These Regulations will come into machinery. force on the 1st August, 1938.

Whitehall,

1st July, 1938.

#### FACTORIES ACT, 1937.

The Secretary of State for the Home Department hereby gives notice that on the 30th June, 1938, he made an Order under Subsection (2) of Section 160 of the Factories Act, 1937, postponing from the 1st July, 1938, until the 1st January, 1940, the coming into operation of certain requirements of the Act in certain cases, namely:-

1. The requirement contained in Subsection (2) of Section 13 of the said Act as respects factories in which the main transmission machinery is driven wholly or partly by water power.

2. The requirements contained in Subsections (3), (4), (5) and (7) of Section 22 of the Act as respects hoists or lifts constructed or reconstructed before the passing of the Act in warehouses to which the provisions of Section 22 are applied by Subsection (3) of Section 105 of the Act, subject to the condition that the hoist or lift shall be securely fenced so far as is reasonably practicable.

Copies of this Order which may be cited as the Factories Act (Postponement of Certain Requirements, Order, 1938, can be purchased directly from H.M. Stationery Office at the following addresses:—Adastral House, Kingsway, W.C.2; 120, George Street, Edinburgh 2; 26, York Street, Manchester 1; 1, St. Andrew's Crescent, Cardiff; or through any bookseller.

Whitehall,

1st July, 1938.

Board of Trade, Mercantile Marine Department, Great George Street, S.W.1. 4th July, 1938.

The Board of Trade hereby give notice under the General Rules for Formal Investigations into Shipping Casualties and Appeals and Rehearings, 1923, that they have received during the month of June, 1938, Reports of Formal Investigations, held by Courts in His Majesty's dominions, into the circumstances attending casualties to the following vessels:—

I. S.S. "Chyebassa" and S.S. "Nord." 2. I.S.V. "Travancore" and a country boat.

3. I.S.V. " Angus " and " Hopper Barge No. 8."

#### LAND TAX COMMISSIONERS' NAMES.

The following additions and corrections are made to the Schedule of names of Land Tax Commissioners published in the London Gazette dated Friday, 22nd April, 1938.

Page 2629, first column, after line 40 insert: For the rest of the County of Bucks.

Walter Gray, "Westleigh", Edlesborough, George Henry Southall, "The Tollgate", Wing.

Page 2637, second column, at end add:

Harry Hutton Ashton, Barrow Road, Barton on Humber, Fred Hopper, Mount House, Barton on Humber, Dr. Thomas Hobson Kirk, Burgate, Barton on Humber, Alfred Henry Thompson, Fleetgate, Barton on Humber, Robert James Blakeborough, Habrough, Digby Binney Sowerby, Kirmington, Charles Bratton Riggall, Burnham Manor, Thornton Curtis, John Harrison Davey, Thornton Curtis, Tom Francis Raby, Goxhill, Thomas Till, Barrow on Humber, Gerald Tinley, South Ferriby, William Smith, Brigg, Arthur Henry Page, Brigg, Edmund Addison, Keelby, Arthur John Jackson, Caistor, John Elliot Lamb, Caistor, Ralph Godfrey, Melton Ross, John de Blaquiere Tindall Lucas, Brocklesby, Habrough, Henry Anningson Marshall, Saxby Grange, Frederick John Ellis Hare, Bonby, The Rev. Archibald George Edward Smith, Wootton.

Page 2640, first column, line 36, for Steel read Shiel.

Page 2649, first column, line 51, for Carmarthen read Carnarvon.

#### COAL MINES ACT, 1911.

The Secretary for Mines gives notice that on the 29th day of June, 1938, the Board of Trade made General Regulations for Quarries under the powers conferred by Section 86 of the Coal Mines Act, 1911, as applied to quarries by the Factories Act, 1937.

The Regulations, which are operative from 1st July, 1938, are divided into six parts.

Part I. requires notice to be given to His Majesty's Inspectors of the opening and closing of quarries, lays down general requirements for the safe working of quarries, and defines in general terms the respective duties of the Owner, the Agent or Manager and the persons employed.

Part II. deals with health and welfare, and includes requirements with regard to first-aid; sanitary, messing and changing accommodation; and the suppression of dust which may be injurious to health.

Part III. deals with the employment of women and of young persons under sixteen years of age, and, as required by Section 158 (3) of the Factories Act, 1937, applies without change the provisions of the Coal Mines Act, 1911, relative to the employment of women and young persons above ground.

Part IV. regulates the use of machinery and plant and contains provisions about the cleaning, oiling and fencing of machinery, the use of cranes and other lifting machines, the care and maintenance of chains, ropes, etc., and the use of steam boilers and compressed air receivers.

Part V. is a code of rules, substantially the same as those already applying at coal mines and factories, to provide for the safe working of railway lines and sidings of standard gauge.

Part VI. prescribes the date on which the Regulations are to come into force and directs that they shall be in addition to and not in diminution of the provisions of Special Rules.

Copies of the Regulations, which may be cited as the Quarries General Regulations, 1938, may be purchased (price 4d. net) through any bookseller, or directly from His Majesty's Stationery Office.

Mines Department, Dean Stanley Street, Millbank, S.W.1. 1st July, 1938.

¢

#### THE METALLIFEROUS MINES GENERAL REGULATIONS, 1938.

#### FIRST-AID.

The Secretary for Mines gives notice that on the 30th day of June, 1938, the Board of Trade made an Order under No. 16 of the Metalliferous Mines General Regulations, 1938. The Order, which comes into force on 1st July, 1938, prescribes the minimum contents of the first-aid boxes or cupboards required to be provided and maintained at the surface of metalliferous mines.

Copies of the Order, which may be cited as the Metalliferous Mines (First-Aid) Order, 1938, may be purchased (price rd. net.) through any bookseller, or directly from His Majesty's Stationery Office.

Mines Department, Dean Stanley Street, Millbank, S.W.I. Ist July, 1938.

## THE QUARRIES GENERAL REGULATIONS, 1938.

#### FIRST-AID.

The Secretary for Mines gives notice that on the 29th day of June, 1938, the Board of Trade made an Order under No. 6 of the Quarries General Regulations, 1938.

The Order, which comes into force on 1st July, 1938, prescribes the minimum contents of the first-aid boxes or cupboards required to be provided and maintained at quarries.

Copies of the Order, which may be cited as the Quarries (First-Aid) Order, 1938, may be purchased (price 1d. net) through any bookseller, or directly from His Majesty's Stationery Office.

Mines Department, Dean Stanley Street, Millbank, S.W.I. 1st July, 1938.

#### COAL MINES ACT, 1911.

The Secretary for Mines gives notice that on the 30th day of June, 1938, the Board of Trade made General Regulations for Metalliferous Mines under the powers conferred by Section 86 of the Coal Mines Act, 1911, as applied to metalliferous mines by the Mining Industry Act, 1920.

The Regulations, which are in six parts, are operative from 1st July, 1938, except as otherwise provided.

Part I contains general safety provisions covering shafts and outlets, duties of winding enginemen and banksmen, ventilation, water dangers, travelling roads and working places, haulage, safety lamps and inflammable and noxious gases, fire precautions, notice of working of the mine, and the duties of Owners, Agents, Officials and persons employed. The use of explosives, and lines and sidings of standard gauge, are left to be dealt with by special rules, as hitherto.

Part II deals with health and welfare, and includes requirements with regard to first-aid; sanitary, messing and changing accommodation; lighting (at shafts and on the surface); and the suppression of dust which may be injurious to health.

Part III deals with the employment of women and young persons under sixteen years of age, and as required by Section 158 (3) of the Factories Act, 1937, applies without change the provisions of the Coal Mines Act, 1911, relative to the employment of women and young persons above ground. Part IV regulates the use of machinery and plant and contains provisions about the cleaning, oiling and fencing of machinery, the use of internal combustion engines below ground, the use of cranes and other lifting machines, the care and maintenance of chains and ropes, and the use of steam boilers, compressed air receivers and winding apparatus.

Part V deals with the use of electricity above and below ground and will be in substitution for the special rules, which apply mainly below ground, and for the Electricity Regulations under the Factory Acts, which at present apply above ground. The new requirements relating to the construction of cables and other electrical apparatus will not, in general, apply below ground until 1st July, 1941, and above ground until 1st July, 1945, to any cable or apparatus which was lawfully in use before 1st July, 1938.

Ist July, 1938. Part VI prescribes the date on which the Regulations are to come into force and directs that they shall be in addition to and not in diminution of the provisions of Special Rules. Copies of the Regulations, which may be

Copies of the Regulations, which may be cited as the Metalliferous Mines General Regulations, 1938, may be purchased (price 10d. net) through any bookseller, or directly from His Majesty's Stationery Office.

Mines Department,

Dean Stanley Street, Millbank, S.W.I. 1st July, 1938.

Amendments of Scheme approved on 30th June, 1938.

#### COAL MINES ACT, 1930.

#### THE WARWICKSHIRE DISTRICT (COAL MINES) SCHEME, 1930.

Whereas in pursuance of Clause 22 of the Warwickshire District (Coal Mines) Scheme, 1930, amendments of the said Scheme as set out in the Schedule hereto have been submitted to the Board of Trade by the Executive Board elected under the provisions of the said Scheme:

Now therefore the Board of Trade, under the provisions of the said Clause and of sub-section (5) of Section 1 of the Coal Mines Act, 1930, hereby approve the said amendments and prescribe the 30th day of June, 1938, as the date from which the said amendments shall come into force.

Harry Crookshank,

Secretary for Mines.

Board of Trade,

Mines Department,

Dean Stanley Street, Millbank, London, S.W.1. 30th June, 1938.

SCHEDULE.

I. There shall be added to Clause 8 of the Scheme after paragraph (iii) thereof the following paragraph:—

ing paragraph: \_\_\_\_\_ ''' (iiia) If at any time after the 31st day of July 1936—(a) a new colliery shall have been or shall be opened, or (b) the working of any colliery not being part of a colliery undertaking shall have been or shall be recommenced following any abandonment or discontinuance of working and in the opinion of the Board the circumstances of the coal mine comprising that colliery make it unfair or inappropriate that an annual standard tonnage in respect of output of coal should be determined therefor the Board shall from time to time determine the standard tonnage in respect of output of coal of such coal mine for each month or quarter. The Board shall determine the annual standard tonnage in respect of output of coal of such coal mine when in their opinion such determination can fairly be made, and thereupon if such coal mine forms part of a colliery undertaking, the standard tonnage so determined shall be added to the annual standard tonnage in respect of output of coal of the other part of that undertaking. For the purposes of this Clause, "colliery" means a shaft or shafts adit or adits with the equipment and development necessary for getting and drawing coal and " new colliery " means a colliery from which coal has not been gotten or drawn save incidentally to the process of development."

2. There shall be added to Clause 8 of the Scheme after paragraph (vii) thereof the following paragraph:-

" (viia) The provisions of paragraph (iiia) of this Clause shall apply with the necessary modifications to the determination of standard tonnages in respect of coal or any class of coal for export supply."

3. In paragraph (xiv) of Clause 8 before the word "standard" wherever that word occurs there shall be inserted the word "annual".

4. In Clause 9 at the end of paragraph (iv) there shall be added the following words

" other than any tonnage relating to a coal mine of which the standard tonnages are determined under paragraphs (iiia) and (viia) of Clause 8 ".

. There shall be substituted for paragraph (ix) of Clause 11 of the Scheme the following paragraph :-

" (ix) If at any time after the 31st day of July, 1936, (a) a new colliery shall have been or shall be opened, or (b) the working of any colliery not being part of a colliery undertaking shall have been or shall be recommenced following any abandonment or discontinuance of working the Board shall determine the basic tonnage of the coal mine comprising the colliery having regard to the development of the colliery and all other relevant circumstances and to the interests of the other owners. Such basic tonnage shall be the annual basic tonnage of the coal mine save that for any period during which in the opinion of the Board the circumstances of that coal mine make it unfair or inappropriate that an annual basic tonnage should be determined therefor the Board may from time to time determine the basic tonnage of that coal mine for each month or quarter having regard to the matters aforesaid. If such coal mine forms part of a colliery undertaking the annual basic tonnage thereof when first determined shall be added to the annual basic tonnage of the other part of that undertaking. Any annual basic tonnage determined under this paragraph shall be

reviewed in accordance with the provisions of paragraph (vi) of this Clause. For the purpose of this Clause the expressions 'colliery' and 'new colliery' shall have the same meanings respectively as are attached thereto under and for the purposes of paragraph (iiia) of Clause 8 of the Scheme.

6. In paragraphs (xii) (xiii) and (xiv) of Clause II the word "annual" shall be omitted.

7. There shall be substituted for the second proviso to paragraph (ii) of Clause 16B of the Scheme the following provision :--

"(iia) Any tonnage in respect of which no compensation is payable under the provisions of the last preceding paragraph shall be divided in proportion to the respective basic tonnages for that month or quarter of the coal mines other than those which having so supplied in that month or quarter a tonnage of coal less than their respective trade shares for that month or quarter are not entitled to the compensation in respect of the whole of the deficiency by virtue of the provisions of the last preceding paragraph. The tonnage thus determined for any coal mine shall be added to the trade share thereof calculated under paragraph (i) of this Clause for that month or quarter, and whenever any such addition is made to any trade share the provisions of the last preceding paragraph and this paragraph shall have effect in relation to that trade share with such addition."

8. In paragraph (ii) of Clause 17 of the Scheme for the words "for the quarter next but one " there shall be substituted the following words :---- " during the calendar year last ".

#### War Office,

5th July, 1938.

#### **REGULAR ARMY.**

The undermentioned Cols. are granted the temp. rank of Brig. :-

H. Finnis, M.C., whilst empld. as a Bde. Comdr. in India. 2nd May 1938.

E. E. Dorman-Smith, M.C., whilst empld. as Dir. of Mil. Trng., Army Hdqrs., India. 10th May 1938.

R. Dening, M.V.O., M.C., whilst empld. as a Bde. Comdr. in India. 18th May 1938.

Col. R. S. Abbott, C.I.E., M.C., A.D.C., is granted the hon. rank of Brig. on retirement. 18th June 1938.

COMMANDS AND STAFF.

The undermentioned appts. are made: —

Lt.-Col. C. Le B. Goldney, M.C., R.A.S.C., to be Asst. Dir. of Supplies & Trans., The British Troops in Egypt, and placed on the h.p. list (empld.). I6th June 1938.

1936. Maj. O. G. Freeman, R.A., from Asst. Experimental Off. (Cl. CC), Chemical Def. Experimental Stn., Porton, to be Dep. Asst. Dir. of Arty., War Office. 7th June 1938. Maj. C. L. Rougier, M.C., Lan. Fus., to be D.A.Q.M.G., W. Comd. 21st June 1938. Capt. A. F. Fisher, 12th L., to be G.S.O. 3rd Grade, War Office. 20th June 1938.

CAVALRY.

- 12th L.-Capt. A. F. Fisher is seed. for serv. on the Staff. 20th June 1938.
- ROYAL REGIMENT OF ARTILLERY.

The undermentioned are secd, for serv. on the Staff:

Maj. G. M. Elliot. 19th May 1938.

Maj. M. Lea-Cox. 13th June 1938.

Capt. H. C. Phillips. 4th June 1938. Capt. T. A. H. Coltman. 25th June 1938. Capt. G. L. Ryan. 1st July 1938.

Lt. (now Capt.) W. G. Stirling to be Adjt. 24th Nov. 1937.

- FOOT GUARDS.
- S. G'ds .-- Lt. Hon. T. W. E. Coke, M.V.O., to be Capt. 7th June 1938.
- INFANTRY.
- Lan. Fus.-Lt. P. G. T. Lutyens to be Capt. 21st June 1938.
- R.W. Fus.—The undermentioned Lts. to be Capts. 6th June 1938:-
  - (local Capt.) T. D. Butler and remains secd.

H. A. S. Clarke.

C. H. V. Pritchard under the prov. of Art. 118A, Royal Warrant for Pay and Promotion, 1931.

Inniskilling Fus.-Capt. W. H. Hynes to be Maj. 9th June 1938.

The undermentioned Lts. to be Capt. 9th June 1938:-

- D. G. Moore and remains secd.
- J. R. Filmer-Bennett and remains secd.
- Ď. M. Shaw.
- Gloster. R .-- Lt. E. B. Cottingham is seed. for serv. under the Air Ministry. 29th Apr. 1938. Lt. A. E. Wilkinson to be Adjt. vice Capt. H. L. W. Bird. 2nd July 1938.
- Stafford R.—Capt. (temp. Maj.) F. G. Rodgers is seed. whilst holding the appt. of Comdt., No. 2. Recruits Phys. Development Depot. 1st June 1938. Capt. B. N. R. Biggie is restd. to the estabt. 1st June 1938.

- R. Berks. R.-Capt. W. A. Bickford is restd. to the estabt. 25th June 1938.
- R.W.K.-Lt. W. E. F. Tuffill to be Capt. 1st Apr. 1938.
- Gordons-Capt. R. E. de S. Pirrie retires on ret. pay. 1st July 1938.
- ROYAL TANKS CORPS.

Maj. E. Munt, M.C., having attained the age for retirement, is placed on ret. pay. 3rd July 1938.

ROYAL ARMY SERVICE CORPS.

Maj. S. G. McBride, O.B.E., M.C., to be Lt.-Col. 16th June 1938.

ROYAL ARMY MEDICAL CORPS.

Maj. (Qr.-Mr.) W. E. Saunders having attained the age for retirement, is placed on

ret. pay. 3rd July 1938. Capt. (Qr.-Mr.) A. Bell to be Maj. (Qr.-Mr.). 3rd July 1938. Serjt.-Maj. William Colquhoun Ross,

Serjt.-Maj. D.C.M., to be Lt. (Qr.-Mr.). 3rd July 1938.

- **OUEEN ALEXANDRA'S IMPERIAL MILITARY NURS-**ING SERVICE.
  - Matron Miss E. S. Emlyn retires on ret. pay. 4th July 1938.
  - undermentioned Sisters be The to Matrons:-

Miss G. C. Ball, A.R.R.C. 24th June 1938. Miss P. M. D. Sowter. 1st July 1938. Miss G. G. Parry, A.R.R.C. 4th July 1938.

- ESTABLISHMENTS.
- R.A.S.C. Trng. Centre-Capt. & Bt. Maj. L. F. Field, R.A.S.C., to be Instr. 5th May 1938.
- Chemical Def. Experimental Stn., Porton-Col. C. Ross, O.B.E., from Member, Ordnance Committee, to be Comdt. (Cl. T). 25th June 1938.
- Ordnance Committee-Col. J. C. G. Hunter, h.p., to be Member. 25th June 1938.

MEMORANDUM.

Capt. H. Colvin, V.C., ret. pay, late Cheshire R., is granted the local rank of Maj. whilst empld. as Zone Recruiting Off., Dewsbury. 1st June 1938.

#### **REGULAR ARMY RESERVE OF** OFFICERS.

The undermentioned having attained the age limit of liability to recall, cease to belong to the Res. of Off. :-

GENERAL LIST.

- Col. A. G. Bayley, C.B.E., D.S.O. (late-Inf.). 5th July 1938.
- Intelligence-Capt. E. G. S. Walker, M.C. 4th. July 1938.
- Miscellaneous-2nd Lt. J. K. Bailey. 5th July 1938.

**REGIMENTAL LIST.** 

Royal Artillery.

Lt.-Col. St. J. R. Nicholson, D.S.O. 5th July 1938.

J. P. Gallivan, D.C.M. Lt. 6th July 1938.

**ROYAL ENGINEERS.** 

Capt. C. G. Banister from Gen. List, Inf., to be Capt. 6th July 1938, retaining his present seniority.

The undermentioned having attained the age limit of liability to recall, cease to belong to the Res. of Off. :-

#### INFANTRY.

- E. Lan. R.-Lt.-Col. W. B. Little, D.S.O., M.C. 6th July 1938.
- S. Stafford R.-2nd Lt. F. G. Deeley, D.C.M. 4th July 1938.

Oxf. & Bucks. L.I.-Capt. S. C. B. Mundey. 3rd July 1938.

#### ROYAL TANK CORPS.

Maj. P. S. Shaw, O.B.E., M.P. 6th July 1938.

SUPPLEMENTARY RESERVE OF OFFICERS.

ROYAL ARTILLERY.

Charles Henry Barnes to be and Lt. 6th July 1938.

ROYAL CORPS OF SIGNALS.

John Lockwood READING late Lt., T.A. Res. of Off. (57th (Home Counties) Fd. Bde., R.A.), to be Lt. 4th June 1938.

INFANTRY.

- Inniskilling Fus.-George Campbell AUSTIN to be 2nd Lt. 6th July 1938.
- E. Lan. R.-2nd Lt. A. J. H. Smith resigns his commn. 6th July 1938.
- Seaforth-Richard Lowther BROAD to be 2nd Lt. 6th July 1938.
- R.U. Rif.-2nd Lt. W. B. de Courcy resigns his commn. 6th July 1938.
- ROYAL TANK CORPS.
  - The undermentioned to be 2nd Lts. 6th July 1938:-

Off. Cadet Philip David REDHEAD from Reading Univ. Contgt., O.T.C.

Edgar Charles Stewart HARGREAVES.

#### TERRITORIAL ARMY.

#### COMMANDS AND STAFF.

The notifn. regarding Maj.-Gen. C. F. Liardet, C.B., D.S.O., T.D., Terr. Army, as Comdr., The Lond. Div. being temp., which appeared in the Gazette of 1st July 1938, is cancelled.

Capt. R. T. R. MacManaway, R.A.S.C., to be D.A.D.S.T., 2nd Anti-Aircraft Div. 27th June 1938.

#### YEOMANRY.

- Shropshire.--Gerald Michael Orlando BRIDGEMAN, the Visct. NEWPORT (late Cadet Serjt., Harrow Sch. Contgt., Jun. Div., O.T.C.), to be 2nd Lt. 6th July 1938.
- ROYAL ARTILLERY.
- 52nd (Manch.) Fd. Bde.—The undermentioned 2nd Lts. to be Lts. :-
  - H. O'B. F. Hayward. 18th Sept. 1937.

  - J. B. Slater. 5th June 1938. J. A. Bowden. 8th June 1938.
- 55th (Wessex) Fd. Bde.—Capt. H. Blackstone resigns his commn. and retains his rank, with permission to wear the prescribed uniform. 1st June 1938.
- Midland) Fd. 68th (S. Bde.—Grosvenor Fleetwood WILLIAMS (late Cadet Serjt., Bromsgrove Sch. Contgt., Jun. Div., O.T.C.) to be 2nd Lt. 6th July 1938.
- 76th (Highland) Fd. Bde.-2nd Lt. H. B. Pirie to be Lt. 6th July 1938.
- 88th (2nd W. Lan.) Fd. Bde.—Anthony Edmond Hubert HAMILTON, late Lt., R.A., to be Lt. with seniority 23rd June 1936. 8th June 1938.
- 98th (Surrey & Sussex Yeo.) Fd. Bde.-Lt. P. G. Evelyn to have seniority 15th Nov. 1934. 18th Dec. 1937.

- 99th (Bucks. & Berks. Yeo.) Fd. Bde.-2nd Lt. C. A. S. Palmer to be Lt. 3rd July 1938. Neville Montague FORD (late Cadet Serjt., Harrow Sch. Contgt., Jun. Div., O.T.C.) to be 2nd Lt. 6th July 1938.
- 62nd (Scottish) Med. Bde.—Capt. S. S. G. C. Harvey to be Maj. 14th June 1938.
- Forth Heavy Bde.-2nd Lt. A. G. Watson to be Lt. 5th July 1938.
- 56th (Cornwall) A.A. Bde.-2nd Lt. G. H. Berriman to be Lt. 14th June 1938.
- 70th (3rd W. Lan.) A.A. Bde.—The undermentioned to be 2nd Lts. 21st June 1938:---Peter Dyson. Donald Percy CROWLE (late Cadet Serjt., St. Bees Sch. Contgt., Jun. Div., O.T.C.).
- 73rd Anti-Aircraft Bde.—Maj. Hon. R. G. Lyttelton, T.A. Res. of Off. (Gen. List, R.A.), to be Hon. Col. 6th July 1938.
- 226th Anti-Aircraft Batt.—The undermentioned to be 2nd Lts. 6th July 1938:-Walter Buchanan Innes GIBB. Charles CALDER.
- 11th (C. of Lond.) Light A.A. Bde.-Derek Rollo HOARE (late Cadet, Malvern Coll. Contgt., Jun. Div., O.T.C.) to be 2nd Lt. 25th June 1938.
- 12th Light A.A. Bde.—Capt. F. W. Webb, R.A., to be Adjt. 27th June 1938.

ROYAL ENGINEERS.

- 43rd (Wessex) Divl. Engrs.—John Skidmore WOOD (late Cadet Lce.-Corpl., Monkton Combe Sch. Contgt., Jun. Div., O.T.C.) to be 2nd Lt. 6th July 1938.
- 55th (W. Lan.) Divl. Engrs.-2nd Lt. W. W. Machin to be Lt. 5th June 1938. Leonard Butterworth GLADWELL to be 2nd Lt. 6th July 1938.
- Dorset (Fortress)-Robert Henry MARSH to be 2nd Lt. 6th July 1938.
- Essex (Fortress)-Capt. & Bt. Maj. R. J. Leeds to be Maj. 13th June 1938.

Lt. H. Baker to be Capt. 12th June 1938.

- Engr. & Rly. Staff Corps-Thomas Wright ROYLE, M.B.E., to be Lt.-Col. 6th July 1938.
- 27th (L.E.E.) A.A. Bn.—The undermentioned to be and Lts. 6th July 1938:-

James Douglas EARLAM. Serjt. Roy Charles Stotter. Lce.-Serjt. James GRAHAM. Corpl. Frank Allen Allen.

28th (Essex) A.A. Bn.-Robert Gordon Scott (late Cadet Corpl., Haileybury Coll. Contgt., Jun. Div., O.T.C.) to be 2nd Lt. 6th July 1938.

32nd (7th C. of Lond.) A.A. Bn.—The undermentioned to be 2nd Lts. 6th July 1938:-

Charles Selwyn PRYOR (late Offr. Cadet, Univ. Contgt., Sen. Div., Cambridge 0.T.C.).

Frederick Colin Dean WOOD (late Cadet Serjt., Tonbridge Sch. Contgt., Jun. Div., O.T.C.).

Lce.-Serjt. Leonard Edward Akhurst, from 27th (L.E.E.) A.A. Bn.

- 33rd (St. Pancras) A.A. Bn.-Frederick George JEWELL (late Cadet Corpl., Highgate Sch. Contgt., Jun. Div., O.T.C.) to be 2nd Lt. 6th July 1938.
- 34th (R.W.K.) A.A. Bn.—Claudius Charles Grigor ANDERSON (late Cadet Serjt., Dulwich Coll. Contgt., Jun. Div., O.T.C.) to be 2nd Lt. 6th July 1938.
- 35th (First Surrey Rif.) A.A. Bn.-Robert Charles BURROWS (late Cadet C.S.M., Dulwich Coll. Contgt., Jun. Div., O.T.C.) to be 2nd Lt. 6th July 1938.
- 41st (N. Stafford R.) A.A. Bn.—The undermentioned to be and Lts. 6th July 1938:-Kenneth Roy LATCHFORD (late Cadet Serjt., Dulwich Coll. Contgt., Jun. Div., 0.T.C.).
  - George Roland HARDING (late Cadet Corpl., Rugby Sch. Contgt., Jun. Div., 0.T.C.).
- 45th (R. War. R.) A.A. Bn.—The under-mentioned to be 2nd Lts. 6th July 1938:— Philip Rufus JONES (late Cadet Serjt., Rugby Sch. Contgt., Jun. Div., O.T.C.). Jack TAYLOR.
- 48th (Hampshire) A.A. Bn.-Richard Noël Fitzroy ALLEN (late Cadet Lce.-Corpl., Portsmouth Gram. Sch. Contgt., Jun. Div., O.T.C.) to be 2nd Lt. 18th June 1938.
- 50th (North'n R.) A.A. Bn.—Sidney William JONES (late Offr. Cadet, Oxford Univ. Contgt. (Arty. Unit), Sen. Div., O.T.C.) to be 2nd Lt. 6th July 1938.

ROYAL CORPS OF SIGNALS.

- 48th (S. Midland) Divl. Sigs.—Lt. P. M. Bowen to be Capt. 5th June 1938.
- 50th (Northumbrian) Divl. Sigs.-Lt. B. Minor to be Capt. 22nd Mar. 1938.

#### INFANTRY.

- 7th/9th Bn. R. Scots.—2nd Lt. W. E. Scott to be Lt. 24th June 1938.
- 7th Bn. The Queen's R.—Alexander Hallam BAIRD (late Cadet C.S.M., Haileybury Coll. Contgt., Jun. Div., O.T.C.) to be 2nd Lt. 6th July 1938. Gordon Leopold Gambier HEATH (late Cadet, Eton Coll. Contgt., Jun. Div., O.T.C.) to be 2nd Lt. 6th July 1938.
- 5th Bn. King's Own R.-Larry Henry Lees WHEELER to be 2nd Lt. 6th July 1938.
- 5th Bn. The King's R.-Lt. R. F. Vickers to be Capt. 15th June 1938.
- 5th Bn. Leicester R.-Malcolm Seymour TAYLOR (late Cadet Serjt., King's Sch. (Worcester) Contgt., Jun. Div., O.T.C.) to be 2nd Lt. 6th July 1938.
- 7th Bn. Cheshire R.—2nd Lt. E. G. S. Mather to be Lt. 13th June 1938.
- 4th Bn. K.O.S.B.-2nd Lt. D. O. Hogg to be Lt. 6th July 1938.
- 4th Bn. Gloster R.—George Ronald BUICK (late Cadet, Bristol Gram. Sch. Contgt., Jun. Div., O.T.C.) to be 2nd Lt. 6th July 1938.

- 4th/5th Bn. D.C.L.I.—2nd Lt. L. S. S. Woolcott to be Lt. 5th June 1938.
- 5th Bn. Border R.-George Ernest GREEN to be 2nd Lt. 3rd July 1938.
- 4th P.W.Bn. Vols.—Edward Arthur HUTCHINGS (late Offr. Cadet, Leeds Univ. Contgt., Sen. Div., O.T.C.) to be 2nd Lt. 26th June 1938.
- 5th Bn. Welch R.—Gwynne Lloyd Phillips to be 2nd Lt. 6th July 1938.
- Bucks. Bn. Oxf. & Bucks. L.I.-Geoffrey Aveline Rowe to be 2nd Lt. 6th July 1938.
- 5th Bn. R. Berks. R.-Lt.-Col. & Bt. Col. A. N. Francis, T.D., is supern. on vacating comd. 21st June 1938. Maj. K. H. H. Allen to be Lt.-Col. 21st

June 1938.

- 4th Bn. K.O.Y.L.I.-James BOOTH to be 2nd Lt. 6th July 1938.
- 5th Bn. K.O.Y.L.I.—Geoffrey Samuel DAVIS (late Cadet, Shrewsbury Sch. Contgt., Jun. Div., O.T.C.) to be 2nd Lt. 6th July 1938. Q.V. Rif., K.R.R.C.—Thomas Gilbert FIELD-
- FISHER (late Cadet Lce.-Corpl., King's Sch. (Bruton) Contgt., Jun. Div., O.T.C.) to be 2nd Lt. 6th July 1938.

5th Bn. Durham L.I.—The undermentioned to be 2nd Lts. 3rd July 1938:-Arthur Peter GARRATT (late Cadet C.S.M., King Edward VI Sch. C. Corps (Stratfordon-Avon)). George CHILD (late Offr. Cadet, Leeds Univ. Contgt., Sen. Div., O.T.C.). Norman Skinner.

7th Bn. H.L.I.—Cyril William Leonard ELLICOTT (late Cadet, Glasgow High Sch. Contgt., Jun. Div., O.T.C.) to be and Lt. 6th July 1938.

ROYAL ARMY SERVICE CORPS.

42nd (E. Lan.) Divl.-2nd Lt. K. R. W. Shrimpton resigns his commn. 27th May 1938.

#### GENERAL LIST.

OFFICERS TRAINING CORPS.

Stanley Wingate ROUTLEDGE (late Cadet Corpl., Merchant Taylors' Sch., Northwood Contgt., Jun. Div.) to be 2nd Lt. for serv. with Bromsgrove Sch. Contgt., Jun. Div. 21st June 1938.

Dulwich Coll. Contgt., Jun. Div.-Lt. A. A. Fletcher-Jones to be Capt. 25th May 1938.

TERRITORIAL ARMY RESERVE OF OFFICERS.

#### GENERAL LIST.

ROYAL ARTILLERY.

Anti-Aircraft-Maj. Bridge Longueville SNOW, late 52nd (Lond.) A.A. Bde., to be Maj. 10th Nov. 1937.

INFANTRY.

Lt.-Col. J. J. O'Hea resigns his commn. and retains his rank. 6th July 1938.

Capt. J. N. F. Morris, from Gen. List (Tonbridge Sch. Contgt., Jun. Div., O.T.C.), to be Capt. 6th July 1938.

. REGIMENTAL LIST.

#### ROYAL ARTILLERY.

- 57th (Home Counties) Fd. Bde.—Lt. J. L. Reading resigns his commn. on appt. to the Supp. Res. of Off. 4th June 1938.
- Tynemouth Heavy Bde.—Lt. F. G. Hewitt, from Active List, to be Lt. 6th July 1938.

#### ROYAL ENGINEERS.

39th (Lan. Fus.) A.A. Bn.—Capt. W. R. Pattinson resigns his commn. and retains his . rank. 6th July 1938.

#### INFANTRY.

- 5th Bn. The King's R.—Capt. & Bt. Maj. J. P. Ashton, M.C., from Active List, to be Capt. & Bt. Maj. 15th June 1938.
- 8th Bn. Midd'x R.—Lt.-Col. & Bt. Col. William Henry Parmeter, T.D., is removed from the T.A. on conviction by the Civil Power. 5th Mar. 1938.

### Air Ministry, 5th July, 1938.

#### ROYAL AIR FORCE.

The KING has been graciously pleased to approve of the undermentioned reward for gallant and distinguished service rendered during the operations in Waziristan, 1938:—

Distinguished Flying Cross.

Flight Lieutenant George Eric JACKSON.

Air Ministry, 5th July, 1938.

#### ROYAL AIR FORCE.

GENERAL DUTIES BRANCH.

Acting Pilot Officer on probation Lindon JONES (since deceased) is confirmed in his appointment and graded as Pilot Officer. 31st May 1938.

The undermentioned Flying Officers are promoted to the rank of Flight Lieutenant on the dates stated:—

7th May 1938.

Rowland Gascoigne Musson.

15th June 1938.

Hubert Mortimer STYLES. Howard Duncan FRASER. Donald Henry LEE.

22nd June 1938.

John Cornelius Taylor.

The undermentioned Pilot Officers are promoted to the rank of Flying Officer on the dates stated:—

26th Apr. 1938.

Bertram Eric DOBB.

21st May 1938.

Patrick Edward Geoffrey Gunnell CONNOLLY.

25th May 1938.

Michael Nicholson CROSSLEY.

26th May 1938. Michael Warnford HAMLYN.

16th June 1938.

Gordon Learmonth RAPHAEL.

18th June 1938.

Robert Henry WARCUP. Kenneth Thomas THOMPSON.

19th June 1938.

Harry Emlyn Bufton. Howard Frizelle Burton. Ian Scovil Soden. Leslie Frank Cooper. Louis Douglas WILSON. Patrick Edgar Sempill Fairfax Monteagle BROWNE. Kenneth AULT. Arthur Richard VAUGHAN WILLIAMS. Guy Percy ROBINSON. Douglas Anthony ffrench-Mullen.

25th June 1938.

John Alan RONCORONI. Frederick Clement RIVERS. Philip Robert BEARE. Gerald William Chavallier GLEED. William Hedley CLIFF. James Alan RICHARDSON.

The undermentioned Flying Officers are granted the acting rank of Flight Lieutenant on the dates stated:—

24th Mar. 1938.

Royd Martin FENWICK-WILSON.

20th May 1938.

Lionel Woodland SABEN.

25th May 1938.

Richard Winter HASE.

Wing Commander Thomas Frederick Wailes THOMPSON, D.F.C., is placed on the retired list at his own request. 30th June 1938.

Squadron Leader Noel William Aubrey CULLUM is transferred to the Reserve, class A. 6th July 1938.

The short service commission of Acting Pilot Officer on probation Derek Bell WILLIAMS is terminated on cessation of duty. 24th June 1938.

The short service commission of Acting Pilot Officer on probation Terence Walter CALLINGHAM is terminated on cessation of duty. 24th June 1938.

EQUIPMENT BRANCH.

Group Captain Frederick Alfred BALDWIN is placed on the retired list at his own request. 1st July 1938.

ROYAL AIR FORCE RESERVE.

Reserve of Air Force Officers.

GENERAL DUTIES BRANCH.

Flight Lieutenant John Reginald Shelton AGAR is transferred from class A to class C. and July 1938. Flying Officer Eric Herbert AllIOTT is Sept. 1937.

transferred from class A to class C. 21st Flying Officer John Molony LONGLEY relinquishes his commission on appointment to a commission in the Royal Air Force Volunteer Reserve. 2nd Apr. 1938.

Pilot Officer John William STEWART relinquishes his commission on appointment to a commission in the Royal Air Force Volunteer Reserve. 8th May 1938.

Flight Lieutenant Harold Fraser GAME, A.F.C., relinquishes his commission on completion of service. 1st June 1938.

EQUIPMENT BRANCH.

Wilfred Beresford Joe HUMPHREY is granted a commission as Flight Lieutenant in class C. 16th Feb. 1938.

Royal Air Force Volunteer Reserve.

GENERAL DUTIES BRANCH.

John Molony LONGLEY is granted a commission as Flying Officer with effect from 2nd Apr. 1938 and with seniority of 2nd Oct. 1929.

Reginald George SOLLARS is granted a commission as Acting Pilot Officer on probation. 5th July 1938.

John William STEWART is granted a commission as Pilot Officer with effect from 8th May 1938 and with seniority of 9th Mar. 1937.

EQUIPMENT BRANCH.

The undermentioned are granted commissions as Acting Pilot Officers on probation with effect from 5th July 1938 and with seniority of the dates stated:—

17th May 1938.

Otto Maxwell BERKELEY.

21st June 1938.

John Gill DAVIS.

28th June 1938. Ronald Ernest William LITTLE.

5th July 1938.

Charles COOKE. Reginald Gordon KINGSTON. Ernest David Wallers MERCER. Frederic Leo MILLS. Arthur Ernest RICH. Edwin Arthur SADLER. Alec Harold Kenneth STEVENS. Ernest Hector PROCTOR.

#### AUXILIARY AIR FORCE.

GENERAL DUTIES BRANCH.

No. 502 (Ulster) (Bomber) Squadron. Gordon Henderson is granted a commis-

sion as Pilot Officer. 14th Apr. 1938.

No. 603 (City of Edinburgh) (Bomber) Squadron.

Flying Officer Patrick GIFFORD is promoted to the rank of Flight Lieutenant. 1st Apr. 1938.

No. 605 (County of Warwick) (Bomber) Squadron.

12th June 1938.

Ralph HOPE.

#### 13th June 1938.

Graham Walter Beech AUSTIN.

Pilot Officer Patrick George LEESON is promoted to the rank of Flying Officer. 5th June 1938.

No. 609 (West Riding) (Bomber) Squadron.

George Desmonde Ayre Ayre is granted a commission as Pilot Officer. 13th May 1938.

No. 612 (County of Aberdeen) (Army Cooperation) Squadron.

Ian George Flaxington STEPHEN is granted a commission as Pilot Officer. 3rd June 1938.

No. 614 (County of Glamorgan) (Army Cooperation) Squadron.

The undermentioned are granted commissions as Pilot Officers on the dates stated:----

2nd May 1938. William Lonsdale Drummond.

3rd May 1938. John Dudley Rollinson.

4th May 1938.

Alexander Glen PALLOT.

5th May 1938.

Richard Geoffrey England.

EQUIPMENT BRANCH.

Major Edward Johnson MOORHEAD (R.A. Ret.) is granted a commission as Flight Lieutenant. 16th May 1938.

#### Civil Service Commission,

July 5, 1938.

The Civil Service Commissioners hereby give notice that a Competitive Examination for situations as Shorthand-Typist and Clerk-Shorthand-Typist in Departments of the Civil Service will be held in Belfast, Birmingham, Bristol, Cardiff, Edinburgh, Glasgow, Leeds, Liverpool, London, Manchester, Newcastle-on-Tyne, Plymouth and Portsmouth, commencing on the 3rd November, 1938, under the Regulations dated the 5th February, 1937, and published in the London Gazette of the same date.

No person will be admitted to Examination from whom the Secretary of the Civil Service Commission has not received on or before the 8th September, 1938, an application, in the handwriting of the candidate, on the prescribed form, which may be obtained from the Secretary at once.

#### RULES PUBLICATION ACT, 1893.

IN THE MATTER OF THE LONDON TRAFFIC ACT, 1924.

The Minister of Transport hereby gives notice under the Rules Publication Act, 1893, that he proposes after the expiration of at least 40 days from this date in pursuance of the powers con-ferred upon him by Section 10 of the London Traffic Act, 1924, to make the London Traffic (Prescribed Routes) (No. ) (Vauxhall Cross, (Prescribed Routes) (No. ) (Vauxhall Cross, Lambeth) Regulations, 1938, in regard to the following scheme of "one-way" traffic working : \_

traffic working.
~ 0
West to East.
East to West.
South West to North East.
•
South to North.
North to South.
North to South.

Any observations or suggestions in connection with the draft Regulations should be sent in writing to the Secretary, Ministry of Transport, Metropole Buildings, Northumberland Avenue, London, W.C.2, from whom copies can be obtained price 1d. per copy.

Dated this fourth day of July, 1938.

H. W. W. Fisher,

An Assistant Secretary. Ministry of Transport, Metropole Buildings, Northumberland Avenue, W.C.2.

MERSEY CONSERVANCY ACT.

Mersey Conservancy, Ministry of Transport, Metropole Buildings, Northumberland Avenue, London, W.C.2. 1st July, 1938.

I hereby give Notice, pursuant to the 9th Section of the Act 5 & 6 Victoria, cap 110, intituled "An Act for better preserving the

Navigation of the River Mersey," that I have received a Notice from Messrs. Shell Refineries Limited given pursuant to the provisions of the above mentioned Act, of their intention to widen an existing bridge over the River Gowy in accordance with plans and sections deposited at this Office, a copy of which Notice is subjoined.

#### H. P. Douglas,

#### Vice-Admiral,

#### Acting Conservator of the Mersey.

Vice-Admiral Sir H. Percy Douglas, K.C.B., C.M.G., F.R.G.S. Acting Con-To Acting Conservator of the Mersey.

In pursuance of an Act for better preserving the Navigation of the River Mersey (5 and 6 Vic., cap. 110)

Shell Refineries, Limited, We, Stanlow Installation, nr. Ellesmere Port, Wirral, Cheshire, Refiners of Petroleum Products, hereby give you notice that we propose to construct at Stanlow, nr. Ellesmere Port, Wirral, Cheshire, in accordance with Plans and Sections deposited at the Ministry of Transport a widening of the existing road and rail bridge across the River Gowy for the purpose of carrying pipelines. The new work will be 20 feet wide, of reinforced concrete construction supported on piles. The openings will be such as not to impede the flow of the river. The work will be proceeded with as soon as possible.

Dated this 27th day of June, 1938.

For Shell Refineries, Ltd.,

(Sgd.) C. J. Matthews.

#### MINISTRY OF TRANSPORT. TRUNK ROADS ACT, 1936.

#### TRA. 105/3.

Notice is hereby given that the Minister of Transport has made an Order under Subsection 3 of Section 1 of the Trunk Roads Act, 1936, whereby the route specified in Part I of the Schedule to the Order becomes a Trunk Road.

Copies of the Order, the short title of which London-Fishguard Trunk is The Road (Llangyfelach and Bryntywod By-pass) Order, 1938, can be purchased from His Majesty's Stationery Office at the following addresses: — Adastral House, Kingsway, London, W.C.2; 120, George Street, Edinburgh 2; 26, York Street, Manchester 1; and 1, St. Andrew's Crescent, Cardiff.

Dated this thirtieth day of June, one thousand nine hundred and thirty-eight.

P. Wilson,

Assistant Secretary. Metropole Buildings,

Northumberland Avenue,

London, W.C.2.

West to East. East to West. South West to North East.

Direction of

South to North.

#### POST OFFICE

#### IMPERIAL AND FOREIGN PARCEL POST

The rates of postage which are directed to be charged in place of rates previously published on outgoing parcels addressed to the undermentioned territories by the routes specified are as follows—

Place	of Dest	ination	•	Route.	F	Rate				e or eedi			Parce	el
					2	lb.	3	lb.	7	lb.	11	lb.	22 ]	lb.
Austria Bulgaria Hungary		····	 	Sea direct to Hamburg Via Belgium Via Germany Via Belgium and Germany	s. 1 1 3 2	d. 6 9 0 3	s. 2 2 4 3	d. 6 9 3 3	s. 2 2 4 3	d. 6 9 3 3	s. 3 3 4 3	d. 0 3 9 9	s. 4 5 8 6	d. 9 0 3 6

#### BOROUGH OF HORNSEY.

#### CONFIRMATION OF BYELAW.

NOTICE is hereby given that the Town Council of Hornsey intend, after the expiry of one calendar month from the date of the publication of this notice to apply to the Minister of Health for confirmation of a Byelaw made by the Council in pursuance of Section 104 (I) of the Public Health Act, 1936 for regulating in the Borough of Hornsey the emission of black smoke from a chimney of any building other than a private house.

H. Bedale,

Town Clerk.

Town Hall, Crouch End, N.8. 27th June, 1938. (190)

#### URBAN DISTRICT OF MABLETHORPE AND SUTTON.

#### CONFIRMATION OF BYELAWS.

NOTICE is hereby given that the Urban District Council of Mablethorpe and Sutton, intend, after the expiry on one calendar month from the date of the publication of this notice, to apply to the Minister of Health for confirmation of Byelaws made by the Council with respect to Building.

> R. VINCENT LEWIS, Clerk of the Council.

30th June, 1938. (027)

(152)

•1

#### LEEDS CORPORATION LIGHT RAIL-WAYS (EXTENSION) ORDER, 1935.

WHEREAS the period limited for the completion of the whole of the Railways authorised by the Leeds Corporation Light Railways (Extension) Order, 1935, would, unless further extended by the Minister of Transport, expire on the sixteenth day of July, 1938.

Now therefore the Minister of Transport by virtue and in exercise of the powers vested in him by the said Order, does hereby extend the said period to the fifteenth day of July, 1940.

Signed this 1st day of July, 1938.

(Sgd.) S. J. PAGE, Assistant Secretary, Ministry of Transport.

#### COUNTY OF DEVON.

ROAD TRAFFIC ACT, 1930.

ROAD AND RAIL TRAFFIC ACT, 1933.

NOTICE is hereby given that on the 22nd June, 1938, the Minister of Transport confirmed the Devon County Council (Roads Restriction) (No. 3) Order, 1937, made by the Devon County Council under Section 46 (1) of the Road Traffic Act, 1930, and Section 29 of the Road and Rail Traffic Act, 1933. The Order provides that subject to certain exemptions no person shall drive any heavy locomotive, light locomotive, motor tractor or heavy motor car (as respectively defined in Section 2 of the Road Traffic Act, 1930) or any motor car (as defined as aforesaid) with seats for more than 14 persons exclusive of the driver on the road from the junction at the west end of Station Road, South Brent by the Lower Bridge, North via Lydia County Bridge, Aish, Binnamore Cross, Badworthy Bridge,

The Order is available for inspection at the Castle, Exeter.

A. J. WITHYCOMBE,

Clerk of the Council.

The Castle, Exeter. 1st July, 1938.

(207)

#### BOROUGH OF TODMORDEN.

#### ROAD TRAFFIC ACT, 1930.

#### ROAD AND RAIL TRAFFIC ACT, 1933.

NOTICE is hereby given that on the 23rd June, 1938, the Minister of Transport confirmed the Borough of Todmorden (Road Restriction) Order, 1938, made by the Todmorden Borough Council under Section 46 (1) of the Road Traffic Act, 1930 and Section 29 of the Road and Rail Traffic Act, 1933. The Order, as confirmed provides that subject to certain exceptions no person shall drive any vehicle on Walton Lane, Todmorden between its junctions with Cross Stone Road and the road known as Back Millwood.

The Order is available for inspection at the Municipal Offices, Todmorden. (208)

#### THE BANKRUPTCY ACTS, 1914 AND 1926.

RETURN showing the Number of Receiving Orders and Administration Orders in Bankruptcy (Deceased Debtors' Estates) in the High Court, and in the several County Courts having Bankruptcy Jurisdiction in England and Wales, gazetted in the Quarters and in the Half-years ending respectively 30th June, 1938, 1937 and 1936, and also the total number of Deeds of Arrangement registered during the same periods :---

Court.		For Quarter ending 30th June, 1938.	For Quarter ending 30th June, 1937.	For Quarter ending 30th June, 1936.	ending	For Half Year ending 30th June, 1937.	For Half Year ending 30th June, 1936.
High Court		158	162	131	309	318	298
Aberdare and Mountain Aberystwyth	Ash	2 I	2	3	4 I	2 3	і 3
Ashton-under-Lyne and	i''' (				1	3	3
Stalybridge		I	3 6	4	3	9	5
Aylesbury Banbury		3 I	0 2	3	7 1	9 3	7 I
Bangor			4	4		14 14	9
Barnsley		3 3	5	<u> </u>	5 5 6	7	I
Barnstaple Barrow - in - Furness a	nd	3		2	. 6	-	3
· Ulverston	ла 		I	I	3	4	5
Bath		4	· <u>-</u>		4	2	5 7 5 6
Bedford		I	4	3	3	5	5
Birkenhead Birmingham		5 16	3	4 26	7	5	
Birmingnam Blackburn and Clitheroe		10 5	20 7	20 4	. 25 I3	31 15	41 16
Blackpool		J 7	7 6	T I	13	12	4 ·
Bolton	]	4 3	• 7	9 6	II	14	14
Boston Bournemouth		3	10		10 6	15	12
Bradford		4 16	6 17	5 11	0 29	11 30	24 . 36
Brentford		4	17	10	II	16	15 15
Bridgend		3	' I	2	5	4	5 5
Bridgwater		. 4	4	I	IO	4	5
Brighton and Lewes Bristol	[	21	12 10	15 11	36 25	25	27
Bristol Burnley		9 3	10	4	25 7	22 3	17 7
Burton-on-Trent		2	5		5	9	7
Bury St. Edmunds		I	I	3 3 7 3	5 3 6	2	3
Cambridge Canterbury		5	3	7	6 18	6	12 10
Cardiff and Barry		9 3	5 3	3. 4	8	14 8	10
Carlisle		3	I	2	4	3	5
Carmarthen	[	4	IO	3	12	15	5 26
Chelmsford		9	18	12	19	26	
Cheltenham Chester		2	2 I	2	I 3	3 5	1 4
Chesterfield		2	4		3 6	3 5 5	
Colchester	•••	9 3	2	6	15	4	15 8
	nd	3	3	6	7	7	8
Candhaah	]	I	5	2	5	12	8
Croydon		25	5 18	28	42	43	51
Derby and Long Eaton	{	10	II	8 8	15	22	16
Dorchaster	•••	I	I	8	7	5	II
Dudlar		3 5	5 3	3	7 5 6 3 6	7 5	12 3
Durham				5 2	3	2	3
Eastbourne		I	2 `			2	4 18
Trator		5 5	II	9	12 6	17 6	
Fromo		5	I I	9 5 2 9 4	0 3	0 2	12 5
Gloucester	••••	3	r	2	3 5		5 4
Great Grimsby		7	8	9.	20	5 16	24
Creamaniah		2	5	4	6	II	7
Cwildford		7	7	5	8	II T2	13
Unlifor	•••	3	7	4	7 11	13 12	9. 9.
		<b>~</b> '	- 1	- 1	1	1	7

4338

-

Court	For Quarter ending 30th June, 1938.	For Quarter ending 30th June, 1937.	For Quarter ending 30th June, 1936.	For Half-year ending 30th June, 1938.	For Half-year ending 30th June, 1937.	For Half-year ending 30th June, 1936.
Hanley and Stoke-upon-		}			1	
Trent	8	4	6	14	11	a
Harrogate	6 ·	3	3	7	4	
Unating	I	J	3	3	3	9 7
TT	ī	2				9 9 7 2 3
Uaroford	2	Ĩ	I	5 5	2	2
Trantford	I	3		J J	4	5
Huddersfield	5	I	3 6	6		12
Ipswich	9	5	3	14	4 10	7
Kendal	2		2	2	I	3
Kidderminster			Ĩ		<u> </u>	3
King's Lynn	7	4	8	16	14	13
Kingston, Surrey	4	4 8		10	16	-3 I3
Kingston-upon-Hull	13	20	3	34	36	22
Leeds	13	16	23	28	29	52
Leicester	8	9	13	14	23	28
Leominster	I —		-5			3.
Lincoln and Horncastle	8	8	. 10	II	13	20
Liverpool		17	13	25	33	26
Luton	9 6	7	3	11	13	II
Macclesfield	4	3	I _	5	3	2
Maidstone	2	\ <u> </u>	I	3	I I	3
Manchester	23	20	• 13	39	40	27
Merthyr Tydfil	2	2	2	3	5	4
Middlesbrough	2	I	2	7	Ğ	4
Neath and Port Talbot	) —	2	2	Í	3	4
Newbury	2			3	Ĩ	3
Newcastle-on-Tyne	6	13	8	14	21	17
Newport and Ryde	—	3	4	2	4	5
Newport, Mon	-	3		2	7	9 I
Newtown	4	I	r	5	I	
Northallerton	36	I		3	2	2
Northampton	6	3 8	9	12	7	{ I3
Norwich	8	8	9	22	20	17
Nottingham	8	16	15	17	32	28
Oldham	6	6	3	8	13	6
Oxford	2	2	4	8	6	9
Peterborough	5 5	56	34	12	13	9
Plymouth	5	0	4	16	17	II
Pontypridd, Ystradyfodwg					Į	
and Porth	— ·	I.	4 I	I	2	II
Portmadoc and Festiniog	6		I		I	2
Portsmouth Preston	6 3 3	9	5 8 5	15 6	18	18
Deading			Š	0	8	15
Destration	3		5	32	14	0
"Decharter"	1		4		8	5
St Albong	4	9 3 7 5 7 5 7 5 9	2 5 13	15	II	15 6 9 9 23 9 13
Calford	9 11	2		13	9 16	<u>9</u> .
Calisburg		9	2	19 17		43
Searborough	5	2		17 15	9	Y   T2
Sheffield	22	14	17	42	3 28	42
Shrewsbury		4		6	6	44
Southampton	3	5	I 6	12	8	II
Southend		10	-	6	14	II
Stafford	Ĩ		5 2	I	<u></u>	2
Stockport	2		1 _	7	14	
Stockton-on-Tees	3	53	4	7	7	8
Stourbridge	2	<u> </u>		2	2	2
Sunderland	_	5	4	2		7
Swansea	I	5 3	4		7 5 5 3	9 8 2 7 6
Swindon	2	J I	3	76	5	4
Taunton	2	ī	4 3 I	3	~	3
Torquay	2	5	5	3. 6	9	17
Tredegar, Abertillery and				1		
Bargoed	1	2	6	I	3	10
Truro and Falmouth	6	IO	7	14	3 18	15
Tunbridge Wells	4	4	I	7	6	2
-	-	-				

Court.	For Quarter ending 30th June, 1938.	For Quarter ending 30th June, 1937.	For Quarter ending 30th June, 1936.	For Half Year ending 30th June, 1938.	For Half Year ending 30th June, 1937.	For Half Year ending 30th June, 1936.
WakefieldWalsallWandsworthWarringtonWarwickWellsWest BromwichWhitehaven and MillomWinchesterWinchesterWindsorWolverhamptonWorcester	3 36 5 2 2 3   - 36 3 3	8 4 5 4 4 2 5 4 3 9 4 4	7 5 4 1 1 1 3 2 7 2 2	10 7 15 6 3 4 5  1 4 16 9 10	17 58 56 2 5 1 7 4 14 6 9	17 9 10 3 3 3 3 3 5 2 7 5 3 16 4 6
WorkingtonandCocker- mouthWrexhamWrexhamYeovilYork	<u>п</u> 4 5		1 3 3 3	2 3 6 12	9 8 5 11	1 3 5 7
Total for County Courts Total for High Court and County Courts	600  75 <sup>8</sup>	688 850	652 783	1,262  1,571	1,331 1,649	1,392 1,690
Total number of Deeds of Arrangement registered	419	422	380	895	<sup>8</sup> 57	865

STATEMENT showing the Quantities Sold and Average Price of BRITISH CORN, per Hundredweight of 112 Imperial Pounds, FOR THE QUARTER ending Midsummer, 1938, pursuant to the Corn Returns Act, 1882, and the Corn Sales Act, 1921, with comparative figures for three earlier years :--

	ter endi	ç	Juantities Solo	<b>1.</b> ·	Avera	age Price pe	r Cwt.
Mid	summer	Wheat.	Barley.	· Oats.	Wheat.	Barley.	Oats.
1935 1936 1937 1938	···· ···	   Cwt. 2,169,773 1,936,160 2,149,572 1,535,989	Cwt. 428,783 314,514 243,197 87,240	Cwt. 260,655 340,678 364,658 246,954	s. d. 5 3 6 7 9 10 7 10	s. d. 7 I 7 6 9 II 10 6	s. d. 7 2 6 1 8 10 8 3

Ministry of Agriculture and Fisheries, 4th July, 1938. R. H. FRANKLIN, Assistant Secretary.

4340

STATEMENT showing the QUANTITIES SOLD and AVERAGE PRICE of BRITISH CORN per Hundredweight of 112 Imperial Pounds,\* as received from the INSPECTORS of CORN RETURNS in the week ended 2nd July, 1938, pursuant to the Corn Returns Act, 1882, and the Corn Sales Act, 1921:--

]	British	Corn.			Quantities Sold.	Average Price per Cwt.
•••		••••			<b>Cwt.</b> 46,819	s. d. 7 II
	•••	•••			1,266	9 5
*,* *		•••	•••		12,339	8 9
						Cwt.                       1,266

COMPARATIVE STATEMENT for the Corresponding	g Week in each of the Years from 1934 to 1937	:
---	---	---

Week ended		Q	uantities Sol	d.	Average Price per Cwt.					
WEEK Children	•	Wheat.	Barley.	Oats.	Wheat.	Barley.	Oats.			
30th June, 1934	••••	Cwt. 75,291	Cwt. 4,780	Cwt. 25,650	s. d. 5 7	s. d. 6 6	s. d. 6 5			
29th June, 1935		84,653	9,8 <u>3</u> 4	10,353	56	72	76			
4th July, 1936		63,413	3,554	13,897	69	66	6 I			
3rd July, 1937		97,856	4,421	12,431	96	98	93			

COMPARATIVE STATEMENT for each of the Four previous Weeks :---

Week ended.	Qı	uantities Sol	ld.	Average Price per Cwt.					
week chucu.	Wheat.	Barley.	Oats.	Wheat.	Barley.	Oats.			
4th June, 1938	 Cwt. 68,277	Cwt. 3,413	Cwt. 22,826	s. d. 8 o	s. d. 11 3	s. d. 8 7			
11th June, 1938	 113,255	1,622	26,492	8 o	98	84			
18th June, 1938	 54,910	1,630	13,942	7 10	10 8	8 10			
25th June, 1938	 73,588	853	9,731	7 II	95	83			

\* Section 8 of the Corn Returns Act, 1882, as amended by Section 2 of the Corn Sales Act, 1921, provides that, in the weekly summary of quantities and prices, each sort of British corn shall be computed with reference to the hundredweight of one hundred and twelve imperial standard pounds.

NOTE.—The prices in the first statement above are based on returns received from Inspectors during the week ending 2nd July, 1938, and represent on the whole the average prices ruling in the week ending 25th June, 1938. The weeks cited in the comparative statements are similarly to be regarded as weeks of Inspectors' returns.

R. H. FRANKLIN,

Assistant Secretary.

в

Ministry of Agriculture and Fisheries, Whitehall Place, London, S.W.I. 2nd July, 1938.

#### Crown Copyright Reserved.

AN ACCOUNT of the IMPORTATIONS of BULLION and SPECIE registered from documents received in the Statistical Office, Customs and Excise, London, from mid-day on the 30th June, 1938, to mid-day on the 4th July, 1938.

			•	Gold.					SILVER.			
		Bullion.		Co	in.		Bull	lion.	Co	in.		Total of
Countries.	Unrefined, in dust, amalgam ( <i>i.e.</i> , lumps), and bars.	Refined, in 1 In the form accepted in inter-bank transactions.	oars, etc. Other.	Of legal tender in the United Kingdom.	the tender in the Gold. Unrefined. Refined.		Of legal tender in the United Kingdom.	Not of legal tender in the United Kingdom.	Total of Silver.	Gold and Silver.		
British West Africa British East Africa British India Hong Kong Australia Belgium France Switzerland Other Countries	£ 198,872 6,477     486	£ 	. £ 	£ 16,750 577 1,700	£ 	£ 198,872 6,477 16,750 65,522 12,300 612,115 35,587 9,585 3,818	£  	£ 	£   	£  	£ — — 2,800 132 48 841	£ 198,872 6,477 16,750 65,522 12,300 614,915 35,719 9,633 4,659
Total Declared Commodity Value of the Importa- tions registered from mid- day on the 30th June, 1938, to mid-day on the 4th July, 1938.	205,835	677,637	45.900	19,027	12,627	<b>961,026</b>	2,809		848	164	3,821	964,847

IMPORTED INTO THE UNITED KINGDOM.

Statistical Office, H.M. Customs and Excise, Dudley House, Endell Street, W.C.2. W. HENDERSON, Controller. . . .

4th July, 1938.

.

,

4342

THE LONDON GAZETTE, 5 JULY,

1938

### Ĉrown Ĉopyright Reserved.

AN ACCOUNT of the EXPORTATIONS of BULLION and SPECIE registered from documents received in the Statistical Office, Customs and Excise, London, from mid-day on the 30th June, 1938, to mid-day on the 4th July, 1938.

-		•	•	Gold.	-				SILVER.			
• , ,		Bullion.		. Ca	oin.		Bul	lion.	· Co	in.		Total of
Countries.	Unrefined, in dust, amalgam	Refined, in 1	oars, &c.	Of legal tender in the		Total of Gold.	Unrefined.	Refined.	Of legal tender in the	Not of legal tender in the	Total of Silver.	Gold and Silver.
· · · · ·	(i.e., lumps), A and bars.	In the form accepted in inter-bank transactions.	Other.	United Kingdom.	United Kingdom.		Onreimed.	Kenned.	United Kingdom.	United Kingdom.		
 Sweden	<u>_</u> £	£ 770,260	£	£	£	£ 770,260	. <u>f</u>	£	<u>ب</u>	<u>+</u>	<u>£</u>	£ 770,260
Germany		—	` ·	—			—	4,057		<u> </u>	4,057	4,057
Netherlands	— <sup>.:</sup>	70,848		37,864	180,008	288,720			·	-		288,720
France		-	-	. 17,540	8,691	26,231		823	. —	_	823	27,054
Şyria		-		8,459		8,459			. — .		_	8,459
United States of America	53,229	-		· 		53,229		16,6 <b>7</b> 0	·	860,000	876,670	929,899
Other Countries		-	450	841	·;	1,291		7 <sup>8</sup> 5	100	216	1,101	2,392
Total Declared Commodity Value of the Exports of	:		-			·						
Bullion and Specie regis- tered from mid-day on the 30th June 1938, '' to mid-day on the 4th	53,229	841,108	450	64,704	188,699	1,148,190		22,335	100	860,216	882,651	2,030,841
July, 1938.	7.6	· · · · · ·			··.			· ·		·		
Statistical Office, H. 4th July,	M. Customs an 1938.	d Excise, Du	idley Hou	ise, Endell Str	eet, W.C.2.					W. HEND	DERSON, Co	ontroller.

#### EXPORTED FROM THE UNITED KINGDOM.

<u>.</u>

ç.

. . . . . . .

Average Price of BRITISH WHEAT, BARLEY and OATS per Cwt. of 112 Imperial lbs. as received by the Ministry of Agriculture and Fisheries from the Inspectors of Corn Returns at each of the undermentioned Towns during the week ended Saturday, the 2nd July, 1938:--

. .

·	<del></del>			· · · ·	<del></del>		
Towns.	Wheat.	Barley.	Oats.	Towns.	Wheat.	Barley.	Oats.
Bedfordshire :	s. d.	s. d.	s. d.	Hampshire :	s. d.	s. d.	s. d.
Bedford	8 I		90	Andover	8 I		_
Luton	Nil.			Basingstoke	8 2	-	8 10
Berkshire:				Fareham	80 81		
Abingdon	Nil.			Newport Southampton	Nil.	_	_
Hungerford	Nil.			Winchester	7 9		9 2≍
Newbury	7 5		9 I				,
Reading	7 9		90	Herefordshire :	•		
Wallingford	7.9	12 0		Hereford	8 0 NU	-	
Buckinghamshire:	1	Í Í		Ross	Nil.		
Aylesbury	8 0	_	94	Hertfordshire:			
Newport Pagnell	Nil.	ŀ	<b>у</b> т	Bishop's Stortford	Nil.		
				Hertford	8 I	—	
Cambridgeshire:—				Hitchin	7 II		
Cambridge	8 0 8 1	-	88	Royston	8 o	-	
Ely Wisbech	Nil.			Huntingdonshire :			
				St. Ives	8 0	_	
Cheshire :				St. Neots	8 I		
Chester	Nil.						
Cornwall:				Kent:-			
Truro	84		8 2	Ashford Canterbury	79 Nil.	-	
Wadebridge	Nil.		0 2	Maidstone	Nil.	[	
				Rochester	Nil.		
Cumberland :				Sandwich	Nil.		
Carlisle		-	7 6	7		Í	
Penrith	-	-	79	Lancashire:— Manchester	8 т	_	9 O-
Derbyshire :—				Wanchester Warrington	Nil.		90
Derby	Nil.	i j		, e	-		
Demonshing				Leicestershire:	• _		
Devonshire :— Barnstaple				Leicester Loughborough	8 1 Nil.	-	
Exeter	7479		8 6	Loughborough Melton Mowbray	8 5		
Kingsbridge	Nil.		ŨŨ	indicon indicary			
Newton Abbot	85		—	Lincolnshire:			
Okehampton	Nil.			Boston	Nil.	Ì	
Plymouth Tiverton	Nil. Nil.	•		Brigg Gainsborough	Nil. 8 o	}	
Totnes	8 6		8 0	Grantham	8 0	· _	_
			ŬŬ	Lincoln	8 3	7 6	93:
Dorsetshire :				Louth	8 0	76	_
Blandford	8 0	-		Sleaford	Nil.		
Bridport Dorchester	89 81	76		Spalding Stamford	Nil. Nil.	ļ	
Wareham	Nil.			Stamford	1711.		
Wimborne	7 11		_	London:			
· ·				London	8 O	II 3	
Durham :		. 1	8 2	Middlesex :			
Darlington Stockton-on-Tees	7974		8 2	Uxbridge	Nil.		
Sunderland	7 4 Nil.		_		1,11.		
_	{			Monmouthshire :		ł	
Essex :				Abergavenny	Nil.	4	
Braintree Chelmsford	Nil. Nil.			Chepstow	8 o Nil.	- (	-
Colchester	8 3		97	Newport	1411.	l	
Romford	Nil.	-	9/	Norfolk :		]	
Saffron Walden	8 2			Diss	7 11	-	8 <b>10</b> .
<u>.</u>				East Dereham	7 10	69	
Gloucestershire :	Nil.			Fakenham	Nil.		
Cheltenham Cirencester	1		89	Harleston Holt	7 9 Nil.	-	_
Gloucester	7978	_		Lynn	7 6	_	
Tewkesbury	78 Nil.	{	· · [	North Walsham	·	- [	8 10-

### Average Price of BRITISH WHEAT, BARLEY and OATS-continued.

		·		1	ŀ		
Towns.	Wheat.	Barley.	Oats.	Towns.	Wheat.	Barley.	Oats.
Norfolk—continued— Norwich Watton Yarmouth	s. d. 8 0 8 0 7 7	s. d. — —	s. d. '9 0  8 10	Sussex :	s. d. Nil. 8 I . — Nil.	s. d. 	s. d. 9 10 8 10
Northamptonshire: Kettering Northampton Peterborough Northumberland:	7 11 8 2 Nil.			Lewes Warwickshire : Birmingham Coventry Stratford - on - Avon	Nil. 8 1 7, 10		11 0 
Alnwick Berwick Newcastle - on - Tyne Nottinghamshire :	Nil. 7 I 7 4	9 6	85 84	Warwick Wiltshire : Devizes Salisbury	Nil. 8 o 8 o		
Mansfield Newark Nottingham Retford	Nil. 8 3 8 0	9 0 	9 I 8 0	Swindon Warminster Worcestershire :	Nil. Nil. Nil.		·
Oxfordshire: Banbury	7 II 8 0	_	9 0	Worcester Yorkshire, E.R. :	<b>83</b>	80	8 10
Bicester Oxford Shropshire :	Nil. 7 10	—	<b>10</b> 10	Beverley Bridlington Driffield Hull	Nil. Nil. 76 Nil.	-	
Bridgnorth Market Drayton Oswestry Shrewsbury	Nil. 79 Nil. Nil.			Yorkshire, N.R. : Bedale Malton Northallerton	83 76 79		
Somersetshire: Bath Bridgwater Bristol Taunton	Nil. Nil. 82 Nil.	-	93	Scarborough Thirsk Yorkshire, W.R.:— Doncaster	8 o		8_4  8_10
Yeovil Staffordshire:	Nil. 8 o Nil. Nil.	. <del>.</del>	95	Goole Knaresborough Leeds Pontefract Ripon Selby Wakefield York	7 10 Nil. 7 9 7 8 8 1 7 8 		8_6 
Beccles Bungay Bury St. Edmunds Eye	Nil. Nil. 7 II Nil.		9 O	Anglesey: Llangefni	Nil.		
Framlingham Hadleigh Halesworth	7 8 Nil. Nil.	-	84	Caernarvonshire:	Nil.		
Haverhill Ipswich Saxmundham Stowmarket	8 4 8 0 Nil. Nil.	_	 	Denbighshire : Denbigh Wrexham	Nil. Nil.		
Sudbury Woodbridge	8 o Nil.	-	-	Glamorgan : Cardiff Montgomeryshire :	Nil.		
Surrey: Farnham Guildford Redhill	Nil. Nil. Nil.		-	Welshpool Pembrokeshire: Haverfordwest	Nil. 9 6	10 4	9.6

NOTE:—The above prices are based on returns received from Inspectors during the week ending 2nd July, 1938. They represent on the whole the average prices ruling in the week ending 25th June, 1938.

Ministry of Agriculture and Fisheries, Whitehall Place, S.W.I, 2nd July, 1938.

#### H.M. LAND REGISTRY.

The following land is about to be registered. Objections (if any) should be addressed to 'H.M. Land Registry, London, W.C.2," before the 19th day of July, 1938.

#### FREEHOLD.

- (1) Land in Newberry Gardens, Hornchurch, Essex, by M. Austin, 105, Station Road, Forest Gate, Essex.
- (2) Land to the south east of Burton to Repton Road, Repton, Derby, by The Derbyshire County Council.
- (3) Fenlands, 9, Chalgrove Road, Sutton, Surrey, by J. W. Hardy, 1175A, Finchley Road, Golders Green, N.W.11.
- (4) 5, Amersham Vale, New Cross, S.E.14, by M. M. Curry, 2, Essex House, Station Road, Barnes, S.W.13.
- (5) 3, Amersham Vale, New Cross, S.E.14, by A. Hills, 69, Marine Parade, Worthing, Sussex.
- (6) Seacroft, Rodmell Avenue, Saltdean, Sussex, by L. M. Marchini, 22, Clarendon Road, Harrow, Middlesex.
- (7) Land fronting a lane leading out of the road leading from Three Bridges to East Grinstead, Sussex, by J. W. W. Western, Little Ridge, Chislehurst Road, Chislehurst, Kent.
- (8) Land in Chalgrove Crescent, Ilford, Essex, by B. C. Tappin, 35, Lyndhurst Gardens, Newbury Park, Ilford, Essex.
- (9) Pevensey, Briarwood Drive, Eastwood, Southend-on-Sea, Essex, by F. A. Wilson of that address.
- (10) 9, Kingsway, Mortlake, Surrey, by H. A. Smith and E. J. Smith, both of 48, Munster Road, Fulham, S.W.6.
- (11) 6, Tower Grove, Walton-on-Thames, Surrey, by E. F. J. Plumridge, 4, Wandle Road, Wandsworth Common, S.W.18.
- (12) Land at the rear of The White Cottage, Seaville Drive, Pevensey Bay, Sussex, by
   S. R. V. Pazzi, The White Cottage, Seaville Drive, Pevensey Bay, Sussex.
- (13) 27, King Street, Aldeburgh, Suffolk, by M. E. Pawsey, Staneway Rayne, Braintree, Essex.
- (14) Land in The Ridgeway and Parkthorne Drive, North Harrow, Middlesex, by Garnetts Ltd.
- (15) 187, Booth Road, Hendon, N.W.9, by A. F. Nichols, 15, Oswestry Road, Oxford.
- (16) 1194, 1196, 1200 and 1202, London Road, Norbury, S.W.16, by H. H. H. Castle, Broxholme, 36, Sandilands, Croydon, Surrey.
- (17) The Old Cottage, Holyport, Berks, by M. M. Orr, Rose Cottage, Bray, Bucks.
- (18) St. Davids, Wood Ride, Petts Wood, Orpington, Kent, by J. M. R. Heaword, 35, Friern Barnet Lane, Finchley, N.II.
- (19) Ellington House, Ellington Road, Taplow, Bucks, by M. S. C. Lister, 115, Cromwell Road, Kensington, S.W.7.
- (20) 72, Belmont Avenue, Merton, Surrey, by Barclays Bank Ltd.
- (21) Sitzbaden, Foley Road, Claygate, Surrey, by S. M. Woolford, Dummer Down, Limmer Lane, Felpham, Sussex.
- (22) 16, Sandy Beach Estate, Eastoke, Hayling Island, Hants, by A. A. Barclay of that address.
- (23) Land in Church Road, Doddinghurst Hall Estate, Doddinghurst, Essex, by R. Walker, Suitsus, Ongar Road, Pilgrims Hatch, Nr. Brentwood, Essex.
- (24) 26 to 36 (even), Leythe Road, Acton, Middlesex, by M. H. Withers, 9, Beadon Road, Bromley, Kent.
- (25) 167, Higham Road, Tottenham, N.17, by W. C. Kennell of that address.
- (26) Mon Abri, The Ridgeway, Chatham, Kent, by M. E. Brampton, 9, New Road Avenue, Chatham, Kent.
- (27) Worthingholm, Bexhill, Sussex, by Bacobex Ltd.
- (28) The County Theatre, High Road, Hertford, Herts, by A. R. Shipman and S. S. King, both of Shell-Mex House, Strand, W.C.2.
- (29) Sudbury House, 167, Hammersmith Road, Hammersmith, W.6, by J. Lyons and Co. Ltd.
- (30) Land in Southgate Road and Ridgeway Road, Potters Bar, Middlesex, by The County Council of the Administrative County of Middlesex.
- (31) 26, Wellington Street, Strand, W.C.2, by J. H. Langdon of that address.
- (32) Land in Ditton Spur Road, Datchet, Bucks, by A. W. Barker, Tangiers Lodge, Vine Road, Barnes, S.W.13.

#### LEASEHOLD.

- (1) 26, Highview Road, Upper Norwood, S.E.19, by L. L. Kemp, 4, Avenue Chambers, Southampton Row, W.C.1,
- (2) 2nd 3rd 4th Floors and Basement of 80, St. Pauls Churchyard, E.C.4, by F. G. Williams, G. H. Williams, and L. Williams, all of 71, St. Pauls Churchyard, E.C.4.

J. S. STEWART-WALLACE, Chief Land Registrar.

### DISEASES OF ANIMALS ACTS, 1894 TO 1937, AND AGRICULTURE ACT OF 1937 (PART IV),

#### GREAT BRITAIN. MINISTRY OF AGRICULTURE AND FISHERIES.

-

۰.

### RETURN No. 12/1938. OUTBREAKS OF SCHEDULED (NOTIFIABLE) DISEASES WHICH HAVE BEEN CONFIRMED BY THE MINISTRY during

the period 16th to 30th JUNE, 1938.

Return of outbreaks of certain scheduled (notifiable) Diseases which have been confirmed by the Ministry during the period 16th to 30th June, 1938, and of the number of Infected Places under restrictions on 30th June.

The first figures under the respective headings indicate the number of Parishes in which were situated Infected Premises under restrictions on 30th June, and the number of Infected Premises under restrictions on that date; and the figures in brackets indicate the number of Parishes and Premises in which disease was declared during the period 16th to 30th June.

Counties (including		Fever. Suum).	Foot-and Disease Epizod	(Aphtæ	Anthrax (Anthrax).				
all Boroughs therein).*	Parishes.	Premises.	Devictor	Deservisore	Desister	Premises.	Animals	Attacked.	
	ransnes.		Parishes.	Premises.	Parishes.	r remises.	Cattle.	Other Animals.	
ENGLAND.									
Bedford Cambridge Isle of Ely Chester Derby Devon Essex Gloucester Hereford Hereford Hertford Huntingdon Huntingdon Kent Lancaster Lincoln, Holland Lincoln, Holland Lincoln, Holland Middlesex Norfolk Northampton Salop Somerset Southampton Southampton Southampton Southampton Suffolk Suffolk Surrey Sussex, East York, East Riding York, West Riding	$\begin{array}{c} \mathbf{I} & (\mathbf{I}) \\ 2 & (\mathbf{I}) \\ 2 & (\mathbf{I}) \\ 2 & (\mathbf{I}) \\ 19 & (6) \\ 19 & (7 & (2) \\ 19 & (7 & (1) \\ 19 & (1) \\ 10$	$\begin{array}{c} 2 & (2) \\ 2 & (1) \\ 2 & (-) \\ 7 & (2) \\ 28 & (7) \\ 7 & (1) \\ 10 & (2) \\ 6 & (1) \\ 1 & (-) \\ 4 & (1) \\ 1 & (-) \\ 7 & (-) \\ 12 & (3) \\ 1 & (-) \\ 3 & (-) \\ 1 & (-) \\ 3 & (-) \\ 1 & (-) \\ 3 & (-) \\ 1 & (-) \\ 3 & (-) \\ 1 & (-) \\ 3 & (-) \\ 1 & (-) \\ 3 & (-) \\ 1 & (-) \\ 3 & (-) \\ 1 & (-) \\ 3 & (-) \\ 1 & (-) \\ 3 & (-) \\ 1 & (-) \\ 3 & (-) \\ 1 & (-$	2 (-)  I (-) 2 (2)  I (-)  I (-)  I (-)    	2 (-)  I (-) 2 (2)  I (-)  I (-)  I (-)   	       	       		···· ··· ··· ··· ··· ··· ··· ··· ··· ·	

\* For convenience, in all returns, Berwick-upon-Tweed is considered to be in Northumberland, Stockport in Cheshire, and the city of London in the County of London.

DISEASES OF ANIMALS ACTS, 1894 to 1937, AND AGRICULTURE ACT OF 1937 (PART IV)—continued.

Counties (including			Fever Suum).	Foot-and Disease Epizod	(Aphta		Anthrax (Anthrax.)			
all B	oroughs erein).	·							Animals	Attacked.
		-	Parishes.	Premises.	Parishes.	Premises.	Parishes.	Premises.	Cattle.	Other Animals.
Monmout	ALES: h						- (I)	- (I)	I	
Aberdeen Angus Ayr Lanark Perth	  	· · · · · · · · · · · · ·	···· ··· ···	··· ··· ···	···· ···· ···	  	- (I) I (I) - (4) I (I) 2 (4)	- (I) I (I) - (4) I (I) 2 (4)	I 4 1 4	••••
TOTAL	•••	•••	143 (36)	166 (39)	8 (2)	8 (2)†	12 (21)	12 (21)	19	12
Position e 15th Ju	xisting ne, 193	on 8.	122	144	II	II	23 .	23	31	3

 $\dagger$  The approximate compensation payable in respect of animals slaughtered in connexion with these outbreaks is estimated at  $\pounds 47^{1}$ .

#### **PARASITIC MANGE.§**

Cou	nties (ir	cluding	all Borough	s therei	a).	Outbreaks confirmed.	Animals Attacked.
London	••••	ENG	LAND.		- · …	No. _	No. 1**
			TOTAL	• ••		 	I

§ Excluding outbreaks in Army Horses. \*\* This figure relates to a further animal attacked notified in connexion with outbreak previously reported.

. Cour	nties (in	Ontbreaks confirmed.						
		ENG	LAND	).				No.
Lincoln, Lindsey	•••	•••	•••	•••	•••	***		I
· ; · · ·		•		TOTAL	•••	•••		I

#### SHEEP-SCAB.

4348

#### DISEASES OF ANIMALS ACTS, 1894 TO 1937, AND AGRICULTURE ACT OF 1937 (PART IV)—continued.

SUMMARY OF RETURNS.

	Anthrax.			t-and-Mouth Para Disease. Man			Sheep Scab.	Swine Fever.	
. Period.	Outbreaks confirmed.	Animals attacked.	Outbreaks confirmed.	A n i m a l s slaughter e d as diseased or exposed to infection.	Outbreaks Confirmed	Animals attacked.	Outbreaks confirmed.	Outbreaks confirmed.	Swine slaughtered.
Period 16th to 30th June, 1938 Corresponding period in 1936 1935	No. 21 27 5 9	No. 31 29 5 16	No. 2 1 	No. 38 131 — 97	No.  3 1 3	No. 1 3 2 10	No. 1 5 7 5	No. 39 51 81 112	No. 34 39 35 88
Total 1st Jan. to 30th June, 1938           Corresponding period in         1937           1936         1935	481 411 258 209	533 472 305 240	155 6 4 34	19,939 671 379 10,208	45 85 70 70	74 122 207 154	103 127 144 191	300 620 1,071 973	195 396 723 510

Norg.—The figures for the current year are approximate only. § Excluding outbreaks in Army Horses.

Note.—The following diseases were eradicated in Great Britain in the years indicated :— Cattle plague, or rinder-pest (pestis bovina) 1877; pleuro-pneumonia (peripneumonia contagiosa bovum) 1898; sheep pox (variola ovium) 1850; rabies 1922; epizootic lymphangitis (lymphangitis epizootica) 1906; and glanders (including farcy) (malleus) 1928. Dourine (exanthema coitale paralyticum) has never existed in Great Britain.

No case of Fowl Pest (*pestis avium*) has been confirmed in Great Britain since the 11th January, 1937, the date as from which the disease was first notifiable. No case of Fowl Pest has been known to exist in Great Britain since 1933.

#### TUBERCULOSIS (ATTESTED HERDS) SCHEMES.

The number of Attested Herds, i.e. herds officially certified as free from tuberculosis as at 30th June, 1938, was as follows:--

England.

785

WALES. 1.104 SCOTLAND. 801 TOTAL (GREAT BRITAIN) 2,690

#### TUBERCULOSIS ORDER OF 1938.

Number of Bovine Animals in respect of which Notices of Intended Slaughter have been issued by the Ministry of Agriculture and Fisheries under the Tuberculosis Order of 1938 during the period 16th to 30th June, 1938.

Counties (including all Boroughs therein).				Number of Bovine Animals.	. Counties (including all therein).	hs	Number of Bovine Animals.		
ENGLAND.					ENGLAND—continued.				
Berks	•••	•••		10	Northampton	•••		4	
Cambridge	•••	•••		2	Northumberland	•••		5	
Chester	•••	•••		55	Nottingham	•••		20	
Cornwall	•••	•••		15	Oxford	•••	: •••	12	
Cumberland	•••	•••		5 18	Salop	•••	•••	46	
Derby	•••		••••	18	Somerset	•••		12	
Devon	•••		÷	6	Southampton	•••		9	
Dorset	•••	•••	•••	9	Isle of Wight		:	í	
Durham	•••	•••		17	Stafford			77	
Essex	•••	•••		33	Suffolk			17	
Gloucester		•••		2	Surrey			13	
Hereford		•••		I	Sussex, East			7	
Hertford		•••		3	Sussex, West			15	
Huntingdon .	•••	•••		4	Warwick	•••		. 20	
Kent				25	Wilts			28	
Lancaster				30	Worcester			5	
Leicester				ĨI	Yorks, East Riding			4 4	
Lincoln, Holland				3	Yorks, North Riding			19	
Lincoln, Kestever	n			3 6	Yorks, West Riding	•••		58	
Lincoln, Lindsey.				33	-,				
Middlesex .	•••	•••		3	Total			679	
Norfolk	•••	•••		ığ		•			

DISEASES OF ANIMALS ACTS, 1894 TO 1937 AND AGRICULTURE ACT OF 1937 (PART IV)-continued.

Counțies (incl t	uding all herein).	Borou	ġhs	Number of Bovine Animals.	Counties (including all Boroughs therein).	Number of Bovine Animals.
· · ·						
- W	ALES.				SCOTLAND—continued.	
Anglesey	3	·		. I	Banff	4
Brecon	•••			2	Caithness	i
Caernarvon			:	2	Clackmannan	I
Cardigan	•••		]	12	Dumfries	3
Carmarthen	•••		[	2	Dunbarton	2
Denbigh				· 6	Fife	5 .
Flint	•••			9	Inverness	I.
<b>Glamorg</b> an		•••		IÓ	Kincardine	. 2
Monmouth	•••			6	Lanark	10
Montgomery	•••			2	Midlothian	2
Pembroke				2	Moray and Nairn	2
• •			· [	·····	Perth and Kinross	· 5
	TOTAL			· 54	Renfrew	9
				. ,	Ross & Cromarty	í
					Roxburgh	í ī
	•			-	Stirling	. 4
					West Lothian	
			1		Wigtown	7
		•		•	-	1
SCO	TLAN	D.	Í			
Aberdeen				18	<b>TOTAL</b>	86
Angus				5		· · · · · · · · · · · · · · · · · · ·
Argyll	•••			2	TOTAL (GREAT BRITAIN)	819

#### TUBERCULOSIS ORDER OF 1938-continued.

#### SHEEP SCAB MOVEMENT AND DOUBLE-DIPPING AREAS.

The following Areas are "Movement Areas" for the purposes of Part II of the Sheep Scab Order of 1938:-

Inverness.—An Area comprising :---

The Islands known as the Outer Hebrides (except the Islands of St. Kilda, Dune, Soay, and Boreray; and except the parish of North Uist and all the Islands therein, together with the island of Berneray in the parish of Harris and all the grazing islands attached thereto and the Island of Barra and the adjacent islets attached to the holdings of Barra), in the County of Inverness.

Ross and Cromarty .- An Area comprising :----+

The Island of Lewis, including the smaller islands adjacent thereto in the County of Ross and Cromarty.

The movement of sheep out of the following area is regulated by the North Uist (Movement of Sheep) Order of 1934.

An area in the county of Inverness comprising the parish of North Uist and all the Islands therein, together with the island of Berneray in the Parish of Harris and all the grazing islands attached thereto. 3,4

้ง รูเรีย เ

Ministry of Agriculture and Fisheries. 5th July, 1938.

•••

NOTICE is hereby given that the Place of Meeting for religious worship described as METHODIST CHURCH, situated at East Lane, North Wembley in the civil parish of Wembley in the registration district of Hendon in the county of Middlesex which was duly certified for worship on the roth day of April, 1922 has wholly ceased to be used as a Place of Meeting for religious worship by the cona Place of Meeting for religious worship by the con-gregation on whose behalf it was so certified, and that the Registrar-General has caused the record of the certification thereof to be cancelled pursuant to the Act 18 and 19 Vict., cap. 81 from the 1st day of July 1938.—Witness my hand this 1st day of July 1938.

A. W. MUNDY, Registrar-General of Births, Deaths and Marriages in England and Wales.

NOTICE is hereby given that the Building for-merly known as EBENEZER CHAPEL, situated at merry known as EBERGER CHAPEL, situated at Tower Place, in the civil parish of Shrewsbury in the registration district of Shrewsbury in the county of Salop which was duly registered for marriages pursuant to the Act 6 and 7 Will IV, cap. 85 is pursuant to the Act 6 and 7 Will. IV, cap. 85 is now no longer used as a Place of Meeting for reli-gious worship by the congregation on whose behalf it was so registered, and that the registry thereof was therefore on the 27th day of June 1938 formally cancelled by the Registrar-General of Births, Deaths and Marriages for England and Wales.—Witness my hand this acth day of June 1938 hand this 30th day of June 1938.

E. P. EVEREST, Superintendent Registrar.

#### In the High Court of Justice (Chancery Division) .--- Companies Court.

Mr. Justice Simonds.

#### No. 00462 of 1938.

the Matter of MARITZA PRODUCTIONS Limited, and in the Matter of the Companies In Act, 1929.

Act, 1929. NOTICE is hereby given, that a petition for the winding-up of the above named Company by the High Court of Justice was on the thirtieth day of June 1938, presented to the said Court by Lucie Mannheim of 4. Blenheim Road, St. Johns Wood in the county of London, Spinster, a creditor of the said Company, and that the said petition is directed to be heard before the Court sitting at the Royal Courts of Justice, Strand, London, on the eighteenth day of July 1938 and any creditor or contributory of the said Company desirous to sup-port or oppose the making of an Order on the said petition may appear at the time of hearing, in per-son or by his Counsel, for that purpose, and a copy of the petition will be furnished to any creditor or contributory of the said Company requiring the same by the undersigned, on payment of the regu-lated charge for the same. GEORGE A. HERBERT, 10, Old Square,

GEORGE A. HERBERT, 10, Old Square, Lincoln's Inn, W.C.2, Solicitor for the Petitioner.

Nore.—Any person who intends to appear on the hearing of the said petition must serve on, or send by post to, the above named, notice in writing of his intention so to do. The notice must state the name and address of the person, or if a firm the name and address of the firm, and must be signed by the person or firm or his or their Soli-citor (if any), and must be served, or if posted, must be sent by post, in sufficient time to reach the above named not later than one o'clock in the afternoon of the sixteenth day of July 1938. (153) (153)

In the High Court of Justice (Chancery Division). Companies Court, Manchester District Registry.

#### Mr. Justice Simonds. No. 003 of 1938.

In the Matter of the SICK AND BURIAL SOCIETY connected with the ROYAL GEORGE and HIGH GROVE MILLS, SADDLEWORTH, and in the Matter of the Companies Act, 1929.

NOTICE is hereby given that a petition for the winding-up of the above named Society by the

High Court of Justice was on the 28th day of June 1938 presented to the said Court by Frank Mills of 3, Brookside, Grotton, within Saddleworth in the county of York, Dryer, John Jackson of Stores Cottages, Grasscroft Greenfield within Saddleworth aforesaid, Foreman Miller Finisher and Carl Shaw of Armit Cottages, Royal George, Greenfield within Saddleworth aforesaid Hardener. And that the said petition is directed to be heard before the Court sitting at the Royal Courts of Justice, Strand, London W.C. on the 18th day of July 1938, and any creditor of the said Society desirous to support or oppose the making of an Order on the said petition may appear at the time of hearing in per-son or by his Counsel for that purpose, and a copy of the said Society requiring the same by the under-signed on payment of the regulated charge for the same.—Dated this 1st day of July 1938. ROWNTREE and RITSON, 11, Church Ter-

ROWNTREE and RITSON, 11, Church Ter-race, Oldham, Solicitors for the Petitioners.

Note.—Any person who intends to appear on the hearing of the said petition must serve on or send by post to the above named Rowntree & Ritson notice in writing of his intention so to do. The notice must state the name and address of the person, or, if a firm, the name and address of the firm, and must be signed by the person or firm, or his or their Solicitors if any, and must be served, or, if posted must be sent by post in sufficient time to reach the above named not later than one o'clock in the afternoon of the r6th day of July roa8. in the afternoon of the 16th day of July 1938. (051)

#### In the Brighton and Lewes County Court -(At Brighton).

#### No. 2 of 1938.

In the Matter of the Companies Act, 1929, and in the Matter of LENTRIDGE FARM DIRECT SUPPLY Limited, and in the Matter of the Peti-tion of ROBERT GUNN MACKAY, dated the 22nd day of June, 1938.

NOTICE is hereby given that a petition for the winding-up of the above named Company by the County Court of Sussex holden at Brighton and Lewes (at Brighton) was on the 22nd day of June 1938 presented to the said Court by Robert Gunn Mackay of Wantley Cottage, Bolney in the county of Sussex a shareholder and a Member of Lentridge Farm Direct Supply Limited. And that the said petition is directed to be heard before the Court sitting at the Court House. Church Street, Brighton sitting at the Court House, Church Street, Brighton in the county of Sussex on the 21st day of July 1938 at the hour of twelve o'clock noon and any creditor or contributory of the said Company desirous to support or contributory of the said Company desitous to support or oppose the making of an Order on the said petition may appear at the time of hearing in person or by his Solicitor or Counsel for that pur-pose; and a copy of the petition will be furnished to any creditor or contributory of the said Company requiring the same by the undersigned on payment of the semidad charge for the same Datid this of the regulated charge for the same .--- Dated this 29th day of June 1938.

GATES McCULLY and BUCKWELL, Regent House, Princes Place, North Street, Brighton I, Solicitors for the Petitioner.

The address for service in London is c/o Messrs. Jennings Son & Ash, of 69, Leadenhall Street, London, E.C.3.

NOTE.—Any person who intends to appear on the hearing of the said petition must serve on or send by post to the above named petitioner's Solicitors of Regent House, Princes Place, North Street, Brighton their London agents, Messrs. Jennings Son & Ash of 69 Leadenhall Street, London E.C.3, notice in writing of his intention so to do. The notice must state the name and address of the person, or, if a firm, the name and address of the firm and must be signed by the person or firm or his or their be signed by the person or firm or his or their Solicitor (if any) and must be served or if posted must be sent by post in sufficient time to reach the above named not later than 6 o'clock in the afternoon of the 20th day of July 1938. (011)

In the High Court of Justice.-Chancery Division. Mr. Justice Simonds.

, No. 00345 of 1938.

In the Matter of BRITISH MATRIX panies Act, 1929

panies Act, 1929. NOTICE is hereby given that the Order of the High Court of Justice, Chancery Division, dated the 20th day of June 1938, confirming the reduction of the capital of the above named Company, from  $\sharp_{150,000}$  to  $\sharp_{33,750}$  and the Minute approved by the Court showing with respect to the share capital of the Company as altered, the several particulars required by the above mentioned Act, were regis-tered by the Registrar of Companies on the 29th day of June 1938.—Dated this 4th day of July 1038. 1938.

NICHOLSON GRAHAM and JONES, 19-21, Moorgate, London, E.C.2, Solicitors for the (052) said Company.

In the High Court of Justice.-Chancery Division. Mr. Justice Simonds.

#### No. 00449 of 1938.

In the Matter of The PRINTING MACHINERY COMPANY Limited, and in the Matter of the Companies Act, 1929.

NOTICE is hereby given that a petition for the confirmation of the reduction of the capital of the commution of the reduction of the capital of the above named Company from £250,000 to £250 was on the 27th day of June 1938 presented to His Majesty's High Court of Justice and is now pending. And that by an Order dated the 4th July 1938 an inquiry was directed as to the debts, claims and liabilities of the said Company on the 4th July 1938. A list of the persons admitted to have been creditors of the Company on the 4th July 1938 have been creditors A list of the persons admitted to have been creditors of the Company on the 4th July 1938 may be inspected at the offices of the above named Company at 9 Kingsway, London, W.C.2, or at the offices of Messrs. Hays Roughton & Dunn (the Solicitors of the Company) at 110 Cannon Street, London, E.C.4 at any time during usual business hours, on payment of the charge of one shilling. Any person who claims to have been on the 4th July 1938 and still to be a to have been on the 4th July 1938 and still to be a creditor of the said Company and who is not entered on the said list and claims to be so entered, must, on or before the 13th July 1938 send in his name and address, and the particulars of his claim and the name and address of his Solicitor (if any) to Mr. Stanley Saul Spurling a Member of the said firm of Messrs. Hays Roughton & Dunn at the address mentioned helps of in default thereof he address mentioned below or in default thereof he will be precluded from objecting to the proposed reduction of capital.—Dated this 5th day of July, 1938.

HAYS, ROUGHTON and DUNN, 110, Cannon Street, London, E.C.4, Solicitors for the above (213) named Company.

#### MARCUS INVESTMENT (MITRE) Limited.

AT an Extraordinary General Meeting of the Mem-Al an Extraordinary General Meeting of the Mem-bers of the said Company, duly convened, and held at 93, Park Lane in the county of London on the 3oth day of June 1938, the following Special Resolu-tion was duly passed: — "That the Company be wound up voluntarily and that Mr. Albert Edward Lauer of 93, Park Lane, W.I, be appointed Liquidator for the purpose of such winding-up."

(154) S. L. L. MARCUS, Chairman.

#### GEORGE LUNT (STOCKPORT) Limited.

AT an Extraordinary General Meeting of the Com-AT an Extraordinary General Meeting of the com-pany held at the offices of Messrs. Harrop Marshall & Co. of 18 Union Road, Underbank, Stockport, on Thursday the 23rd day of June 1938, the following Resolution was duly passed as an Extraordinary Resolution :

"That it has been proved to the satisfaction of this Meeting that the Company cannot by reason of its liabilities continue its business and it is advisable

to wind up the same and accordingly that the Com-pany be wound up voluntarily and that Mr. Leon Douglas Birkett of 18 Union Road, Underbank, Stockport, Chartered Accountant, be and he is hereby appointed Liquidator for that purpose." At the Statutory Meeting of creditors duly con-vened and held on the same day and at the same place subsequent to the above mentioned Meeting the appointment of the said Leon Douglas Birkett as Liquidator was confirmed without a Committee of

as Liquidator was confirmed without a Committee of Inspection.

(163) L. D. BIRKETT, Chairman.

#### Special Resolution of the LEATHERIES Limited. The Companies Act, 1929.

AT an Extraordinary General Meeting of the above named Company duly convened and held at 6 Bennetts Hill, Birmingham, on the 30th day of June, 1938, the following Resolution was duly passed as a Special Resolution viz.:—

Resolution.

Kesolution. "That the Company be wound up voluntarily and that Edward Bosley of 120 Colmore Row, Birming-ham, Chartered Accountant, be and he is hereby appointed Liquidator for the purpose of such wind-ing-up." Dated this 30th day of June, 1938.

H. B. YATES, Chairman of Meeting. (132)

## Special Resolution of ASSOCIATED ENAMELLERS Limited.

#### The Companies Act, 1929.

AT an Extraordinary General Meeting of the above named Company duly convened and held at the Queen's Hotel, Birmingham, on the 1st day of July 1938 the following Resolution was duly passed as a Special Resolution: —

#### Resolution.

"That the Company be wound up voluntarily and that William Henry Newton of Union Chambers, Temple Row, Birmingham, Chartered Accountant and Percy John Chaplin of 8, Staple Inn, London, W.C.I, Chartered Accountant be and they are hereby appointed Liquidators for the purpose of such winding-up." Dated this 1st day of July 1938.

R. STANTON BAUGH, Chairman of Meeting. (131)

#### ROCOCO (PERFUMERY) Limited.

Extraordinary Resolution (pursuant to Section 225 (c) of the Companies Act, 1929), passed 29th June, 1938.

AT an Extraordinary General Meeting of the Mem-bers of the said Company, duly convened and held at 70, Bartholomew Close, London, E.C.1, on the 29th day of June 1938 the following Extraordinary

29th day of June 1938 the following Extraordinary Resolution was duly passed:— "That it has been proved to the satisfaction of the Company that this Company cannot by reason of its liabilities continue its business, and that it is advisable that the same should be wound up; and that the Company be wound up accordingly. That Mr. Kenneth Miller Scott, of 32, Finsbury Square, London, E.C.2, Chartered Accountant, be and he is bereby appointed the Liquidator of the Company." hereby appointed the Liquidator of the Company." J. H. KRAATZ, Chairman. (127)

#### The GLEDHILL WEAVING MACHINE CO. Limited.

#### The Companies Act, 1929.

up." Dated this 28th day of June 1938.

GEO. C. HIRST, Chairman. (132)

#### SILVER KING FOREIGN PATENTS SYNDICATE Limited.

The Companies Act, 1929.

AT an Extraordinary General Meeting of the Mem-AT an Extraordinary General Meeting of the Mem-bers of the above named Company, duly convened and held at the offices of Messrs. Glaisyer Porter & Mason, Solicitors, 126 Colmore Row in the city of Birmingham, on the 29th day of June 1938 the following Special Resolution was duly passed:— "That the Company be wound up voluntarily, and that Frank Garfield Stephens of 115 Colmore Row in the city of Birmingham, Chartered Account-ant be appointed Liquidator of the Company."

CORNELIUS CHAMBERS, Chairman. (120)

#### WRIGHT & TURNER (SUCCESSORS) Ltd. The Companies Act, 1929.

AT an Extraordinary General Meeting of the Mem-bers of the above named Company, duly convened and held at the offices of Messrs. Massey & Ellison, Civic House, Great Charles Street, Birmingham 3 on the 28th day of June 1938 the following Special Resolution was duly passed:— "That the Company be wound up voluntarily, and that Mr. John Ellison of Civic House, Great Charles Street, Birmingham 3, Chartered Account-ant, be appointed Liquidator of the Company."

GEORGE J. BROWN, Chairman. (130)

#### SLATTER & STOPFORTH Limited.

AT an Extraordinary General Meeting of the Members of the above named Company, duly con-vened pursuant to the provisions of section 117 (2) of the Companies Act, 1929, and held at Rodney Chambers, Rodney Road, Cheltenham, on the 27th day of June 1938 the following Special Resolutions were duly passed :—

day of June 1938 the following Special Resolutions were duly passed:— "That owing to the Company being unable to pay its debts, it be wound up forthwith." "That Mr. Charles Henry Tortonese, Certified Accountant, of Rodney Road, Cheltenham, be nominated by the Members as the Liquidator." ARTHUR W. BONE, Chairman. (117)

#### YEWHURST TRUST Limited.

Special Resolution (pursuant to Sections 117 (2) and 225 (1) (b) of the Companies Act, 1929), passed 30th June, 1938.

AT an Extraordinary General Meeting of the above named Company, duly convened and held at 1, Oxford Court, Cannon Street, London, E.C.4, on the 30th day of June, 1938, the subjoined Special Resolution was duly passed, viz.:--

#### Resolution.

"That the Company be wound up voluntarily and that Mr. Duncan McKellar, Chartered Accountant, of 1, Oxford Court, Cannon Street, London, E.C.4, be and he is hereby appointed Liquidator for the purpose of such winding-up."

G. BRADLEY, Chairman of the Meeting. (172)

#### SOUTH WALES HOUSE & INVESTMENT CO. Ltd.

AT an Extraordinary General Meeting of the Members of the above named Company duly convened and held at Church Street Chambers, Pontypridd on the 29th day of June 1938, the following Special

Resolution was passed:— "That the Company be wound up voluntarily and that Mr. Percy Augustus Hayes of No. 12 Pembroke Terrace, Cardiff, Incorporated Accountant, be hereby appointed Liquidator of the Company.'

THOS. JENKINS, Chairman. (118)

#### SURREY WHOLESALE ELECTRICAL COMPANY Limited.

#### The Companies Act, 1929.

Special Resolution passed the 30th day of June, 1938.

AT an Extraordinary General Meeting of Surrey Wholesale Electrical Company Limited duly con-vened and held at 62 Great Russell Street, London, W.C.1, on Thursday the 30th day of June, 1938 at

Liquidator for the purposes of such winding-up." (180)

F. D. PRATT, Chairman,

#### The BLAENAVON GAS & WATER COMPANY Limited.

#### (Members' Voluntary Winding-up.)

AT an Extraordinary General Meeting of the Mem-AT an Extraordinary General Meeting of the Mem-bers of the above named Company, duly convened, and held at the Company's offices, Avon Road, Blaenavon in the county of Monmouth, on the 30th day of June, 1938, the following Special Resolution was duly passed:— "That the Company be wound up voluntarily and that Mr. Francis Charles White of 'Beechwood' Blaenavon be and is hereby appointed Liquidator for the purpose of such winding-up."

(076)

ALFRED H. JAMES.

#### DUFFRYN CINEMA (NEW TREDEGAR) Limited.

#### (Members' Voluntary Winding-up.)

The Companies Act, 1929.

#### D. J. TREASURE, Chairman. (073)

#### **KEESING'S MEDICAL DIGEST Limited.**

(Members' Voluntary Winding-up.)

The Companies Act, 1929.

AT an Extraordinary General Meeting of the above named Company, duly convened, and held on the 28th day of May, 1938 the following Special Resolu-tion was duly passed:— "That the Company be wound up voluntarily under the provision of the Companies Act, 1929, and that Mr. L. Fles of Amsterdam be and he is hereby appointed Liquidator for the purposes of such winding-up."

L. FLES, Chairman.

#### The LADY BAY SOCIAL CLUB Limited.

(Members' Voluntary Winding-up.)

(074)

#### The Companies Act, 1929.

AT an Extraordinary General Meeting of the above named Company duly convened at short notice with the consent of all the Members, and held at the offices

the consent of all the Members, and held at the omcess of Messrs. Boaler & Flint, 5, Regent Street, Notting-ham on the 1st day of July 1938, the following Special Resolution was duly passed:— "That the Company be wound up voluntarily, and that Mr. Frank Walter Flint, Incorporated Accountant of 5 Regent Street, Nottingham, be and is hereby appointed Liquidator for the purpose of such winding-up." such winding-up.

#### E. SOMERFIELD, Chairman. (075)

The BLYTH PORCELAIN COMPANY Limited.

The Companies Act, 1929.

AT an Extraordinary General Meeting of the Mem-bers of the above named Company, duly held at 17, Albion Street, Hanley, in the city of Stoke-on-Trent, on the 30th day of June, 1938, the following Special Resolution was duly passed:— "That the Company be wound up voluntarily, and that Richard Ecroyd Clark of 17, Albion Street, Hanley, in the city of Stoke-on-Trent, Chartered Accountant, be and he is hereby appointed Liquida-tor of the Company."

H. VICTOR FORESTER, Chairman. (071)

(064)

• •

(028)<sup>.</sup>

(012)

SOUTH LONDON FACTORS Limited.

Extraordinary Resolution (pursuant to Sections 117 (1) and 225 (1) (c) of the Companies Act, 1929), passed 30th June, 1938.

AT an Extraordinary General Meeting of the above named Company, duly convened, and held at 7, Warwick Court, Gray's Inn, W.C.I, on the 30th day of June, 1938, the subjoined Extraordinary Resolution was duly passed, viz.:—

#### Resolution.

Resolution. "That it has been proved to the satisfaction of this Meeting that the Company cannot by reason of its liabilities continue its business, and that it is advisable to wind up the same, and accordingly that the Company be wound up voluntarily, and that Henry Charles Treby of 7, Warwick Court, Gray's Inn, W.C.I, be and he is hereby appointed Liquidator for the purposes of such winding-up."

CHAS. E. COOPER, Secretary. (162)

## Resolution of MARGATE SCENIC RAILWAY Limited. (Members' Voluntary Liquidation.)

The Companies Act, 1929.

Line Companies Act, 1929. AT a General Meeting of the Members of the above named Company, duly convened and held at 4, Charterhouse Square, London, E.C.1, on Thursday, the 23rd June, 1938, the following Resolution was passed:— "That the resignation of Mr. Ivor George Milne Iles be and is hereby accepted, and that Mr. Ronald Frederick Bendall, A.C.A., of 4, Charterhouse Square, London, E.C.1, be and he is hereby appointed Liquidator of the Company in place of Mr. Ivor George Milne Iles."

H. E. WEIGHT, Chairman. (161)

#### GEORGE SKEY & COMPANY Limited.

Special and Extraordinary Resolutions.

AT an Extraordinary General Meeting of the Members of the above Company, duly convened and held at 28, Lambeth High Street, on Thursday, 30th June, 1938, the undermentioned Resolutions were passed as a Special and an Extraordinary Resolution respectively:

1. Special Resolution.

"That the Company be wound up voluntarily and that Mr. Edward Basil Green, of 28, Lambeth High Street, London, S.E.I, be, and is hereby, appointed Liquidator thereof."

2. Extraordinary Resolution. "That the said Liquidator be, and he is hereby, authorised to distribute all or any of the assets of the Company in specie."

Dated this 30th day of June, 1938.

ROBERT WOOD, Chairman. (119)

#### A.C.C. HOLDINGS Limited.

#### The Companies Act, 1929.

Special Resolution passed 1st July, 1938.

AT an Extraordinary General Meeting of the AT an Extraordinary General Meeting of the Members of the above named Company, duly con-vened and held at 70, Billiter Buildings, Billiter Street, London, E.C.3, on Friday, the 1st day of July, 1938, the following Resolution was duly passed, as a Special Resolution:— "That the Company be wound up voluntarily and that Mr. John Morison, Chartered Accountant, of I, Oxford Court, Cannon Street, London, E.C.4, be and is hereby appointed Liquidator for the purposes

and is hereby appointed Liquidator for the purposes of such winding-up."

Dated, this 1st day of July, 1938.

D. JONES, Chairman of the Meeting. (173)

### NYMAN & COOPER Limited.

#### The Companies Act, 1929.

AT an Extraordinary General Meeting of the share-holders of the Company, duly convened and held at 9, Princes Park Avenue, Golders Green, N.W.II, on Tuesday the 28th day of June 1938 at 10 o'clock in the forenoon the following Resolution was passed as an Extraordinary Resolution: — "That this Company cannot by reason of its liabilities continue its business and it is advisable that the same be wound up and accordingly that the

that the same be wound up and accordingly that the

Company be wound up voluntarily and that Mr. John Bennett Rubens of 329, High Holborn, W.C. Accountant be, and he is hereby appointed Liquida-tor for the purposes of such winding-up." At a Meeting of the creditors of the Company held pursuant to sections 238, 239, and 240 of the Com-panies Act 1929 on the same day at the Holborn Restaurant, Kingsway W.C. at 12 o'clock noon it was resolved that the Resolution be confirmed and adopted. adopted.

(063) ESTHER NYMAN, Chairman.

#### DIGBY TOBIAS DYEING WORKS Limited. The Companies Act, 1929.

AT an Extraordinary General Meeting of the Members of the above named Company, duly convened, and held at "Chiswell House," 133-139, Finsbury Pavement, London, E.C.2, on the 30th day of June, 1938, the following Special Resolution was duly

S. TOBIAS, Chairman.

#### VOYSEY & HUNT Limited.

#### (Members' Voluntary Winding-up.)

AT an Extraordinary General Meeting of the Mem-

AT an Extraordinary General Meeting of the Mem-bers of the above named Company, duly convened, and held at 107 Poole Road, Westbourne, Bourne-mouth, in the county of Hants, on the 2nd day of July, 1938, the following Special Resolution was duly passed:— "That it is desirable to reconstruct the Company, and accordingly that the Company be wound up voluntarily, and that Messrs. Reginald Frank Pounsett and Frederick Arthur Voysey be and they are hereby appointed Liquidators for the purposes of such winding-up." (072) W T HUNT Chairman

W. T. HUNT, Chairman. (072)

#### HARVEY BROTHERS SUCCESSORS Limited.

AT an Extraordinary General Meeting of the Members of the above named Company duly con-vened and held at the offices of Messrs. Handley, Wilde & Charlton, Bank of England Chambers, Tib Lane, Manchester, on the 30th day of June 1938 the following Extraordinary Resolution was duly passed to the second s

the following Extraordinary Resolution was duly passed:— "That it has been proved to the satisfaction of this Meeting that the Company cannot by reason of its liabilities continue its business and that it is advisable to wind up the same, and accordingly that the Company be wound up voluntarily and that Mr. William Patrick Masterson of Bank of England Chambers, Tib Lane, Manchester, Chartered Accountant be and be is bereby nominated Liquida. Accountant, be and he is hereby nominated Liquida-tor for the purpose of such wind-up."

E. M. MORDY, Chairman.

#### VULCAN CLOTHING CO. Ltd.

AT an Extraordinary General Meeting of the Members of the above named Company held at the offices of Hollings Crowe Storr & Co., 39, Park Square, Leeds I, on Monday, the 27th day of June 1938, the following Extraordinary Resolution was passed:— "That it has been proved to the satisfaction of this Meeting that the Company cannot, by reason of its liabilities, continue its business, and that it is advisable to wind up the same and accordingly that

or its national continue its business, and that it is advisable to wind up the same and accordingly that the Company be wound up voluntarily and that Mr. William Ewart Storr of Hollings Crowe Storr & Co., 39, Park Square, Leeds I, be and he is hereby nominated Liquidator for the purpose of such winding-up." At a Meeting of the creditors of the above named Company, subsequently held on the same

At a Meeting of the creditors of the above named Company, subsequently held on the same day and at the Law Institute, Albion Place, Leeds, the appointment of the said William Ewart Storr as Liquidator was confirmed.—Dated this the 30th day of June 1938.

F. PASHLEY, Chairman.

(054)

#### Messrs. W. T. BEDDOWS & COMPANY Limited. The Companies Act, 1929.

Special Resolution passed 1st July, 1938.

AT an Extraordinary General Meeting of the Members of the above named Company, duly con-vened and held at 43 St. Mary Axe, London E.C.3 on Friday the 1st day of July 1938 the following Resolution was duly passed as a Special Resolution:

(188)

E. CRAWLEY, Chairman.

#### M.X.H. Limited.

#### The Companies Act, 1929.

#### Special Resolution passed 19th April, 1938.

AT an Extraordinary General Meeting of the Members of the above named Company, duly con-vened and held at Carnaby House, Beak Street, Regent Street, London, W.I, on Tuesday, the 19th day of April, 1938, the following Special Resolution

day of April, 1938, the following Special Resolution was duly passed:— "That the Company be wound up voluntarily as a Members' voluntary winding-up, and that Mr. John Bennett Rubens, Chartered Accountant, of 329, High Holborn, London, W.C.I, be appointed Liquidator for the purpose of such winding-up." All the Members of the Company being present in person, it was resolved that the necessary notice under section 117 of the Companies Act, 1929, be waived.

waived.

(006)

#### N. BRENNER, Chairman.

#### FUMASIL COMPANY Limited.

Special Resolution (pursuant to ss. 117 (2) and 225 (1) (b) of the Companies Act, 1929) passed 1st July, 1938.

AT an Extraordinary General Meeting of the above named Company, duly convened, and held at 31-2 King Street, Cheapside, London, E.C.2, on the first day of July, 1938, the subjoined Special Resolution was duly passed, viz.:—

#### Resolution.

"That the Company be wound up voluntarily, and that Charles Dalrymple Gairdner, of 31-2 King Street, Cheapside, in the city of London, Chartered Accountant, be and he is hereby appointed Liquidator for the purposes of such winding-up." G. F. H. FAITHFULL, Chairman. (008)

#### BRUTON CLUB Limited.

AT an Extraordinary General Meeting of the Members of the above named Company, duly con-vened at short notice with the consent in writing of all the Members, and held at 27, Ely Place London, E.C.r, on the 30th day of June, 1938, the following Special Resolution was duly passed:— "That the Company be wound up voluntarily and that Jesse Samuel Barrington of Budleigh House, Frinton-on-Sea, Essex, be appointed Liqui-dator for the purposes of such winding-up." J. S. BARRINGTON, Chairman. (044)

#### KATE ROSE Limited.

#### (Creditors' Voluntary Winding-up.)

#### The Companies Act, 1929.

Extraordinary Resolution, passed 30th day of June, 1938.

AT an Extraordinary General Meeting of the Members of Kate Rose Limited duly convened and held at 4 London Wall Avenue London E.C.2 on Thursday the 30th day of June 1938 the following Resolution was duly passed as an Extraordinary Resolution:

"That it has been proved to the satisfaction of this Meeting that the Company cannot by reason of its liabilities continue its business and that it is advisable to wind up the same and accordingly that the Company be wound up voluntarily." And at the same Meeting Hobart Harold de Courcy Moore of 4 London Wall Avenue, London E.C.2

Chartered Accountant was nominated Liquidator for the purposes of such winding-up. At the Meet-ing of the creditors of the above named Company duly convened pursuant to section 238 of the Com-panies Act 1929 and held at 4 London Wall Avenue London E.C.2 on Thursday the 30th day of June 1938 the creditors resolved "That Hobart Harold de Courcy Moore of 4 London Wall Avenue London E.C.2 Chartered Accountant be and he is hereby appointed Liquidator for the purposes of hereby appointed Liquidator for the purposes of such winding-up."

#### J. H. HARRIS, Chairman.

#### LADIES FASHIONS Limited. The Companies Act 1929.

AT an Extraordinary General Meeting of the shareholders of the Company, duly convened and held at 9, Princes Park Avenue, Golders Green, N.W.11, on Wednesday the 29th day of June 1938 at 10 o'clock in the forenoon the following Resolu-tion was passed as an Extraordinary Resolution:— "That this Company cannot by reason of its liabilities continue its business and it is advisable that the same be wound up and accordingly that

liabilities continue its business and it is advisable that the same be wound up and accordingly that the Company be wound up voluntarily and that Mr. John Bennett Rubens of 329, High Holborn, W.C. Accountant be, and he is hereby appointed Liquidator for the purposes of such winding-up." At a Meeting of the creditors of the Company held pursuant to sections 238, 239 and 240 of the Companies Act 1929 on the same day at the Holborn Restaurant, Kingsway W.C. at 12 o'clock noon it was resolved that the Resolution be con-firmed and adopted.

ESTHER NYMAN, Chairman. (062)

#### The CYGNET STEAM FISHING COMPANY Limited.

AT an Extraordinary General Meeting of the Members of the said Company, duly convened, and held at the offices of the Fleetwood Fishing Vessel Owners Association Limited, of 192 Dock Fleetwood, at 2.30 p.m. on the first day of July 1938 the following Special Resolution was duly passed :----Special Resolution

"That the Company be wound up voluntarily, and that Mr. Thomas Cardwell, of 200 Dock Street, Fleetwood, be and he is hereby appointed Liquidator for the purposes of such winding up."

JOHN WILLIAM ROBINSON, Chairman. (191)

#### N: C. B. (LIVERPOOL) Limited.

Extraordinary Resolution (pursuant to Section 117 (I) of the Companies Act, 1929) passed 30th June. 1938.

. .

AT an Extraordinary General Meeting of the above named Company duly convened and held at the offices of Messrs. T. W. Stanfield & Co., 11-13, Victoria Street, Liverpool 2, on the thirtieth day of June 1938, the subjoined Extraordinary Resolution was duly passed, viz .:-

#### Resolution.

"That the Company cannot by reason of its liabilities continue its business and that the same be wound up voluntarily and that Mr. Thomas Alphaeus Stanfield, Accountant, 11-13, Victoria Street, Liverpool 2, be and is hereby appointed Liquidator for the purpose of such winding-up.'

H. BLACKLEDGE, Secretary. (192)

## E. F. SANDS Limited.

...

#### The Companies Act, 1929.

AT an Extraordinary General Meeting of the above named Company, duly convened, and held at the office of Messrs. Robt. A. Page & Co., Char-tered Accountants, Milton Chambers, Milton Street, Nottingham, on the 30th day of June, 1938, the following Extraordinary Resolution was duly passed :

passed:— "That it has been proved to the satisfaction of this Meeting that the Company cannot, by reason of its liabilities, continue its business, and that it is advisable to wind up the same, and accordingly that the Company be wound up voluntarily; and that Mr. R. A. Page, of Milton Chambers, Milton Street, Nottingham, be and he is hereby nominated Viguidator for the purpose of such winding-up." Liquidator for the purpose of such winding-up."

(077)

H. REID, Chairman.

#### EADY & DULLEY Limited.

AT an Extraordinary General Meeting of the Members of the above named Company, duly convened, and held at the registered office of the Company, 116

and held at the registered office of the Company, 116 Bridge Street, Northampton, in the county of North-ampton, on the 30th day of June, 1938, the following Resolutions were duly passed, the Resolution num-bered (1) being passed as a Special Resolution, and the Resolution numbered (2) being passed as an Extraordinary Resolution:— (1) "That the Company be wound up volun-tarily, and that Mr. Rupert Sladden, of Liverpool House, 15-17 Eldon Street, London, E.C.2, Char-tered Accountant, and Mr. Henry Abraham Jones, of 47 Park Avenue South, Northampton, in the county of Northampton, Secretary, be and they are hereby appointed Liquidators for the purpose of such winding-up." (2) "That the Liquidators be and are hereby empowered to divide among the contributories in specie any part of the assets of the Company." (o78) W. I. WATKINS, Chairman.

(078) W. J. WATKINS, Chairman.

#### BENSONS (MANCHESTER) Limited.

Special Resolution passed 30th June, 1938.

Special Resolution passed 30th June, 1938. AT an Extraordinary General Meeting of the above named Company duly convened and held at 18 Chepstow Street Manchester on Thursday the 30th day of June 1938 the following Special Reso-lution was duly passed:— "That the Company be wound up voluntarily and that Hr. Henry Whittaker Garnett of 61 Brown Street Manchester Chartered Accountant be and is hereby appointed Liquidator for the purposes of such winding-up."

(200)

O. SIMPSON, Chairman.

#### SYDNEY HUDSON & N. P. NATHANS Limited. Special Resolution passed 30th June, 1938.

AT an Extraordinary General Meeting of the above named Company duly convened and held at 18 Chepstow Street Manchester on Thursday the 30th day of June 1938 the following Special Reso-lution was duly passed:--"That the Company be wound up voluntarily and that Hr. Henry Whittaker Garnett of 61 Brown Street Manchester Chartered Accountant be and is hereby appointed Liquidator for the purposes of such winding-up."

(199)

#### O. SIMPSON, Chairman.

#### H. THOMPSON & CO Limited.

Special Resolution passed 30th June, 1938.

AT an Extraordinary General Meeting of the above named Company duly convened and held at 18 Chepstow Street Manchester on Thursday the 30th day of June 1938 the following Special Reso-lution was duly passed:— "That the Company be wound up voluntarily and that Hr. Henry Whittaker Garnett of 61 Brown Street Manchester Chartered Accountant be and is hereby appointed Liquidator for the purposes of such winding-up."

O. SIMPSON, Chairman. (201)

#### KNOWLES EAST & COMPANY Limited. The Companies Act, 1929.

AT an Extraordinary General Meeting of the Members of the above named Company duly con-vened and held at the Fountain Hotel, Garrett Lane, Tooting, S.W.7, on Monday, the 27th day of June, 1938, the following Extraordinary Resolution was duly passed as an Extraordinary Resolution. "That it had been proved to the satisfaction of this Meeting that the Company cannot, by reason of its liabilities continue its business and that it is

this Meeting that the Company cannot, by reason of its liabilities, continue its business and that it is advisable to wind up the same and accordingly that the Company be wound up voluntarily and that Mr. Charles Edwin James Miell of Tower House, Carfax, in the city of Oxford, be and is hereby appointed Liquidator for the purpose of such winding-up." At the Statutory Meeting of creditors of the above named Company duly convened and held at the

same place and on the same day, the appointment of the Liquidator was confirmed and a Committee of Inspection was appointed consisting of Mr. Arthur Henry St. John Dolton, Mr. Robert Webb, and Mr. Lawrence Ingram Baker.—Dated this 30th day of June, 1938.

EAST Т. KNOWLES, Chairman of both (079) Meetings.

#### The WARREN ROW WHITING CO. Ltd.

AT an Extraordinary General Meeting of the above named Company, duly convened and held at 46 Herga Court, Harrow on the 28th day of June, 1938, the subjoined Extraordinary Resolution was passed, viz.:— "That the Company cannot by reason of its liabilities continue its business, and that it is advis-

liabilities continue its business, and that it is advis-able that the Company be wound up and accordingly that it be wound up voluntarily." At the subsequent Meeting of creditors held on the 29th June 1938 at Warren Row, Berks, Mr. J. Hackett, F.C.A. A.S.A.A. of 48 Uxbridge Road, Ealing, W.5, was appointed Liquidator.

(202) A. BURNETT. Chairman of both Meetings.

#### The FRENCH WINES IMPORTING CO. Limited.

Extraordinary Resolution (pursuant to ss. 117 (1) & 225 (1) (c) of the Companies Act, 1929), passed 29th June, 1938.

AT an Extraordinary General Meeting of the above named Company, duly convened, and held at Winchester House, Old Broad Street, London E.C.2, on the 29th day of June, 1938, the subjoined Extraordinary Resolution was duly passed, viz.:— Resolution.

"That it has been proved to the satisfaction of this Meeting that the Company cannot by reason of its liabilities continue its business, and that it is advisable to wind up the same, and accordingly that the Company be wound up voluntarily, and that Walter Brian Hall, Chartered Accountant, of 82 King William Street, London, E.C.4, be and he is hereby appointed Liquidator for the purposes of such winding-up."

RICHARD M. HARROLD, Chairman of the (210) Meeting.

#### • BRYON FLETCHER AND COMPANY Limited. The Companies Act, 1929.

IN pursuance of section 238 of the Companies Act, 1929, a Meeting of the creditors of the above named Company will be held at the offices of Poppleton and Appleby, I, George Street, Sheffield I, on the 12th day of July 1938 at 3.15 o'clock in the afternoon, for the purposes provided for in the said section.—Dated this 1st day of July 1938. F. B. MELVILLE, Director. (165)

#### WEARN'S AUTOFLEX DRIVES Ltd.

NOTICE is hereby given, pursuant to section 238 of the Companies Act, 1929, that a Meeting of the creditors of the above named Company will be held at 2, Sloane Terrace, London S.W.I, on Thurs-day, the 28th day of July 1938 at 12.30 o'clock in the afternoon, for the purpose if thought fit of nominating a Liquidator and of appointing a Com-mittee of Inspection.—Dated this thirtieth day of June 1038. June 1938.

ROUNDWAY, Director.

#### WEEKS & COOPER Limited.

(014)

(128)

NOTICE is hereby given pursuant to section 238 of the Companies Act 1929 that a Meeting of the creditors of the above named Company will be held at the Institute of Chartered Accountants, Moorgate Place, London, E.C.2, on Thursday the 14th July 1938 at three o'clock in the afternoon for the pur-poses mentioned in sections 238, 239 and 240 of the Companies Act 1929.—Dated the 4th day of July 1938 July 1938.

4356

EDGAR WALKER, Director.

#### PETER DENSHAM Limited. (In Voluntary Liquidation.) The Companies Act, 1929.

Ine Companies Act, 1929. NOTICE is hereby given, pursuant to section 238 of the Companies Act, 1929, that a Meeting of the creditors of the above named Company will be held at No. 17 Clarges Street London W, on Thurs-day, the 21st day of July, 1938, at 11.30 o'clock in the forenoon for the purpose, if thought fit, of nominating a Liquidator and of appointing a Com-mittee of Inspection. Notice is hereby also given that, for the purpose of voting, secured creditors are required (unless they surrender their security) to lodge at the registered office of the Company before the Meeting a statement giving particulars before the Meeting a statement giving particulars of their security, the date when it was given, and the value at which it is assessed.—Dated this 4th day of July, 1938.

J. B. DENSHAM, Director.

#### Re TUBELIGHT SALES Limited. (In Members' Voluntary Liquidation.) The Companies Act, 1929.

(164)

NOTICE is hereby given that the creditors of the above named Company are required, on or before the rst day of August 1938, to send in their names and addresses, with particulars of their debts or and addresses, with particulars of their debts or claims, and the names and addresses of their Solici-tors (if any), to the undersigned, Mr. Wilfred E. Ridsdale, Chartered Accountant, of Messrs. Moore, Stephens & Co., 4, London Wall Avenue, E.C.2, the Liquidator of the said Company; and, if so re-quired by notice in writing by the said Liquidator, are, by their Solicitors or personally, to come in and prove their said debts or claims at such time and place as shall be specified in such notice, or in default thereof they will be excluded from the benefit of any distribution made before such debts are proved.—Dated this 2nd day of July 1938. (053) WILFRED E, RIDSDALE, Liquidator. (053) WILFRED E. RIDSDALE, Liquidator.

## Re The GLEDHILL WEAVING MACHINE CO. Limited. (Members' Voluntary Winding-up.)

NOTICE is hereby given that I intend shortly to make a first and final distribution to shareholders in the above Matter. All persons (if any) having claims against the Company which have not been admitted by me are required on or before Saturday, the 30th July, 1938, to send their names and addresses and particulars of their debts or claims to me, the undersigned John Gordon Turner, Station addresses and particulars of their debts of claims to me, the undersigned John Gordon Turner, Station Street Buildings, Huddersfield, the Liquidator of the Company, and if so required by notice in writing, are by their Solicitors or personally to come in and prove their debts or claims at such time and place as shall be specified in such notice, or in de-fault thereof they will be excluded from the benefit of any distribution made before such debts are of any distribution made before such debts are proved.—Dated this 30th day of June, 1938.

J. G. TURNER, Liquidator. (133)

#### EADY & DULLEY Limited. (In Voluntary Liquidation.) The Companies Act, 1929.

NOTICE is hereby given, that the creditors of the above named Company are required, on or before the 20th day of August, 1938 to send in their names and addresses, with particulars of their debts or claims, and the names and addresses of their Solici-tors (if any), to the undersigned Henry Abraham Jones c/o Northampton Brewery Co. Ltd. of 116 Bridge Street Northampton one of the Liquidators of Bridge Street Northampton one of the Liquidators of the said Company, and, if so required by notice in writing by the said Liquidators, are, by their Solici-tors or personally, to come in and prove their said debts or claims at such time and place as shall be specified in such notice, or in default thereof they will be excluded from the benefit of any distribution made before such debts are proved.—Dated this 1st day of July, 1938.

R. SLADDEN, H. A. JONES,

Liquidators.

(080)

No. 34528

In the Matter of the FRENCH WINES IMPORT-ING CO. Limited, and in the Matter of the Companies Act, 1929.

NOTICE is hereby given that the creditors of the above named Company, which is being voluntarily wound up, are required, on or before the 31st day of July, 1938, to send in their full Christian and surnames, their addresses and descriptions, full particulars of their debts or claims, and the names and addresses of their Solicitors (if any), to the undersigned Walter Brian Hall, Chartered Accountant, of 82 King William Street, London E.C.4, the Liquidator of the said Company, and, if so required by notice in writing from the said Liquidator, are, personally or by their Solicitors, to come in and prove their debts or claims at such time and place as shall be specified in such notice, or in default thereof they will be excluded from the benefit of any distribution made before such debts are proved.—Dated this 1st day of July, 1938. (120) W. BRIAN HALL, Liquidator. NOTICE is hereby given that the creditors of the W. BRIAN HALL, Liquidator. (120)

#### WREFORD BUSH Limited.

#### (In Voluntary Liquidation.)

NOTICE is hereby given that the creditors of the above named Company are required on or before the 25th July 1938 to send in their names and addresses, with particulars of their debts or claims, and of any security held by them, and the names and addresses of their Solicitors (if any) to the under-signed Lawrence W. Robson, Chartered Accountant, of Melbourne House, Aldwych, London, W.C.2, the Liquidator of the said Company, and, if so required by notice in writing by the said Liquidator, are, by their Solicitors or personally. to come in and prove by their Solicitors or personally, to come in and prove by their solutions or personally, to come in and prove their said debts or claims, and to establish any title they may have to priority, at such time and place as shall be specified in such notice. or in de-fault thereof they will be excluded from the benefit of any distribution made before such debts are proved, or such priority is established, or, as the case may be, from objecting to such distribution.-Dated the 30th day of June 1938.

LAWRENCE W. ROBSON, Liquidator. (050)

#### BLYTH PORCELAIN COMPANY Limited. (Members' Voluntary Winding-up.)

#### The Companies Act, 1929.

NOTICE is hereby given that the creditors of the above named Company are required on or before the 8th day of August, 1938, to send in their names and addresses, with particulars of their debts or claims, and the names and addresses of their Solicitors (if any), to the undersigned Richard Ecroyd Clark of 17, Albion Street, Hanley, in the city of Stoke-on-Trent, Chartered Accountant, the Liquidator of the Trent, Chartered Accountant, the Laquidator of the said Company, and if so required by notice in writing from the said Liquidator are, by their Solicitors or-personally, to come in and prove their said debts or claims at such time and place as shall be specified in such notice, or in default thereof, they will be ex-cluded from the benefit of any distribution made before such debts are proved.—Dated this rst day of July 1028 of July, 1938.

(015) R. E. CLARK, Liquidator.

#### BRITISH ELECTRIC DOMESTIC APPLIANCES Ltd. (In Voluntary Liquidation.)

The Companies Act, 1929.

NOTICE is hereby given that the creditors of the NOTICE is hereby given that the creditors of the above named Company are required, on or before Tuesday, the and day of August, 1938, to send their names and addresses, and the particulars of their debts or claims, and the names and addresses of their Solicitors (if any) to Charles Latham of 185-188, High Holborn, London, W.C.r the Liquida-tor of the above named Company; and, if so required by notice in writing from him, the said Liquidator, are, by their Solicitor or personally, to come in and prove the said debts or claims at such time and place as shall be specified in such notice, or in default thereof they will be excluded from the benefit of any distribution made before such the benefit of any distribution made before such debts are proved.—Dated the 1st day of July, 1938. (167)

C. LATHAM, Liquidator,

С

(168)

#### E. W. HARRIS & COMPANY Limited.

E. W. HARKIS & COMPANY Limited. NOTICE is hereby given that the creditors of the above named Company are required, on or before the 26th day of July, 1938, to send in their names and addresses, with particulars of their debts or claims, and the names and addresses of their Solicitors (if any), to the undersigned, Mr. Frederick Robert Hopkins, Incorporated Ac-countant, of 9 Kings Bench Walk, Temple, E.C.4, the Liquidator of the said Company, and, if so required by notice in writing by the said Liquidator, are, by their Solicitors, or personally, to come in required by notice in writing by the said Liquidator, are, by their Solicitors, or personally, to come in and prove their said debts or claims at such time and place as shall be specified in such notice, or in default thereof they will be excluded from the bene-fit of any distribution made before such debts are proved.—Dated this 1st day of July 1938. F. R. HOPKINS, Liquidator. (047)

MAISON FERRIS Ltd. (in Voluntary Liquidation), and in the Matter of the Companies Act, 1929.

NOTICE is hereby given that the creditors of the above named Company, which is being voluntarily wound up, are required on or before the 13th day of August 1938 to send in their full Christian and surnames, their addresses, and descriptions, full par-ticulars of their debts or claims and the names and addresses of their Solicitors (if any) to the under-signed H. Rose of 32, London Wall, E.C.2, the Liquidator of the said Company.—Dated this 28th day of June 1938. (070)

H. ROSE.

#### GEORGE LUNT (STOCKPORT) Limited. (In Voluntary Liquidation.)

NOTICE is hereby given that the creditors of the above named Company are required on or before the 5th day of August 1938 to send in their before the 5th day of August 1938 to send in their names and addresses and particulars of their debts or claims and the names and addresses of their Solicitors (if any) to the undersigned, Leon Douglas Birkett, of 18 Union Road, Underbank, Stockport in the county of Chester, Chartered Accountant, the Liquidator of the said Company; and if so required by notice in writing by the said Liquidator are, by their Solicitors or personally, to come in and prove their said debts or claims at such time and place as shall be specified in such notice or in default thereof they will be excluded from the benefit of any distribution made before such debts are proved. -Dated this 1st day of July 1938.

(166)

L. D. BIRKETT, Liquidator.

## In the Matter of the BRIGHTON AND PRESTON CONSTITUTIONAL CLUB Limited.

CONSTITUTIONAL CLUB Limited. NOTICE is hereby given, in pursuance of section 236 of the Companies Act, 1929, that a General Meeting of the Members of the above named Com-pany will be held at 3 Richmond Terrace, Brighton on Friday the 5th day of August 1938, at 11 o'clock in the forenoon precisely, for the purpose of having an account laid before them, and to receive the Liquidator's report, showing how the winding-up of the Company has been conducted and the property of the Company disposed of, and of hearing any explanation that may be given by the Liquidator; and also of determining by Extraordinary Resolu-tion, the manner in which the books, accounts, papers, and documents of the Company, and of the Liquidator thereof, shall be disposed of.—Dated this first day of July 1938. (116) BEN BROOK, Liquidator.

(116)

### BEN BROOK, Liquidator.

#### JOHNSON & WRIGHT Limited.

#### The Companies Act, 1929.

NOTICE is hereby given that a General Meeting of the Members and creditors of the above named Company will be held at the offices of Poppleton & Appleby King Edward House, New Street, Birming-ham 2, on the 10th day of August 1938, at 12 o'clock noon, for the purpose of having the Liquidator's accounts showing the manner in which the winding-up has been conducted and the property of the Com-pany disposed of laid before such Meeting, and of

hearing any explanation that may be given by the Liquidator; and also of passing a Resolution as to the manner in which the books, accounts, and docu-ments of the Company and of the Liquidator thereof shall be disposed of.—Dated this rst day of July 1938.

#### FREDK. E. BENDALL, Liquidator.

#### EFFINGHAM ESTATES Limited.

#### (In Voluntary Liquidation.)

(In Voluntary Liquidation.) NOTICE is hereby given pursuant to section 245 of the Companies Act 1929 that a Meeting of the Members and creditors of the above named Company will be held at 41, Bedford Row, London W.C.T. on Friday the 12th day of August 1938 at 3 o'clock in the afternoon, for the purpose of having an account and report laid before them by the Liquidator, show-ing how the winding-up of the Company has been conducted and its property disposed of, to hear any explanation that may be furnished by the Liquidator, and to determine the manner of disposal of the books, documents and accounts of the Company.—Dated this 1st day of July, 1938. (160) ARTHUR F. NEWTON, Liquidator.

ARTHUR F. NEWTON, Liquidator. (169)

### The AUDLEY MILL INVESTMENT COMPANY Limited. (Members' Voluntary Winding-up.) The Companies Act, 1929.

Ine Companies Act, 1929. NOTICE is hereby given that, pursuant to sec-tion 236 of the Companies Act, 1929, a General Meeting of the Members of the above named Com-pany will be held at Alexandra House, 49, Preston New Road, Blackburn on Monday the 8th day of August, 1938 at 12 o'clock noon, for the purpose of having an account laid before them showing the manner in which the winding-up has been con-ducted and the property of the Company disposed of and of hearing any explanation that may be given by the Liquidators.—Dated this 30th day of June, 1938. June, 1938.

# J. P. DUXBURY, GEORGE WHITESIDE, Joint Liquidators.

(048)

#### J. E. McMULLEN & SONS Limited. (In Creditors' Voluntary Liquidation.)

(In Creditors' Voluntary Liquidation.) NOTICE is hereby given (pursuant to section 245 of the Companies Act 1929) that a General Meeting of the creditors of the above named Company will be held at the offices of Messrs. Womersley & Tweedale, 47 Mosley Street, Manchester 2, on Fri-day, 19th August 1938 at 12.15 p.m. precisely, to receive the account of the Liquidators, showing how the winding-up of the Company has been conducted and its property disposed of, and to hear any ex-planation that may be furnished; and also of deter-mining the manner in which the books, accounts and documents of the Company and of the Liquidators shall be disposed of; and notice is hereby given (pursuant to section 245 of the Companies Act 1929) that a General Meeting of the Members of the above Company will be held at the offices of Messrs. Womersley & Tweedale, 47 Mosley Street, Manches-ter, on the 19th August 1938 at 12 o'clock noon pre-cisely for the purpose of having an account laid before them and to receive also the Liquidators' report, showing how the winding-up of the Company has been conducted and the property of the Com-pany disposed of, and of hearing any explanations that may be given by the Liquidators.—Dated this 30th day of June 1938. GEOFFREY TWEEDALE, UPDETET SUMPD

(082)

### GEOFFREY TWEEDALE, HERBERT SHARP, Joint Liquidators.

#### FISHERIES BUILDINGS Limited. (In Voluntary Liquidation.)

DEREK G. BEE, Liquidator.

(030)

#### The MONEYSTONE SAND COMPANY Limited. (Members' Voluntary Winding-up.)

The Companies Act, 1929.

NOTICE is hereby given that a General Meeting of the Members of the above named Company will be held at 24, High Street, Cheadle, Staffordshire on Friday, the ninth day of September, 1938 at half-past ten o'clock in the forenoon precisely, to receive the account of the Liquidator showing how the wind-ing-up of the Company has been conducted and its property disposed of; to hear any explanation that may be furnished by the Liquidator; and to pass an Extraordinary Resolution as to the disposal of the books. accounts, and documents of the Company. books, accounts, and documents of the Company. Dated this second day of July, 1938.

S. W. MALKIN, Liquidator. (081)

#### WILLIAM ELMES Limited.

#### (In Voluntary Liquidation.)

The Companies Act, 1929.

The Companies Act, 1929. NOTICE is hereby given that a General Meeting of the Members of the above named Company will be held at 18, East Park Terrace Southampton on Friday, the fifth day of August, 1938 at three-forty-five o'clock in the afternoon precisely, to receive the account of the Liquidator showing how the winding-up of the Company has been conducted and its pro-perty disposed of; to hear any explanation that may be furnished by the Liquidator; and to pass an Ex-traordinary Resolution as to the disposal of the books, accounts, and documents of the Company.— Dated this 30th day of June, 1938. (o8a) W A ROBERTS Liquidator

W. A. ROBERTS, Liquidator. (083)

#### LONDON SERVICE SUITES Limited.

#### (In Voluntary Liquidation.)

The Companies Act, 1929

NOTICE is hereby given that a General Meeting of the Members of the above named Company will be held at 18, East Park Terrace, Southampton, on Friday, the fifth day of August, 1938 at eleven-thirty o'clock in the morning precisely, to receive the account of the Liquidator showing how the windthe account of the Liquidator showing how the wind-ing up of the Company has been conducted and its property disposed of; to hear any explanation that may be furnished by the Liquidator; and to pass an Extraordinary Resolution as to the disposal of the books, accounts, and documents of the Company.— Dated this 30th day of June, 1938.

W. A. ROBERTS, Liquidator. (084)

#### BARIUM CONSOLIDATED Ltd.

(In Voluntary Liquidation.)

#### The Companies Act, 1929.

NOTICE is hereby given pursuant to section 245 of the Companies Act 1929, that General Meetings of the creditors and of the Members of the above named Company will be held at Bassishaw House, Basinghall Street, London, E.C.2 on the 15th day of August 1938, for the purpose of receiving the account of the Liquidator showing how the wind-ing up of the Company has been conducted and its ing-up of the Company has been conducted and its Ing-up of the Company has been conducted and its property disposed of, and to hear any explanation that may be given by the Liquidator. The General Meeting of the creditors of the Company will be held at 12 o'clock noon and the Meeting of the Members of the said Company at 12.15 p.m.—Dated this 1st day of July 1938.

(095)

#### G. C. MUMFORD, Liquidator.

#### P.E.G. TEA CO. Ltd.

#### (Members' Voluntary Winding-up.)

(Members' Voluntary Winding-up.) NOTICE is hereby given pursuant to section 236 of the Companies Act 1929 that a General Meeting of the Members of the above Company will be held at 50-51, Lime Street, London, E.C.3, on the 5th day of August, 1938, at 12 o'clock noon, for the purpose of having an account laid before them show-ing the manner in which the winding-up has been conducted and the property of the Company dis-posed of and of hearing any explanation that may be given by the Liquidator.—Dated this 4th day of July, 1938.

R. C. STILEMAN, Liquidator. (211)

## The LOMAX CONSTRUCTION COMPANY Limited. (In Voluntary Liquidation.)

Limited. (In Voluntary Liquidation.) NOTICE is hereby given pursuant to section 245 of the Companies Act, 1929, that a General Meeting of the Members of the above named Company will be held at the offices of Messrs. Wm. Latham & Co., 12, Park Street, Lytham, Lytham St. Annes, on Wednesday, the 31st day of August, 1938, at 11 a.m., to be followed at 11.30 a.m. by a General Meeting of the creditors of the said Company, for the purpose of having an account laid before them by the Liquidators, showing how the winding-up of the Company has been conducted and the property of the Company disposed of, and of hearing any explanations that may be given by the Liquidators. —Dated this 29th day of June, 1938. WILLIAM LATHAM. Joint Liquidator.

WILLIAM LATHAM, Joint Liquidator. (020)

#### SAMUEL BODDINGTON & SONS Limited. (Members Voluntary Winding-up.)

NOTICE is hereby given that a General Meeting of the Members of the above named Company will be held at the offices of Kimberley, Morrison & Co., be held at the offices of Kimberley, Morrison & Co., Chartered Accountants, 55, Temple Row, Birming-ham, on Tuesday, the 9th day of August, 1938, at 3 p.m., to receive the account of the Liquidator, showing how the winding-up of the Company has been conducted and its property disposed of, and to determine the manner in which the books, accounts, and documents of the Company shall be disposed of.—Dated this 30th day of June, 1938.

P. A. MORRISON, Liquidator. (016)

In the Matter of W. E. GREENWOOD & SONS Limited. (In Voluntary Liquidation.)

NOTICE is hereby given, in pursuance of section 245 of the Companies Act, 1929, that General Meet-ings of the Members and creditors of the above named Company will be held at Audrey House, Ely Place, London, E.C.r, on Thursday the 4th day of August, 1938 at 12 o'clock noon and 12.30 p.m. respectively, for the purpose of having an account laid before them, and to receive the Liquida-tor's report. showing how the winding-up of the tor's report, showing how the winding-up of the Company has been conducted and the property of the Company disposed of, and of hearing any ex-planation that may be given by the Liquidator.— Dated this 4th day of July, 1938. (046)

R. S. FORD, Liquidator.

#### JOHN SUTCLIFFE & SONS (BUILDERS) Limited. (Creditors' Voluntary Winding-up.)

#### The Companies Act, 1929.

NOTICE is hereby given that, pursuant to sec-NOTICE is hereby given that, pursuant to sec-tion 245 of the Companies Act, 1929, General Meet-ings of the Members and the creditors of the above named Company will be held at Alexandra House, 49, Preston New Road, Blackburn, on Tuesday the 9th day of August, 1938 at 3 o'clock and 3.30 o'clock in the afternoon respectively for the purpose of having an account laid before them showing the manner in which the winding-up has been conducted and the property of the Company disposed of and of hearing any explanation that may be given by the Liquidator.—Dated this 29th day of June, 1938. J. P. DUXBURY, Liquidator. (102)

In the Matter of HEATHER PRICE Limited (in Voluntary Liquidation), and in the Matter of the Companies Act, 1929.

Companies Act, 1929. NOTICE is hereby given that General Meetings of the Members and creditors of the above named Company will be held at the offices of Messrs. Beecroft, Sons & Nicholson, 12, Wood Street, Cheap-side in the city of London on Monday the 8th day of August 1938 at 2.30 and 3 o'clock respectively in the afternoon for the purposes of having an account laid before them showing the manner in which the winding-up has been conducted and the property of the Company disposed of and of hearing any explanation that may be given by the Liquida-tor and to pass a Resolution as to the disposal of the books accounts and documents of the Company and Liquidator.—Dated this 30th day of June 1928. WILLIAM NICHOLSON, Liquidator. (004)

#### INTEROCEAN ASPHALTE COMPANY Limited. (In Voluntary Liquidation.)

NOTICE is hereby given (pursuant to sections 236 and 283 (1) (b) of the Companies Act, 1929) that a General Meeting of the Members of the above named Company will be held at Finsbury Pavement House, London E.C.2 on the fifteenth day of August 1938 at 3.30 o'clock in the afternoon, for the purpose 1938 at 3.30 o'clock in the afternoon, for the purpose of having an account laid before them showing the manner in which the winding-up has been conducted and the property of the Company disposed of, and of hearing any explanation that may be given by the Liquidator, and also of determining by Extra-ordinary Resolution the manner in which the books, accounts and documents of the Company and of the Liquidator thereof shall be disposed of.—Dated this 4th day of July, 1938.

(058) .G. A. COLE, Liquidator.

#### OIL IMPORT AND TRADING COMPANY Limited. (In Voluntary Liquidation.)

NOTICE is hereby given (pursuant to sections 236 and 283 (1) (b) of the Companies Act, 1929) that a General Meeting of the Members of the above named Company will be held at Finsbury Pavement House, London E.C.2 on the fifteenth day of August 1938 at 2.30 o'clock in the afternoon, for the purpose of having an account laid before them showing the manner in which the winding-up has been conducted manner in which the winding-up has been conducted and the property of the Company disposed of, and of hearing any explanation that may be given by the Liquidator, and also of determining by Extra-ordinary Resolution the manner in which the books, accounts, and documents of the Company and of the Liquidator thereof shall be disposed of.—Dated this 4th day of July, 1938.

(059)

G. A. COLE, Liquidator.

(090)

#### HENRY GUARD Limited. (In Voluntary Liquidation.) The Companies Act, 1929.

NOTICE is hereby given that a General Meeting of the Members of the above named Company will be held at 18, East Park Terrace, Southampton, on Friday, the fifth day of August, 1938 at three o'clock in the afternoon precisely, to receive the account of the Liquidator showing how the winding-up of the Company has been conducted and its property disposed of; to hear any explanation that may be furnished by the Liquidator; and to pass an Extra-ordinary Resolution as to the disposal of the books, accounts, and documents of the Company.—Dated this 30th day of June, 1938. (087) W. A. ROBERTS, Liquidator. (087)

W. A. ROBERTS, Liquidator.

#### Re EVANS & HOWELL Limited.

NOTICE is hereby given (in pursuance of section 245 of the Companies Act, 1929) that a General Meet-ing of the Members of the above named Company will be held at 133, Aldersgate Street, London, E.C.1 on Thursday, the fourth day of August, 1938 at 12 o'clock noon, for the purpose of having an account laid before them showing the manner in which the winding-up has been conducted and the property of the Company disposed of, and of hearproperty of the Company disposed of, and of hear-ing any explanation that may be given by the Liquidator.—Dated this 4th day of July, 1938. W. A. CHAPMAN, Liquidator. (069)

#### LEWIS BROS. (FULHAM) Limited. (In Voluntary Liquidation.)

(In Voluntary Liquidation.) NOTICE is hereby given pursuant to section 245 of the Companies Act, 1929, that (I) a General Meeting of the above Company, and (2) a Meeting of the creditors of the said Company, will be held at 6, York Street, Twickenham, Middlesex, on Wednesday the 10th day of August, 1938, at 12 noon and 12.30 p.m. respectively, for the purpose of having laid before them the Liquidator's account showing how the winding-up has been conducted and the property of the Company disposed of, and of hearing any explanation that may be given by the Liquidator; and to pass a Resolution as to the disposal of the books, accounts and documents of the Company.--Dated this 1st day of July, 1938. (180) L. LEWIS, Liquidator. L. LEWIS, Liquidator. (189)

#### BLOOM (SALISBURY) Limited. (In Voluntary Liquidation.)

The Companies Act, 1929.

NOTICE is hereby given that a General Meeting of the Members of the above named Company will be held at 18, East Park Terrace, Southampton on Friday, the fifth day of August, 1938, at four o'clock in the afternoon precisely, to receive the account of the Liquidator showing how the windaccount of the Liquidator showing how the wind-ing-up of the Company has been conducted and its property disposed of; to hear any explanation that may be furnished by the Liquidator; and to pass an Extraordinary Resolution as to the disposal of the books, accounts, and documents of the Com-pany.—Dated this 30th day of June, 1938. W. A. ROBERTS, Liquidator. (089)

#### INTOURIST Limited.

#### (Members' Voluntary Winding-up.) The Companies Act, 1929

NOTICE is hereby given that a General Meeting of the Members of the above named Company will be held at Bush House, Aldwych W.C.2 on Satur-day, the sixth day of August, 1938 at eleven o'clock in the forenoon precisely, to receive the account of the Liquidators showing how the wind-ing-up of the Company has been conducted and its property disposed of; to hear any explanation that may be furnished by the Liquidators; and to pass an Extraordinary Resolution as to the disposal of the books. accounts. and documents of the 'Comthe books, accounts, and documents of the Company.-Dated this 30th day of June, 1938:

H. LENCE, G. KIPPEN,

Liquidators.

#### Re EVANS & HOWELL Limited.

NOTICE is hereby given (pursuant to sections 245 (and 283 (I) (b) ) of the Companies Act, 1929) that a Meeting of the creditors of the above named Coma Meeting of the creditors of the above named Com-pany will be held at 133. Aldersgate Street, London, E.C.I on Thursday, the fourth day of August, 1938 at 12.30 o'clock in the afternoon, for the purpose of having an account laid before them showing the manner in which the winding-up has been con-ducted and the property of the Company disposed of, and of hearing any explanation that may be given by the Liquidator.—Dated this 4th day of luby 1028 July, 1938. (068)

W. A. CHAPMAN, Liquidator.

#### C. B. BISCUIT COMPANY Limited.

(In Voluntary Liquidation.)

The Companies Act, 1929.

The Companies Act, 1929. NOTICE is hereby given that a General Meeting of the Members of the above named Company will be held at 18, East Park Terrace, Southampton, on Friday, the fifth day of August, 1938 at eleven o'clock in the morning precisely, to receive the account of the Liquidator showing how the winding-up of the Company has been conducted and its property disposed of; to hear any explanation that may be furnished by the Liquidator; and to pass an Extraordinary Resolution as to the disposal of the books, accounts, and documents of the Com-panv.—Dated this 30th day of June, 1938. -Dated this 30th day of June, 1938. pany.-(088)

W. A. ROBERTS, Liquidator.

#### HENRY GUARD (PETERSFIELD) Limited. (In Voluntary Liquidation.)

#### The Companies Act, 1929.

The Companies Act, 1929. NOTICE is hereby given that a General Meeting of the Members of the above named Company will be held at 18, East Park Terrace, Southampton, on Friday, the fifth day of August, 1938, at three-fifteen o'clock in the afternoon precisely, to receive the account of the Liquidator showing how the winding-up of the Company has been conducted and its property disposed of; to hear any explana-tion that may be furnished by the Liquidator; and to pass an Extraordinary Resolution as to the dis-posal of the books, accounts, and documents of the Company.—Dated this 30th day of June, 1938. (086) W A ROBERTS Liquidator.

(086)

W. A. ROBERTS, Liquidator.

(181)

(000)

(100)

#### HENRY GUARD (RINGWOOD) Limited.

(In Voluntary Liquidation.)

The Companies Act, 1929.

NOTICE is hereby given that a General Meeting of the Members of the above named Company will be held at 18, East Park Terrace, Southampton on Friday, the fifth day of August, 1938, at three-thirty o'clock in the afternoon precisely, to receive the account of the Liquidator showing how the the account of the Liquidator showing how the winding-up of the Company has been conducted and its property disposed of; to hear any explana-tion that may be furnished by the Liquidator; and to pass an Extraordinary Resolution as to the dis-posal of the books, accounts, and documents of the Company.—Dated this 30th day of June, 1938. (085) W. A. ROBERTS, Liquidator.

#### RUSPLATINA Limited.

#### (Members' Voluntary Winding-up.) The Companies Act, 1929.

NOTICE is hereby given that a General Meeting of the Members of the above named Company will be held at 51-53 Hatton Garden E.C.r on Saturday, the sixth day of August, 1938 at 11.30 o'clock in the forenoon precisely, to receive the account of the Liquidators showing how the winding-up of the Company has been conducted and its property dis-posed of; to hear any explanation that may be furnished by the Liquidators; and to pass an Extra-ordinary Resolution as to the disposal of the books, accounts, and documents of the Company.—Dated this 30th day of June, 1938. this 30th day of June, 1938.

(091)

(094)

H. LENCE, M. KLEBANOFF, Liquidators.

#### WALLSON MINERALS Limited. (In Voluntary Liquidation.)

The Companies Act, 1929.

NOTICE is hereby given pursuant to section 245 of the Companies Act 1929, that General Meetings of the creditors and of the Members of the above of the creditors and of the Members of the above named Company will be held at Bassishaw House, Basinghall Street, London, E.C.2 on the 15th day of August 1938, for the purpose of receiving the account of the Liquidator showing how the wind-ing-up of the Company has been conducted and its property disposed of, and to hear any explanation that may be given by the Liquidator. The General Meeting of the creditors of the Company will be held at 12.30 p.m. and the Meeting of the Members of the said Company at 12.45 p.m.—Dated this 1st day of July 1938. 1st day of July 1938.

G. C. MUMFORD, Liquidator.

NOTICE is hereby given that the Partnership which was for some time past carried on by Herbert Joseph Johnson and Ernest John Fuller under the style of FULLER AND JOHNSON at Hall Street Soham in the county of Cambridge, in the trade or business of Builders and Contractors has been dissolved as from the 9th day of Feb-ruary 1938 by reason of the death of the said Herbert Joseph Johnson. All debts due to and owing by the said late firm will be received and owing by the said late firm state of Scham paid respectively by Cecil Ernest Fuller of Soham aforesaid who will carry on the said business under the style or firm of Fuller and Johnson.—As witness our hands this 27th day of June 1938.

AGNES J. JOHNSON. LAURA MAUD STOCKLEY

Executrices of the said Herbert ]oseph Johnson. ERNEST J. FULLER.

(018)

NOTICE is hereby given that the Partnership heretofore subsisting between us, the undersigued David Leslie and John Mallows Young carrying on business as Auctioneers Estate Agents and Sur-veyors at 99 Church Road Upper Norwood S.E.19 and 239 Knights Hill West Norwood S.E.27 under the style or firm of MANSELL AND ROWE has

been dissolved by mutual consent as from the thirtieth day of June 1938. All debts due and owing to or by the late firm will be respectively received and paid by the said David Leslie and John Mallows Young. The said business will be carried on in the future by the said David Leslie at 99 Church Road Upper Norwood S.E.19 and John Mallows Young at 239 Knights Hill West Norwood S.E.27.—As witness our hands this atth day of June 1028 witness our hands this 30th day of June 1938.

D. LESLIE. JOHN M. YOUNG.

NOTICE is hereby given that the Partnership heretofore subsisting between us the undersigned Leonard Flavell Daniel Flavell and Joseph Flavell carrying on business as Manufacturers of Gas Water and Steam Tubes and Fittings at Upper Ettingshall Road Coseley near Bilston in the county of Stafford under the name or style of the TUBE FITTINGS MANUFACTURING COMPANY has been dis-solved by mutual consent as from the thirtieth day of April one thousand nine hundred and thirty-five so far as concerns the said Joseph Flavell. hve so far as concerns the said Joseph Flavell. All debts due to and owing by the said late firm will be received and paid by the said Leonard Flavell and Daniel Flavell who will continue to carry on the said business under the same style or firm.—As witness our hands this 29th day of June one thousand nine hundred and thirty-eight.

L. FLAVELL.
D. FLAVELL.
J. FLAVELL.

NOTICE is hereby given that the Partnership heretofore subsisting between us, the undersigned Rolda Gladys Parkinson and Dorothy Alice Thomas carrying on business as Estate Agents at By the Way, Connaught Avenue, Frinton on Sea under the style or firm of D. THOMAS & CO. has been dis-solved by mutual consent as from the 26th day of May 1028 All debts due and owing to or but the May 1938. All debts due and owing to or by the late firm will be respectively received and paid by the said Dorothy Alice Thomas. The said business will be carried on in the future by the said Dorothy Alice Thomas.—As witness our hands this 27th day of June 1938.

	ROLDA G	. P/	ARKINSON.
(182)	DOROTHY	Ϋ́Α.	THOMAS.

NOTICE is hereby given that the Partnership heretofore subsisting between Leonard James Cole the elder Leonard James Cole the younger and Richard Henry Cole carrying on business as Wet and Fried Fish Proprietors at 241 Trysull Road Merry Hill Wolverhampton in the county of Stafford under the style or firm of COLE BROS. & Son has been dissolved as from the 13th day of June 1938. All debts due and owing to or by the late firm will be respectively received and paid by the said Leonard James Cole the elder and Leonard James Cole the younger who will continue to carry on the said business.—Dated this 28th day of June 1938.

L. J. COLE (the Elder). L. J. COLE (the Younger). R. H. COLE.

NOTICE is hereby given that the Partnership hitherto subsisting between Thomas Martin Hardy and John Gardiner Hardy carrying on business as Woollen Merchants at 16 and 18 Beak Street Golden Square in the county of London under the style or firm of "HOWARD HARDY & COMPANY" was dissolved as from the 1st day of November 1937 by mutual consent. The said business will be carried on in the future by the said Thomas Martin Hardy mutual consent. The said business will be carried on in the future by the said Thomas Martin Hardy who has formed for this purpose the Company known as "Howard Hardy & Co. Limited" whose registered office is situate at 16 and 18 Beak Street Golden Square London aforesaid. The New York house of the said business will be carried on in the future by the said John Gardiner Hardy alone under the name of "Howard Hardy & Co. Inc."—Dated this twenty-fifth day of June 1938.

JOHN G. HARDY.

STONEHAM and SON, Solicitors for the above named Thomas Martin Hardy and for Howard (017) Hardy and Company Limited.

NOTICE is hereby given, that the Partnership heretofore subsisting between us, the undersigned Lewis Brandreth and Alan Brandreth carrying on business as Corn Merchants at The Mill, Helsby, Cheshire under the style or firm of H. BRANDRETH has been dissolved by mutual consent as from the 30th day of June 1938. All debts due and owing to or by the late firm will be respectively received and paid by the said Lewis Brandreth by whom the said business will be carried on in the future under the said style of H. Brandreth at the said address. —As witness our hands this 30th day of June 1938. LEWIS BRANDRETH

(097)

LEWIS BRANDRETH. ALAN BRANDRETH.

..**.** . •

NOTICE is hereby given that the Partnership heretofore existing between us the undersigned carrying on business as Colliery Proprietors, Coal Exporters, Coal Factors and Merchants and Manu-facturers of "Cymric" Ovoid Fuel at Copper-works Yard Llanelly in the county of Carmarthen under the style or firm of PHILIP ROGERS & CO. has been dissolved by mutual consent as from the thirtieth day of June one thousand nine hundred and thirty-eight. All debts due to and owing by the said late firm will be received and paid by me the undersigned Frank Martin Rogers who will carry on the said business under the style or firm of carry on the said business under the style or firm of "Cymric Ovoid Fuel & Coal Co."—Dated this 30th day of June 1938.

#### PHILIP ROGERS. . FRANK MARTIN ROGERS. (101)

NOTICE is hereby given that the Partnership heretofore subsisting between us the undersigned Fred Hurst and Richard George Wilson carrying on business as Manufacturers of Bedspreads at 4 St. Michaels Road West Croydon Surrey under the style or firm of F. HURST & CO. has been dissolved by mutual consent as from the first day of June 1938. All debts due to and owing by the said late firm will be received and paid by the said Fred Hurst.—Dated this 2nd day of July 1938.

		FRED HURST.
(041) · ·	•• •	R. G. WILSON.

.....

NOTICE is hereby given, that the Partnership heretofore subsisting between us, the undersigned Inez Amelot Bishop and Alison Lawton carrying on business as Cafe, Teashop and Quick Lunch Restaurant Proprietors at Roxburghe House, Regent Street, W.I, under the style or firm of HIGHWAYS CAFE has been dissolved by mutual consent as from the 3rst day of March 1938. All debts due and owing to or by the late firm will be respectively received and paid by the said Alison Lawton. The said business will be carried on in the future by the said Alison Lowter As without for both the active details. Lawton .- As witness our hands this 29th and 30th days of June 1938.

I. A. BISHOP. A. LAWTON. (002)

NOTICE is hereby given that the Partnership lately subsisting between us, the undersigned George Boorman of 28, Virginia Road Thornton Heath in the county of Surrey Walter Boorman of 116 Anerley Park London S.E.20 and Percy Thomas Boorman of 118 Anerley Park London S.E.20 carrying on busi-uess as Motor Engineers and Acents under the stella or firm of RED HOUSE GARAGE, has been dis-solved by mutual consent as from the 30th day of June 1938.—As witness our hands this 28th day of June 1938.

(174)	•		G. BOORMAN. WALTER BOORMAN. P. T. BOORMAN.
· •			
	•	•	

NOTICE is hereby given that the Partnership herctofore subsisting between us the undersigned Enoch Smith and Thomas Herbert Barratt carrying on business as General Dealers at "Monument Stores" Old Birmingham Road Rednal in the parish of Bromsgrove in the county of Worcester under the style or name of BARRATT & SMITH has been dissolved by mutual consent as from the

5th day of January 1938 so far as concerns the said Enoch Smith who retires from the said busi-ness. All debts due to and owing by the said late firm will be received and paid respectively by the said Thomas Herbert Barratt who will continue on the said business under the existing style or name. —As witness our hands this 24th day of June 1938.

(098)

THOMAS HERBERT BARRATT. ENOCH SMITH.

NOTICE is hereby given that the Partnership heretofore subsisting between George Passingham and heretofore subsisting between George Passingham and Alfred Edward Passingham carrying on business as Solicitors at Hitchin in the county of Hertford under the style or firm of PASSINGHAM & HILL has been dissolved by mutual consent as and from the 30th day of June 1938. All debts due to and owing by the said late firm will be received and paid by the said Alfred Edward Passingham who will continue to carry on the said business under the style or firm of Passingham & Hill.—Dated this 30th day of June 1938. 30th day of June 1938.

PASSINGHAM and HILL, Solicitors for both (125) Partners.

NOTICE is hereby given that the Partnership heretofore subsisting between us the undersigned Charles Andrew Little and Leonard Seymour Shaw carrying on business as Potato Merchants at 10, Earl's Court Gardens, London, S.W.5 and 15, The Grove, Teddington, Middlesex, under the style or firm of LITTLE, SHAW & CO. has been dissolved by mutual consent, as and from the 12th day of May 1938.—Dated this 30th day of June 1938.

CHARLES ANDREW LITTLE. LEONARD SEYMOUR SHAW. (126)

NOTICE is hereby given that the Partnership heretofore subsisting between Alan Hartnell Mortimer and Claude Douglas Weatherhead carrying Mortimer and Claude Douglas Weatherhead carrying on business as Motor and Electrical Engineers at Burway Bridge Ludlow in the county of Salop under the style or firm of MORTIMER & WEATHERHEAD has been dissolved as from the seventh day of June one thousand nine hundred and thirty-eight so far as concerns the said Alan Hartnell Mortimer who retires from the said firm.—Dated this sixteenth day of June one thousand nine hundred and thirty-eight.

A. H. MORTIMER. C. D. WEATHERHEAD. (134)

NOTICE is hereby given, that the Partnership heretofore subsisting between us, the undersigned Cyril Stanley Thomas Charles Wilfrid Mallord Turner and Francis Arnold Bradshaw Everington carrying on business as Solicitors at Haddon House, 66A Fenchurch Street, London, E.C.3 under the style or firm of HATCHETT JONES & CO. has been dissolved by mutual consent as from the 30th day of June 1938. All debts due and owing to or by the late firm will be respectively received and paid of June 1938. All debts due and owing to or by the late firm will be respectively received and paid by the said Charles Wilfrid Mallord Turner. The said business will be carried on in the future by the said Charles Wilfrid Mallord Turner.—As witness our hands this 30th day of June 1938. C. STANLEY THOMAS. C. W. M. TURNER. (o60) F. A. B. EVERINGTON.

NOTICE is hereby given that the Partnership heretofore subsisting between us the undersigned Enoch Alfred Hughes and Sidney Rupert Hughes carried on under the style or firm of E. HUGHES & SONS at 52 King Street, Dudley, Worcestershire in the trade or business of Cabinet Makers and Furniture Manufacturers was on the 24th day of June 1938 dissolved by mutual consent. All debts due to and owing by the said late firm will be respectively received and paid by the said Sidney Rupert Hughes who will continue to carry on the said business alone under the style or firm of "E. Hughes & Sons."—Dated this 24th day of June 1938. E. A. HUGHES. E. A. HUGHES. S. R. HUGHES.

.

(031)

#### Re JOHN JAMES HAWORTH, Deceased. Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all creditors and others having any claims against the estate of John James Haworth late of 55 Preston Old Road, Witton, Blackburn in the county of Lancaster Witton, Blackburn in the county of Lancaster Retired Mill Manager who died on the 29th day of January 1937 and whose Will was proved in the Principal Probate Registry on the 8th day of March 1937 by Lucas Haworth, Herbert Haworth and John Haworth the executors therein named are hereby required to send particulars thereof in writ-ing to the undersigned Solicitors on or before the 12th day of September 1938 after which date the said executors will proceed to distribute the assets said executors will proceed to distribute the assets said executors will proceed to distribute the assets of the deceased among the persons entitled thereto having regard only to the claims of which they shall then have had notice and will not be liable for the assets of the deceased or any part thereof so distributed to any person or persons of whose claims or demands they thal not then have had notice.—Dated this 2nd day of July 1938.

CAMPBELL and CO., County Bank Chambers, New Market Street, Blackburn, Solicitors for (043) the said Executors.

#### Re WILLIAM SHAW, Deceased.

Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all creditors and other persons having any debts, claims or demands against the estate of William Shaw late of Salcombe against the estate of William Snaw late of Salcombe House Seaton in the county of Devon formerly of Moorcroft Sticklepath Okehampton in the said county deceased (who died on the 2nd day of December 1929 and letters of administration of whose estate (with Will annexed) were granted by the Principal Registry of the Probate Division of His Majesty's High Court of Justice on the 4th day of June 1936 to Violet Maud Lucy Finch) are hereby required to send in the particulars of their debts of June 1936 to Violet Maud Lucy Finch) are hereby required to send in the particulars of their debts, claims or demands to us the undersigned, the Solici-tors for the said administratrix on or before the roth day of September 1938, after which date the said administratrix will proceed to distribute the assets of the said deceased amongst the persons entitled thereto, having regard only to the claims and demands of which she shall then have had notice; and she will not be liable for the assets of the said deceased or any part thereof so dis-tributed to any person or persons of whose debts, claims or demands she shall not then have had notice.—Dated this second day of July 1938. notice .- Dated this second day of July 1938.

CHILCOTT and PEARSE, Tavistock, Devon, (042) Solicitors for the said Administratrix.

#### Estate of the late Mrs. DOROTHY NINA TYRWHITT.

NOTICE is hereby given pursuant to the Trustee Act 1925 that all persons having any claim against the estate of Dorothy Nina Tyrwhitt late of z Russell Road, Kensington, London, W.14, who died on the 27th day of April, 1938 and probate of whose Will was granted by the Principal Probate Registry on the 10th day of June 1938 to Thomas Tyrwhitt are hereby required to send particulars in writing of such claims to the undersigned before the eighth day of September 1938 after which date the executor will proceed to distribute the assets among the persons entitled thereto having regard only to the claims of which they then shall have had notice.—Dated 30th June 1938. CRECORY ROWCULEEE and CO I Bedford NOTICE is hereby given pursuant to the Trustee

GREGORY, ROWCLIFFE and CO., 1, Bedford Row, London, W.C.1, Solicitors for the (107) Executor.

#### Sir THOMAS EVANS FLITCROFT, Deceased.

Pursuant to. the Trustee Act, 1925 (Section 27). Pursuant to the Irustee Act, 1925 (Section 27). NOTICE is hereby given that all creditors and other persons having any claims and demands against the estate of the late Sir Thomas Evans Flitcroft of 28, Somerset Road, Bolton, Lancashire who died on the 14th day of May 1938 and whose Will was proved in the Principal Registry of His Majesty's High Court of Justice on the 16th day of June 1938 by William Carlton Rothwell of Brant-wood, Lostock, Bolton and Arthur William Flitcroft Lee of Lime Grove, Prestwich, Manchester the executors therein named are hereby required to

send particulars thereof in writing to the under-signed the Solicitors for the said executors on or signed the Solicitors for the said executors on or before the 13th day of September 1938 after, which date the said executors will proceed to distribute the assets of the deceased among, the persons entitled thereto having regard only to the claims and demands then notified and will not be liable for the assets of the deceased or any part thereof so distributed to any persons of whose claims and demands they shall not then have had notice.— Dated this 1st day of July 1938.

HINDLE SON and COOPER, 4, Church Street, (037) Darwen, Solicitors for the said Executors. 

#### · · · HARRY SHEPHERD, Deceased.

.

.

Pursuant to the Trustee Act, 1925. ALL persons having any claims against the estate of Harry Shepherd, deceased, late of 103, High Street, Gosforth, Newcastle upon Tyne, who died Street, Gosforth, Newcastle upon Tyne, who died on the 28th September, 1937 and letters of adminis-tration of whose estate were granted by the New-castle upon Tyne District Probate Registry on the 13th December, 1937 to William Thomas Smith and Annie Robson, are required to send particulars thereof to the undersigned on or before the 6th day of September, 1938, after which date the adminis-trators will proceed to distribute the estate of the said deceased amongst the persons entitled thereto having regard only to the claims then notified. Dated the 30th day of June, 1938.

DEES and THOMPSON, 40, Grey Street, New-castle upon Tyne, Solicitors of the Adminis-(157) trators.

1993 A. . ., 5.1 . . 9 1.1211 1

1 1 1 1 1 1 1 1 1 1

56. -

# Re ALICE MARY CUNLIFFE, Deceased.

Pursuant to the Trustee Act, 1925. NOTICE is hereby given that all creditors and, others having any claims against the estate of Alice Mary Cunliffe late of Pencarwick formerly Alice Mary Cunlifie late of Pencarwick formerly of 6, Morton Road both in Exmouth in the county of Devon Spinster who died on the 1st May 1938 and whose Will was proved in the Principal Pro-bate Registry on the 27th June 1938 by Richard Wardle Lynn the executor therein named, are hereby required to send particulars thereof in writ-ing to the undersigned Solicitors on or before the ing to the undersigned Solicitors on or before the 17th September next, after which date the said executor will proceed to distribute the assets of the deceased among the persons entitled thereto; having regard only to the claims of which he shall then have had notice, and will not be liable for the assets of the deceased or any part thereof so distributed, to any person or persons of whose claims or demands he shall not then have had notice.—Dated this 2nd July 1938. ENCLI JOINSON and JUNN 28 For Street

FINCH JOHNSON and LYNN, 18, Fox Street, Preston, Lancashire, Solicitors for the said (106) Executor.

#### Re MARGARET JACKSON, Deceased. Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all creditors and NOTICE is hereby given that all creditors and other persons having any debts, claims, or demands against the estate of Margaret Jackson late of Brae Holme, Ambleside in the county of Westmorland deceased (who died on the 28th day of of March 1938, and whose Will was proved in the Carlisle District Registry of the Probate Division of His. Majesty's High Court of Justice on the 14th day of June 1938, by the Public Trustee of the Public Trustee Office, Kingsway London W.C.2, and Thomas Creighton of Waterhead, Ambleside, aforesaid the executors therein named) are hereby required to send in the particulars of their debts, claims or demands to the undersigned, the Solicitors for the 'said executo the undersigned, the Solicitors for the said execu-tors, on or before the seventh day of September tors, on or before the seventh day of September 1938, after which date the said executors will 'pro-ceed to distribute the assets of the said deceased amongst the persons entitled thereto, having regard... only to the claims and demands of which they shall... then have had notice; and they will not be liable for the assets of the said deceased or 'any' part ... thereof so distributed, to any person or persons of whose debts, claims or demands they shall not then have had notice...Dated this rst day of July, 1938.

ARNOLD, GREENWOOD and SON, Exchange Chambers, Kendal, Wesmorland, Solicitors for (149) the said Executors.

FREDERICK WILLIAM BROOKES, Deceased.

Pursuant to the Trustee Act, 1925 (as amended). NOTICE is hereby given that all creditors and other persons having any claims against the estate of Frederick William Brookes late of 34 Hamlet Gardens Ravenscourt Park London formerly of "Clifton," Hove Street, Hove, Sussex Medical Practitioner, deceased, who died on the 1st day of May 1938 and whose Will was proved in the Principal Probate Registry on the 23rd day of June 1938 by Arthur Auger Keen and William Richmond Churchill the executors therein named are hereby required to send the particulars in writing, of their claims or demands to us the undersigned as Solicitors for the said executors on or before the 12th day of September 1938 after which date the said executors will proceed to distribute the assets of the said deceased amongst the persons entitled thereto, having regard only to the debts, claims, and demands of which they shall then have had notice.—Dated this 30th day of June 1938.

CROPLEY DAVIES and SON, 96, Westbourne (155) Grove, W.2, Solicitors for the said Executors.

#### Mrs. EMILY AUGUSTA DRAIN, Deceased. Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all persons having any claim against the estate of Emily Augusta Drain formerly of 157 Northdown Road Margate but late of "La Cabane" 47, By Pass Road, St. Paul's Cray, Kent, Widow who died on the 13th day of February 1938 and whose Will was proved in the Principal Probate Registry by Arthur Edward Drain (the surviving executor named in the said Will) are hereby required to send particulars thereof in writing to the undersigned the Solicitors for the executor on or before the eninth day of September 1938 after which date the estate will be distributed having regard only to the claims then notified.— Dated this 30th day of June 1938

WILSON and TOYNE, 3, Cecil Square, (104) Margate, Solicitors for the Executor.

# Re MARGARET (otherwise MAGGIE) DAVIES, Deceased.

Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all creditors and others having any claims against the estate of Margaret (otherwise Maggie) Davies late of Cernyw Castle Street Criccieth in the county of Caernarvon Spinster formerly of Meirion House Criccieth aforesaid who died on the 5th day of February 1938 and letters of administration with the Will annexed of whose estate were granted out of the Bangor District Probate Registry on the 23rd day cf June 1938 are hereby required to send particulars thereof in writing to the undersigned Solicitor on or before the 12th day of September 1938 after which date the administrators will proceed to distribute the assets of the deceased among the persons entitled thereto having regard cnly to the claims of which they shall then have had notice; and will not be liable for the assets cf the deceased or any part thereof so distributed to any person or persons of whose claims or demands they shall not then have had notice.—Dated this 30th day of June 1938.

E. LLOYD JONES, 239, High Street, Bangor, Caernarvonshire, Solicitor for the Administrators.

(066)

# LUCY PRICE VAUGHAN FREEMAN, Deceased. Pursuant to the Trustee Act, 1925 (as amended).

NOTICE is hereby given that all creditors and other persons having any claims or demands against the estate of Lucy Price Vaughan Freeman late of 19, Chiltern Gardens, Hornchurch, Essex, formerly of 9, North Street, Barking, Essex, Widow, deceased, who died on the 31st day of May, 1938 and whose Will was proved in the Principal Registry of the Probate Division of His Majesty's High Court of Justice on the 27th day of June, 1938 by Joseph Rickard Crellin and William Daniel Brown the executors therein named are hereby required to send the particulars in writing, of their claims or demands to us the undersigned as Solicitors for the said executors on or before the roth day of September 1938 after which date the said executors will proceed to distribute the assets of the said deccased amongst the persons entitled thereto, having regard only to the debts, claims and demands of which they shall then have had notice; and they will not be liable for the assets of the said deceased, or any part thereof, so distributed, to any person or persons of whose claims or demands they shall not then have had notice.—Dated this 1st day of July 1938.

CHARLES HUMPHRIES and CO., 36, Basinghall Street, London, E.C.2, Solicitors for the (156) said Executors.

#### Re WILLIAM MORLEY, Deceased.

#### Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all creditors and other persons having any debts, claims, or demands against the estate of William Morley, late of Chapel Hill Farm, Shipley in the county of Derby, Farmer, deceased (who died on the 17th day of March 1938, and whose Will was proved in the Principal Registry of the Probate Division of His Majesty's High Court of Justice on the 30th day of June 1938, by Philip Walker and Arthur Toplis Annibal, the executors therein named) are hereby required to send in the particulars of their debts, claims or demands to us the undersigned, the Solicitors for the said executors, on or before the 8th day of September 1938, after which date the said executors will proceed to distribute the assets of the said deceased amongst the persons entitled thereto, having regard only to the claims and demands of which they shall then have had notice; and they will not be liable for the assets of the said deceased or any part thereof so distributed, to any person or persons of whose debts, claims or demands they shall not then have had notice.—Dated this 2nd day of July 1938.

W. HOLLIS BRIGGS and CO., 6, The Strand, (148) Derby, Solicitors for the said Executors.

#### Re THOMAS WINTER LEE, Deceased. Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all persons having any claims against the estate of Thomas Winter Lee late of 18, Princes Road, Wimbledon, Surrey, Draper, who died on the 8th March 1938 and to whose estate letters of administration were granted on the 29th day of June 1938 by the Principal Probate Registry, are hereby required to send particulars thereof in writing to us, the undersigned, on or before the 24th day of September 1938, after which date the administratrices will proceed to distribute the assets of the said deceased amongst the persons entitled thereto, having regard only to the claims or demands then notified.—Dated this 1st day of July 1938.

H. H. WELLS and SONS, 17, Paternoster Row, London, E.C.4, Solicitors for the (007) Administratrices.

#### Re ANDREW EMM MELVILLE, otherwise ANDREW MELVILLE, Deceased.

Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all persons having claims against the estate of Andrew Emm Melville (otherwise Andrew Melville) of 36 Foundry Street Brighton and "Whychcote" South Street Portslade both in the county of Sussex Theatrical Producer who died on the 4th day of March 1938 intestate and letters of administration to whose estate were granted out of the Lewes District Probate Registry on the 25th day of June 1938 to Dorothy Hughes Melville and Andrew Emm Melville are hereby required to send particulars in writing of their claims and demands to the undersigned the Solicitors for the said administrators on or before the said administrators will proceed to distribute the assets of the said deceased amongst the persons entitled thereto having regard only to the claims and demands of which they shall then have had notice and they will not be liable for the assets of the said deceased or any part thereof so distributed to any person or persons of whose claims or demands they shall not then have had notice.— Dated this 30th day of June 1938.

NYE and DONNE, 58, Ship Street, Brighton  $\tau$ , (009) Solicitors for the Administrators.

4364

#### Re CHARLES SAMUEL, Deceased. Pursuant to the Trustee Act, 1925.

:

Fursuant to the inustee Act, 1925. NOTICE is hereby given that all creditors and others having debts claims or demands against the estate of Charles Samuel, late of 66, North Gate, Regents Park N.W. in the county of London (who died on the 8th day of April 1938) and whose Will was proved in the Principal Probate Registry on the 12th day of May 1938 by Stanley Solomon Levene and Stanley Braham Cohen the executors named therein are hereby required to send in par-ticulars in writing of their debts claims or demands to me the undersigned the Solicitor to the executors on or before the 29th day of August 1938 after which date the said executors will proceed to diswhich date the said executors will proceed to dis-tribute the assets of the said deceased amongst the persons entitled thereto having regard only to the claims and demands of which notice has been given and they will not be liable for the assets of the said deceased or any part thereof, so distributed, to any person or persons of whose claims or demands they shall not then have had notice.— Dated 29th day of June 1938.

STANLEY S. LEVENE, 17, Surrey Street, Strand, London, W.C., Solicitor for the said (013) Executors.

#### Mrs. EDITH ELLEN COUPLAND, Deceased, Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all persons having any claim against the estate of Mrs. Edith Ellen Coupland late of "Blenheim," Sylvan Avenue, Woodhall Spa who died on the 15th day of April 1938 are hereby required to send particulars thereof in writing to the Executor and Trustee Department, Lloyds Bank Limited, 39, Threadneedle Street, London E.C.2, the executor of the Will of the said Mrs. Edith Ellen Coupland or to the undersigned, the Solicitor to the executor on or before the 10th day of September 1938, after which date the executor will proceed to distribute the said estate, having regard only to the claims then notified.—Dated this regard only to the claims then notified.—Dated this 4th day of July 1938.

R. CHATTERTON, Horncastle, Lincs, Solicitor (150) for the said Executor.

#### Re GEORGE HENRY HOWARD, Deceased. Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all creditors and others having any claims against the estate of George Henry Howard late of No. 52 East Street, East Stonehouse in the city of Plymouth, Super-annuated Postman who died on the 27th day of May 1938 and whose Will was proved in the Exeter District Probate Registry on the 29th day of June 1938 by Reginald Holmes Cumming and Richard Robinson Rodd the executors therein named are 1938 by Reginald Holmes Cumming and Richard Robinson Rodd the executors therein named are hereby required to send particulars thereof in writ-ing to the undersigned Solicitor on or before the soth day of September 1938 after which date the said executors will proceed to distribute the assets of the deceased among the persons entitled thereto having regard only to the claims of which they shall then have had notice; and will not be liable for the assets of the deceased or any part thereof so distributed to any person or persons of whose claims or demands they shall not then have had notice.— Dated this 4th day of July 1938. R. ROBINSON RODD, East Stonehouse, (105) Plymouth, Solicitor for the said Executors.

The Right Honourable JOHN WYNFORD VISCOUNT ST. DAVIDS, Deceased. THE

Pursuant to the Trustee Act, 1925, as amended.

NOTICE is hereby given that all creditors and other persons having any debts claims or demands against the estate of the Right Honourable John Wynford the Viscount St. Davids, P.C. late of Landwade Hall Exning near Newmarket Suffolk and also of Cudlow House Rustington Sussex and and also of Cuddow House Rushington Sussex and also of 235. Almoners House St. James' Court Westminster London deceased (who died on the 28th March 1938 and whose Will and two Codicils were proved in the Principal Registry of the Pro-bate Division of His Majesty's High Court of Justice on the 16th May 1938 by John Soame Austen and Harold Charles Drayton the executors therein named) are hereby required to send the

No. 34528

particulars, in writing, of their debts claims or demands to us, the undersigned, the Solicitors for demands to us, the undersigned, the Solicitors for the said executors, on or before the r5th September 1938 after which date the said executors will pro-ceed to distribute the assets of the said deceased amongst the persons entitled thereto, having regard only to the debts, claims and demands of which the executors shall then have had notice; and the executors will not be liable for the assets of the said deceased, or any part thereof so distributed to any person or persons of whose debts, claims or demands the executors shall not then have had notice.—Dated this 30th day of June 1938. BARNETT TUSON and CO 25 New Broad

BARNETT TUSON and CO., 35, New Broad Street, London, E.C.2, Solicitors for the (005) .Executors. . . . .

## Re HARRY SCOTT FORTUNA, Deceased: Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all creditors and other persons having any debts, claims, or demands against the estate of Harry Scott Fortuna late of Mill House, Stambourne, Great Yeldham, in the county of Essex deceased (who died on the 8th day of November 1937, and whose Will was proved in the Principal Registry of the Probate Division of His Majesty's High Court of Justice on the 1st day of January 1938, by Alma Jesse Wiffin of Collins Farm Stambourne Great Yeldham aforesaid the execution therein named) are hereby required to send executor therein named) are hereby required to send executor therein named) are hereby required to send in the particulars of their debts, claims or demands to us the undersigned, the Solicitors for the said executor, on or before the 5th day of September 1938, after which date the said executor will pro-ceed to distribute the assets of the said deceased amongst the persons entitled thereto, having regard only to the claims and demands of which he shall there have had notice: and he will not be liable for then have had notice; and he will not be liable for the assets of the said deceased or any part thereof so distributed, to any person or persons of whose debts, claims or demands he shall not then have had notice.—Dated this and day of July, 1938.

BATES, ELLISON and CO., Haverhill, (146) Suffolk, Solicitors for the said Executor.

# The Most Honourable GEORGE LOUIS VICTOR HENRY SERGIUS MOUNTBATTEN, MAR-QUESS OF MILFORD HAVEN, G.C.V.O., QUESS Deceased.

Pursuant to the Trustee Act, 1925 (as amended).

ALL persons having claims against the estate of the above named Marquess of Milford Haven late of Lynden Manor Holyport Bray in the county of Berks, who died on the 8th day of April 1938, and whose Will was proved on the 15th day of June 1938, are required to send written particulars to the undersigned by the 9th day of September 1938, after which date the executors will distributed the de-ceased's estate, having regard only to valid claims then notified.—Dated this 29th day of June 1938.

CHARLES RUSSELL and CO., 37, Norfolk Street, Strand, London, W.C.2, Solicitors for. (158) the Executors.

## Re FREDERICK TRAFFORD, Deceased. Pursuant to the Trustee Act, 1925.

Pursuant to the Trustee Act, 1925. NOTICE is hereby given that all creditors and others having any claims against the estate of Frederick Trafford late of I Chichester Terrace Brighton in the county of Sussex who died on the 18th day of April 1938 and whose Will was proved in the Lewes District Probate Registry on the 27th day of June 1938 by Charles James Aubrey Wade and Benjamin Bodycombe the executors therein named are hereby required to send particulars thereof in writing to the undersigned Solicitors on or before the 9th day of September 1938 after which date the said executors will proceed to distribute the assets of the deceased among the persons entitled thereto having regard only to claims of which they shall then have had notice; and will not be liable for the assets of the deceased or any part thereof so distributed to any person or persons of whose claims or demands they shall not then have had notice.—Dated this 1st day of July 1938. GRIFFITH SMITH WADE and RILEY, 47,

GRIFFITH SMITH WADE and RILEY, Old Steyne, Brighton I, Sussex, Solicitors for (122) the said Executors

ï

D

## BENJAMIN WILLIAM BALL, Deceased.

Pursuant to the Trustee Act, 1925 (as amended). ALL persons having claims against the estate of Benjamin William Ball late of 5 Rosebank Road Hanwell in the county of Middlesex, who died on the 13th day of March 1938, and letters of adminis-tration of whose estate were granted by the Prin-cipal Registry of the Probate Division of His Majesty's High Court of Justice on the 7th day of May 1938, are required to send written particulars to the undersigned by the 10th day of September 1938, after which date the administrators will dis-tribute the deceased's estate. having regard only to tribute the deceased's estate, having regard only to valid claims then notified.—Dated this 30th day of June 1938.

LEWIS W. TAYLOR, I, Verulam Buildings, Gray's Inn, W.C.I, Solicitor for the Adminis-(159) trators.

#### 'Re SELINA MARY HISCOCK, Deceased. Pursuant to the Trustee Act, 1925.

Pursuant to the Trustee Act, 1925. NOTICE is hereby given that all creditors and other persons having any debts, claims or demands against the estate of Selina Mary Hiscock late of 215 London Road Villas Balderton Newark in the county of Nottingham formerly of Tiverton in the county of Devon Spinster deceased (who died on the fifth day of May 1938, and whose Will with a Codicil thereto was proved in the Principal Registry of the Probate Division of His Majesty's High Court of Justice on the 24th day of June 1938, by Alfred Southwell of Wisbech Saint Peter in the county of Cambridge Solicitor the surviving execu-tor therein named) are hereby required to send in the particulars of their debts, claims or demands to us the undersigned, the Solicitors for the said executor, on or before the 7th day of September 1938, after which date the said executor will pro-ceed to distribute the assets of the said deceased amongst the persons entitled thereto, having regard only to the claims and demands of which he shall then have had notice; and he will not be liable for the assets of the said deceased or any part thereof only to the claims and demands of which he shan then have had notice; and he will not be liable for the assets of the said deceased or any part thereof, so distributed to any person or persons of whose debts, claims or demands he shall not then have had notice.—Dated this 30th day of June 1938.

SOUTHWELL DENNIS and FARROW, 2, Post Office Lane, Wisbech, Solicitors for the (143) said Executor.

#### ALICE WARREN, Deceased.

#### Pursuant to the Trustee Act, 1925.

ALL persons having claims against the estate of Alice Warren late of 29 Brighton Road Balsall of Alice Warren late of 29 Brighton Road Balsall Heath in the city of Birmingham Spinster (who died on the 26th day of May 1938 and whose Will was proved in the Birmingham District Probate Registry on the 21st day of June 1938 by Midland Bank Executor and Trustee Company Limited the executor named in the said Will) are required to send particulars thereof in writing to us the under-signed on or before the 6th day of September 1938 after which date the said eventor will proceed to after which date the said executor will proceed to distribute the estate having regard only to the claims then notified.—Dated this 30th day of June 1938.

S. J. GREY and WILLCOX, 22, Church Street, Birmingham 3, Solicitors for the (144) Executor.

# CARLOS ALFONSO GONZALEZ DE ORBEGOZO CONDE DE OLMOS, Deceased. Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all creditors and NOTICE is hereby given that all creditors and other persons having any debts claims or demands against the estate of Carlos Alfonso Gonzalez de Orbegozo Conde de Olmos late of Hossegor Landes in France and formerly of San Sebastian in Spain deceased (who died on the 21st day of March, 1937 and letters of administration to whose estate with Will annexed were granted out of the Principal Probate Division of His Majesty's High Court of Justice on the twenty-second day of June 1938 to the Hon. Geoffrey Cokayne Gibbs and Leonard Charles Michael Gibbs both of 22 Bishopsgate in the city of London Merchant Bankers the lawfully appointed Attorneys of Vicente Gonzalez de Pinillos appointed Attorneys of Vicente Gonzalez de Pinillos

Orbegozo Moncada the executor in the said Will named) are hereby required to send particulars in writing of their debts claims or demands to us the writing of their debts claims or demands to us the undersigned Solicitors for the said Attorney-Administrators on or before the sixth day of September next after which date the said Attorney-Administrators will proceed to distribute the assets of the said deceased amongst the persons entitled thereto having regard only to the debts claims and demands of which they shall then have had notice and they will not be liable for the assets of the said deceased or any part thereof so distributed to any person or persons of whose debts claims or demands they shall not then have had notice.— Dated this 1st day of July 1938. LOHNSON JECKS and COLCIOUCH 24

JOHNSON, JECKS and COLCLOUGH, 24, Austin Friars, E.C.2, Solicitors for the said (185) Attorney-Administrators.

#### THOMAS WADMORE BRAVINGTON, Deceased. Pursuant to the Trustee Act, 1925

Pursuant to the Trustee Act, 1925. NOTICE is hereby given that all creditors and other persons having any debts claims or demands against the estate of Thomas Wadmore Bravington late of No. 9 Carlton Road Sidcup and formerly of No. 12 Limes Road Folkestone both in the county of Kent who died on the 30th day of October 1937 and whose Will with a Codicil thereto was proved in the Principal Registry of the Probate Division of His Majesty's High Court of Justice on the 27th day of May 1938 by National Provincial Bank Limited and Thomas Hulme the executors therein named are hereby required to send par-ticulars in writing of their debts claims or demands to us the undersigned the Solicitors for the said executors on or before the 12th day of September 1938 after which date the executors will proceed executors on or before the 12th day of September 1938 after which date the executors will proceed to distribute the assets of the deceased among the persons entitled thereto having regard only to the debts claims and demands of which they shall then have had notice; and that they will not be liable for the assets or any part thereof so distributed to any persons of whose debts claims or demands they shall not then have had notice.—Dated this 1st day of July 1938.

H. B. BRADLEY and HULME, 2, Manor Road, (145) Folkestone, Solicitors for the said Executors.

#### ARTHUR ALLCOCK, Deceased.

#### Pursuant to the Trustee Act, 1925.

ALL persons having any claims or demands against the estate of Arthur Allcock, late of "Hill-side," Sandfield Road, Arnold, Nottinghamshire, Retired Builder, who died on the 6th day of February 1938 and probate of whose Will was granted on the 8th day of April 1938, are required to send written particulars to the undersigned by the 5th day of September 1938 after which date the executors will distribute the deceased's estate having regard only to the claims or demands then notified.—Dated this 3oth day of June, 1938.

PERRY, PARR and FORD, Friary Chambers, Friar Lane, Nottingham, Solicitors for the (142) Executors.

#### LOUIS MILLER, Deceased.

#### Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all persons having any claim against the estate of Louis Miller late of any claim against the estate of Louis Miller late of 55 Lytton Avenue and formerly of 118 Stock Street both in Cheetham, Manchester, and who carried on business as a Timber Merchant at 14 Mayes Street Shudehill (but formerly at 144 Corporation Street) both in Manchester, who died on the 1st day of May 1938 are hereby required to send particulars of their elements writing to the Eventor and Trustee May 1938 are hereby required to send particulars of their claims in writing to the Executor and Trustee Department, Lloyds Bank, King Street, Manchester, the executor (jointly with Mr. George Rollitt) of the Will of the said Louis Miller, or to the under-signed, the Solicitors to the executors, on or before the roth day of September 1938, after which date the executors will proceed to distribute the said estate, having regard only to the claims then notified.—Dated this 1st day of July 1938. W P. W MURRAY and CO. To Braseness

W. R. W. MURRAY and CO., 19, Brazennose Street. Manchester 2, Solicitors for the (061) said Executors.

#### ROBERT GARDNER, Deceased.

Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all creditors and other persons having any claims or demands against the estate of Robert Gardner late of Albany House, Balcombe Road, Branksome Park, Bournemouth deceased (who died on the 10th day of January 1938 and whose Will together with two Codicils 1938 and whose Will together with two Codicils was proved in the Principal Probate Registry at London on the 11th day of March 1938 by Edith Charlotte Gardner, Michael Robert Gardner and Nancy Christina Gardner the executors therein named) are hereby required to send the particulars, in writing, of their claims or demands to us the undersigned on or before the 9th day of September 1938, after which date the said executors will pro-ceed to distribute the assets of the said deceased.— Dated this 2nd day of July 1938. Dated this 2nd day of July 1938.

CROSSE and SONS, 38, Bedford Square, Bloomsbury, London, Solicitors for the (055) said Executors.

# Major LEWIS KERRISON JARVIS, Deceased. Pursuant to the Trustee Act, 1925 (Section 27).

NOTICE is hereby given that all creditors and other persons having any claims or demands against the estate of Major Lewis Kerrison Jarvis late of 4 Stafford Mansions, Stafford House, Buckingham Gate, in the county of Middlesex, who died on the 16th day of May 1938 and whose Will was proved in the Principal Registry of the Probate Division of His Majesty's High Court of Justice on the 28th day of June 1938 by Barclays Bank Limited the executors therein named are hereby required to send particulars thereof in writing to the Trustee Department Barclays Bank Limited, 37 King William Street, London, E.C.4 or to the undersigned on or before the roth day of September 1938 after which date the said executors will proceed to distri-NOTICE is hereby given that all creditors and which date the said executors will proceed to distriwhich date the said executors will proceed to distri-bute the assets of the said deceased amongst the persons entitled thereto having regard only to the claims or demands then notified and will not be liable for the assets of the deceased or any part thereof so distributed to any persons of whose claims or demands they shall not then have had notice.--Dated this rst day of July, 1938.

JANSON, COBB, PEARSON and CO., 22, College Hill, E.C.4, Solicitors for the said (056) Executors.

#### Re JOHN GIBBONS, Deceased.

#### Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all creditors and persons having any claims or demands against the estate of John Gibbons of Garryduff, Bexton, Knuts-ford, Plasterer and Decorator, deceased who died on the 15th day of April 1938 and whose Will and Codicil were proved by John Anthony Gibbons and Midland Bank Executor and Trustee Company Limited on the 28th day of June 1938 in the Man-chester District Registry of the Probate Division of the High Court of Justice are hereby required to send written particulars thereof to the undersigned on or before the 6th day of September 1938 after which date the said executors will distribute the assets having regard only to the claims then received. —Dated this 1st day of July 1938. WILDING, EABLEY and PEGGE, 67, Princess estate of John Gibbons of Garryduff, Bexton, Knuts-

WILDING, EARLEY and PEGGE, 67, Princess Street, Manchester 2, Solicitors for the (109) Executors.

## EDITH LINDSAY, Deceased.

#### Pursuant to the Trustee Act, 1925.

Pursuant to the Trustee Act, 1925. NOTICE is hereby given that all creditors and other persons having claims against the estate of Edith Lindsay late of 2, Wharf Road, Ellesmere in the county of Salop, Spinster, deceased, who died on the 24th day of March, 1938, and probate of whose Will was granted by the Principal Probate Registry on the 13th day of June 1938 to Midland Bank Executor and Trustee Company Limited and Winifred Davey are hereby required to send the particulars, in writing, of their claims to the under-signed, the Solicitors for the executors of the Will of the deceased within two months of the date of the publication hereof after which date the said

executors will proceed to distribute the assets of the said deceased amongst the persons entitled thereto, having regard only to the claims of which they shall then have had notice and they will not be liable for the assets of the said deceased, or any part thereof, so distributed, to any person or persons of whose claims they shall not then have had notice. —Dated this 29th day of June, 1938.

GILES and HORTON, Trimpley, Ellesmere, (065) Salop, Solicitors for the Executors.

# Re HENRY TROUGHTON, Deceased.

#### Pursuant to the Trustee Act, 1925.

Pursuant to the fragment and persons having NOTICE is hereby given that all persons having claims against the estate of Henry Troughton, late of 2 Croftlands, Westbourne Road in the city of Patired Pharmacist (who died on the of 2 Croftlands, Westbourne Road in the city of Lancaster, Retired Pharmacist (who died on the 7th day of April 1938 and whose Will was proved by Winifred Mary Troughton and Charles Frederick Gardner in the Lancaster District Probate Registry on the 5th day of May 1938) are hereby required to send particulars thereof in writing to me the undersigned on or before the 6th day of September 1938, after which date the executors will proceed to distribute the estate of the deceased amongst the parties entitled thereto, having regard only to the claims of which they shall then have had notice. --Dated this 30th day of June 1938.

C. FRED. GARDNER, 31, Sun Street, Lan-(124) caster, Solicitor for the said Executors.

# Re ANNE SUTTON, Deceased.

#### Pursuant to the Trustee Act, 1925.

Pursuant to the Trustee Act, 1925. NOTICE is hereby given that all creditors and other persons having any debts, claims, or demands against the estate of Anne Sutton late of 16, Hall Road in the city and county of Kingston upon Hull, Spinster deceased (who died on the 20th day of May 1937, and whose Will was proved in the Principal Registry of the Probate Division of His Majesty's High Court of Justice on the 15th day of July 1937, by Jane Jackson Dyas and Henry James Barney the executors therein named) are hereby required to send in the particulars of their debts, claims or demands to us the undersigned, the Solicitors for the said executors, on or before the fifth day of September 1938, after which date the said executors will proceed to distribute the assets of the said deceased amongst the persons entitled thereto, having regard only to the claims and demands of which they shall then have had notice; and they will not be liable for the assets of the said deceased or any part thereof so dis-tributed, to any person or persons of whose debts, claims or demands they shall not then have had notice.—Dated this 30th day of June 1938. WOODHOUSE and SON, 8, Parliament Street, (xm) Hyll Solicitors for the said executors

WOODHOUSE and SON, 8, Parliament Street, (140) Hull, Solicitors for the said Executors.

# Re the Estate of the late Mrs. HENRIETTA PYBUS DE POTHONIER.

# Pursuant to the Trustee Act, 1925, Section 27 (as amended).

(as amended). NOTICE is hereby given that all creditors and other persons having any claims or demands against the estate of Henrietta Pybus de Pothonier of No. 51 Carlisle Mansions in the city of Westminster Widow who died on the 27th day of January 1938 and whose Will was proved at the Principal Probate Registry on the 13th day of April 1938 by Vera Henrietta Sligo Spedding and Frederick Herbert Cooper Christmas, the executors named in the said Will, are hereby required to send the particulars in writing of their claims and demands to the undersigned the Solicitors for the said executors on or before the 5th day of September 1938 after which date the said executors will proceed to dison or before the 5th day of September 1938 after which date the said executors will proceed to dis-tribute the assets of the said deceased amongst the persons entitled thereto having regard only to the claims and demands of which the said executors shall then have had notice and will not be liable for the assets of the deceased or any part thereof so distributed to any persons of whose claims or demands the said executors shall not then have had notice.—Dated this 2nd day of July 1938. NICHOLSON EREFIAND and SHEPHERD

NICHOLSON FREELAND and SHEPHERD, 46, Queen Anne's Gate, Westminster, S.W.I, (121) Solicitors to the said Executors.

D 2

#### Re GEORGE THOMAS WHITTLE, Deceased. Pursuant to the Trustee Act, 1925

ALL persons having claims against the estate of George Thomas Whittle late of "South Point" Goff's Lane Cheshunt in the county of Hertford who died on the 22nd day of February 1938 and whose Will was proved by Herbert Ernest Fass the Public Trustee, the executor therein named, on the 7th day of June 1938 in the Principal Probate Registry, are required to send particulars thereof in writing to the undersigned on or before the 16th day of September next after which date the executor day of September next after which date the executor will proceed to distribute the assets having regard only to the claims of which he shall then have had notice.—Dated this first day of July 1938.

(187) for the said Executor.

#### MILDRED AUGUSTA LOVEGROVE TUSTIN, Deceased.

#### Pursuant to the Trustee Act, 1925.

ALL persons having claims against the estate of Mildred Augusta Lovegrove Tustin late of Fern Bank Mildred Augusta Lovegrove Tustin late of Fern Bank Old Marston in the county of Oxford who died on the 5th day of May 1938 and whose Will was proved by the Public Trustee the sole executor therein named on the 22nd day of June 1938 in the Oxford District Probate Registry are hereby required to send particulars in writing of their claims to the under-signed on or before the 12th day of September 1938 after which date the executor will proceed to dis-tribute the assets of the deceased having regard only to the claims of which he shall then have had notice. to the claims of which he shall then have had notice. -Dated 1st day of July 1938.

LINNELL and MURPHY, 12, King Edward Street, Oxford, Solicitors for the said Execu-(108) tor.

#### EMMA CAMM, Deceased.

#### Pursuant to the Trustee Act, 1925.

ALL persons having any claims against the estate of Emma Camm of Smedleys Hydropathic Establish-ment Matlock Derbyshire Widow who died on the 18th April 1938 and whose Will was proved at Manchester on 29th June 1938 by the Public Trustee (Manchester) are required to send particulars thereof to the Deputy Public Trustee, Arkwright House, Parsonage Gardens, Manchester 3, before 13th September 1938 after which date the executor will distribute the estate of the deceased having regard only to the claims then received.—Dated 1st July only to the claims then received .- Dated 1st July 1938.

EMERSON and FIELDEN, 78, King Street, (137) Manchester 2, Solicitors for the said Executor.

# Estate of the late ALBERT CHARLES WRATISLAW, C.B., C.M.G., C.B.E.

Pursuant to the Trustee Act, 1925, as amended.

ALL persons having any claim against the estate of the late Albert Charles Wratislaw late of the Thatch Felixstowe Suffolk who died on the 28th day Inatch Feinstowe Suifolk who died on the 28th day of April 1938 and whose Will was proved in the Principal Probate Registry on the 15th day of June 1938 by the executors therein named are hereby required to send particulars thereof to the under-signed on or before the 15th September 1938 after which date the executors will proceed to distribute the said estate having regard only to the claims then notified.—Dated 29th June 1938.

RIDER HEATON MEREDITH and MILLS, 8, New Square, Lincoln's Inn, W.C.2, Solici-(177) tors for the Executors.

#### Re ALFRED ROBINSON, Deceased.

#### Pursuant to the Trustee Act, 1925.

ALL persons having any claims against the estate of Alfred Robinson late of Falder House, Shaw Road, Bushbury, near Wolverhampton in the county of Stafford, and of 2, Long Common, Claver-ley, near Wolverhampton aforesaid Retired Grocer (who died on the 12th day of January 1938 and letters of administration to whose estate were granted out of the Principal Probate Registry on the 28th day of June 1938) are required to send particulars thereof in writing to the undersigned the Solicitors to the administrator on or before the 7th day of September 1938, after which date the said administrator will distribute the estate having regard only to the claims then received.—Dated this first day of July 1938.

HALL and SON, Waterloo 25, Road. (123) Wolverhampton.

2

#### Re ARTHUR CURTIS HASLAM, Deceased.

#### Pursuant to the Trustee Act, 1925.

ALL persons having claims against the estate of Arthur Curtis Haslam late of "Higham" 4 Bishops Arthur Curtis Haslam late of "Higham" 4 Bishops Avenue, Finchley in the county of Middlesex Ad-vertising Agent (who died on the 1st day of May 1938, and whose Will was proved by Westminster Bank Limited the executor therein named on the 24th day of June 1938 in the Principal Probate Registry) are required to send particulars thereof to the undersigned on or before the 10th day of Sep-tember 1938 next after which date the executor will proceed to distribute the estate having regard only proceed to distribute the estate having regard only to the claims of which it shall then have had notice. -Dated this 29th day of June 1938.

S. D. F. CAMPBELL, 87, Hornsey Lane, High-gate, London, N.6, Solicitor for the said (176) Executor.

ALL persons having claims against the late MAX FRIEDRICH WIDMER-HOCH of Cliffords Inn, Temple Bar, London, E.C., and of Basle, Switzer-land, who died intestate at Basle aforesaid on the Temple Bar, London, E.C., and of Basle, Switzer-land, who died intestate at Basle aforesaid on the and day of February 1938 and to whose estate letters of administration were granted to Eric Half Guy and Reginald William Freeman Bank Officials on the 31st day of May 1938, are requested to send the same, in writing, to me the undersigned, on or before the 5th day of September 1938, after which date the estate will be distributed, having regard only to the claims of which notice has then been received by the undersigned.—Dated this 4th day of July, 1938. of July, 1938.

E. HALL GUY, 99, Gresham Street, London, E.C.2, Administrator of the Estate of Max (186) Friedrich Widmer-Hoch, Deceased.

# Re Mrs. MARY GRAHAM, Deceased.

#### Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all creditors and persons having any claims or demands against the estate of Mary Graham, of 19, The Quadrant, Forbes estate of Mary Granam, of 19, The Quadrant, Forces Road, Stockport, Widow, deceased who died on the 2nd day of May 1938 and whose Will was proved by Francis Henry Earley and Joseph Graham on the 14th day of June 1938 in the Manchester Dis-trict Registry of the Probate Division of the High Court of Justice are hereby required to send written particulars thereof to the undersigned on or before the fith day of Scattering of the the of the test of test of test of the test of t the 6th day of September 1938 after which date the said executors will distribute the assets having regard only to the claims then received.—Dated this 1st day of July 1938.

WILDING, EARLEY and PEGGE, 67, Princess Street, Manchester 2, Solicitors for the (110) Executors.

#### WILLIAM HENRY WEST, Deceased.

#### Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all persons having any claim against the estate of William Henry West late of 207 Foundry Lane and 22 Canal Walk both in the county borough of Southampton Grocer who died on the 20th day of June 1938 are hereby required to send particulars thereof in writing to the Executor and Trustee Department, Lloyds Bank Limited, 39 Threadneedle Street, London E.C.2, the executor of the Will of the said William Henry West, or to the undersigned, the Solicitors to the executor, on or before the 9th day of September 1938, after which date the executor will proceed to distribute the said estate having regard only to the claims then notified.--Dated this 1st day of July. 1938.

EWING and HICKMAN, 26, Portland Terrace, (136) Southampton, Solicitors for the said Executor.

#### Re CHARLES JACKSON SUTTON, Deceased. Pursuant to the Trustee Act, 1925.

Pursuant to the Trustee Act, 1925. NOTICE is hereby given that all creditors and other persons having any debts, claims of demands against the estate of Charles Jackson Sutton late of 147, Beverley Road in the city and county of Kingston-upon-Hull, Retired Pawnbroker, deceased (who died on the 23rd day of November 1925, and whose Will was proved in the Principal Registry of the Probate Division of His Majesty's High Court of Justice on the 16th day of February 1926 by Anne Sutton and Henry James Barney the executors therein named) are hereby required to send in the particulars of their debts, claims or demands to us the undersigned, the Solicitors for the said Henry James Barney, the surviving executor, on or before the 5th day of September 1938, after which date the said executor will proceed to distribute the assets of the said deceased amongst the persons entitled there-to, having regard only to the claims and demands of which he shall then have had notice; and he will not be liable for the assets of the said deceased or any be liable for the assets of the said deceased or any of whose debts, claims or demands he shall not then have had notice.-Dated this 30th day of June 1938.

WOODHOUSE and SON, 8, Parliament Street, (139) Hull, Solicitors for the said Executor.

#### Re Miss ALICE THORNLEY, Deceased. Pursuant to the Trustee Act. 1925.

NOTICE is hereby given that all persons having any claims against the estate of Alice Thornley late of 15, Moorland Road St. Annes-on-the-Sea in the of 15, Moorland Road St. Annes-on-the-Sea in the county of Lancaster, deceased (who died on the 7th day of May, 1938, and letters of administration to whose estate were granted out of the Lancaster District Registry of the Probate Division of His Majesty's High Court of Justice on the 27th day of June, 1938, to Mary Ann Thornley the adminis-tratrix) are hereby required to send particulars in writing of their claim to us, the undersigned, the Solicitors for the administratrix on or before the 5th Solicitors for the administratrix on or before the 5th day of September, 1938, after which date the said administratrix will proceed to distribute the assets of the said deceased amongst the persons entitled thereto, having regard only to the claims of which she shall then have had notice.—Dated this 30th day of June, 1938.

ROLAND W. ROBINSON and SON, 27, Birley Street, Blackpool, Solicitors for the Adminis-(141) tratrix.

# HELEN ELIZABETH WHIELDON, Deceased.

Pursuant to the Trustee Act, 1925. Pursuant to the Trustee Act, 1925. NOTICE is hereby given that all persons having any claim against the estate of Helen Elizabeth Whieldon, Widow, late of No. 7 Meadvale Road, Ealing in the county of Middlesex, who died on the 19th day of April, 1938, at the West Middlesex Hospital, 30 Twickenham Road, Isleworth, Middlesex, and whose Will was proved in the Principal Probate Registry on the 14th day of June, 1938, by Sibert James Lott, of 51 Dorset Road, Merton Park in the county of Surrey, the sole executor thereof, are requested and required to send particulars of such claims to the undersigned Solicitor for the said executor, on or before the 10th day of September, 1938, after which the executor will proceed to distribute the estate, having executor will proceed to distribute the estate, having regard only to the claims of which the said Sibert James Lott shall have had due notice.—Dated this 30th day of June, 1938.

M. GOMM, 12, The Mall, Ealing, W.5, Solicitor to the said Executor Sibert James W.5, WM. (184) Lott.

#### EDITH SWORDER, Deceased.

Pursuant to the Trustee Act, 1925 (Section 27).

NOTICE is hereby given that all creditors and other persons having any claims or demands against the estate of Edith Sworder late of "Rosebank," Constable Road, Ipswich in the county of Suffolk, Widow, who died on the eighth day of April 1938 and whose Will was proved in the Ipswich District Registry of the Probate Division of His Majesty's High Court of Justice on the 24th day of June 1938 by Barclays Bank Limited, the executor therein named, are hereby required to send particulars thereof in writing to the Trustee Department Barclays Bank Limited, Bank Plain, Norwich, or to the undersigned on or before the 8th day of Septem-ber 1938 after which date the said executor will pro-ceed to distribute the assets of the said decreased ceed to distribute the assets of the said decreased amongst the persons entitled thereto having regard only to the claims or demands then notified and will not be liable for the assets of the decreased or any part thereof so distributed to any persons of whose claims or demands they shall not then have had notice.—Dated this 1st day of July 1938.

GOTELEE and GOLDSMITH, 11, Arcade Street, (111) Ipswich, Solicitors for the said Executor.

#### WILLIAM JOSEPH PRICE, Deceased.

#### Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all persons having claims against the estate of William Joseph Price, late of "Shalimar" The Avenue, Hatch End, in the acte of Shahmar The Avenue, Hatch End, in the county of Middlesex, formerly of 8, Stapleton Road, Tooting in the county of London, Retired School-master, deceased, who died on the 5th day of April 1938, and whose Will was proved in the Principal Registry of the Probate Division of His Majesty's High Court of Justice on the 25th day of June 1938, by Midland Bank Executor and Trustee Company by Midland Bank Executor and Irustee Company Limited and William Corbridge, the executors named in the said Will, are required to send particulars thereof in writing to Midland Bank Executor and Trustee Company Limited, Poultry, London, E.C.2, or to the undersigned on or before the roth day of September 1938 after which date the estate will be distributed having regard only to the claims then received.—Dated this 1st day of July, 1938.

WETHERFIELD BAINES and BAINES, 6, New Burlington Street, London, W.I, Solici-(175) tors for the said Executors.

#### Re HERBERT THOMAS EDWARDS, Deceased, Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all creditors and other persons having any claims or demands against the estate of Herbert Thomas Edwards late of The of Sussex Retired Schoolmaster deceased (who died on the r6th day of December 1937 intestate and letters of administration to whose estate have been granted out of the Principal Probate Registry to Francis George Edwards and William Mathew Edwards the administrators of the said estate) are hereby requested to send particulars in writing of their claims or demands to us the undersigned the Solicitors for the said administrators on or before the 5th day of September 1938 after which date the said stin day of September 1938 after which date the said administrators will proceed to distribute the assets of the said deceased amongst the persons entitled thereto having regard only to the debts claims and demands of which they shall then have had notice and they will not be liable for the assets of the said deceased or any part thereof so distributed, to any person or persons of whose claims or demands they shall not then have had notice.-Dated this 30th day of June 1938.

PEARLESS, DE ROUGEMONT and CO., East Grinstead, Sus (093) Administrators. Sussex, Solicitors for the said

#### MARY GORDON, Deceased.

# Pursuant to the Trustee Act, 1925.

Pursuant to the Trustee Act, 1925. ALL persons having claims against the estate of Mary Gordon late of 26 Gay Street Bath in the county of Somerset Wildow who died on the 19th April 1938 and whose Will and Codicil was proved in the Principal Probate Registry on the 22nd June 1938 by Douglas Claude Strathern Evill the executor named therein, are required to send particulars there-of in writing to the undersigned on or before the 6th day of September next, after which date the executor will proceed to distribute the assets having regard only to the claims of which the executor shall then have received notice.—Dated this second day of July 1938. 1938.

GODDEN HOLME and WARD, 34, Old Jewry, London, E.C.2, Solicitors to the said Execu-(178) tor.

#### GEORGE WILLIAM ROBERT HOLDITCH, Deceased.

Pursuant to the Trustee Act, 1925, Section 27.

NOTICE is hereby given that all persons having any claims against the estate of George William Robert Holditch late of 11 Linden Road, Horn-castle who died on the 22nd day of June 1938 must send particulars of such claims to me the under-signed on or before the 12th day of September next after which day the deceased's assets will be dis-tributed having regard only to the claims of which notice shall have then been received. And all persons indebted to the said George William Robert Holditch are requested to pay their respective debts to me at once.-Dated this 4th day of July 1938. (125) H TWEED, Horncastle, Solicitor.

H. TWEED, Horncastle, Solicitor. (135)

#### Miss VIOLETTA PITT, Deceased.

Pursuant to the Trustee Act, 1925, Section 27.

Pursuant to the Irustee Act, 1925, Section 27. NOTICE is hereby given that all creditors and other persons having any claims against the estate of Violetta Pitt late of Wellesbourne in the county of Warwick, Spinster, deceased who died on the 27th day of February 1938 and whose Will was proved in the Birmingham District Probate Registry on the 11th day of June 1938 by the Midland Bank Execu-tor and Trustee Company Limited the executors therein named are hereby required to send in par-ticulars in writing of their claim to me the under-signed the Solicitor for the said executors on or before the 9th day of September 1938 after which date the said executors will proceed to distribute date the said executors will proceed to distribute the assets of the said deceased amongst the persons entitled thereto having regard only to debts, claims and demands of which they shall then have had notice and they will not be liable for the assets of the said deceased or any part thereof so distributed to any person or persons of whose claims they shall not then have had notice.—Dated this 29th day of June 1938.

ROBERT LUNN, 2, Sheep Street, Stratford-(113) upon-Avon, Solicitor for the said Executors.

#### Re ELIZA CROOK, Deceased.

#### Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all persons having any debts, claims or demands against the estate of Eliza Crook deceased late of 6 Green Drive Barton Eliza Crook deceased late of 6 Green Drive Barton near Preston in the county of Lancaster Spinster (who died on the 27th day of March 1938) and whose Will was proved in the Lancaster District Probate Registry on the 4th day of June 1938 by John Crook and Elizabeth Thistlethwaite the executors therein named) are required to send particulars in writing of their debts, claims or demands to me, the undersigned, Solicitor for the said executors, on or before the 7th day of September 1938 after which date the executors will proceed to distribute the assets of the said deceased amongst the persons entitled thereto having regard only to the claims of which they shall then have notice, and will not be liable for the assets of the deceased or any part liable for the assets of the deceased or any part thereof so distributed to any person of whose claims or demands they shall not then have had notice.— Dated this 30th day of June 1938.

ROBERT PARKINSON, 7, Leopold Grove, (114) Blackpool, Solicitor for the said Executors.

#### Re PATTIE NAYLOR, Deceased.

#### Pursuant to the Trustee Act; 1925.

NOTICE is hereby given that all creditors and other persons having any debts, claims or demands against the estate of Mrs. Pattie Naylor late of 21 against the estate of Mrs. Pattie Naylor late of 21 (formerly 8) Clarence Street Upper Gornal in the county of Stafford Widow deceased (who died on the 20th day of May 1938 and whose Will was proved in the Birmingham District Registry of the Probate Division of His Majesty's High Court of Justice on the 21st day of June 1938 by Benjamin Lane of 29 Grange Road Dudley in the county of Worcester Manufacturers' Clerk and Kenneth Lane of 56 High Street Sedgley in the said county of Stafford Solici-tor's Clerk the executors therein named) are hereby tor's Clerk the executors therein named) are hereby required to send the particulars in writing, of their debts, claims or demands to us, the undersigned, the Solicitors for the said executors, on or before

the oth day of September next after which date the said executors will proceed to distribute the assets of the said deceased amongst the persons entitled there-to, having regard only to the debts claims and demands of which they shall then have had notice and they will not be liable for the assets of the said deceased, or any part thereof, so distributed, to any person or persons of whose debts claims or demands they shall not then have had notice.—Dated this ist day of July 1938.

SLATER and CAMM, 198, Wolverhampton (115) Street, Dudley, Solicitors for the Executors.

#### ISRAEL GOLD otherwise ISRAEL GOLDSTEIN, Deceased.

#### Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all creditors having any claims or demands against the estate of Israel Gold otherwise Israel Goldstein of 46, Osbaldeston Road, Stoke Newington in the county of Middlesex who died on the 15th day of November 1937 at 46, Osbaldeston Road, Stoke Newington aforesaid and whose Will was proved in the Principal Probate Registry on the 7th day of June 1938 by David Goldstein and Coleman Sharp the executors named therein are hereby required to send particulars in writing of their claims and demands to us the undersigned as Solicitors for the said executors on or before the 9th day of September, 1938, after which date the said executors will proceed to distri-bute the assets of the said deceased amongst the NOTICE is hereby given that all creditors having which date the said executors will proceed to distri-bute the assets of the said deceased amongst the persons entitled thereto having regard only to the debts claims and demands of which they shall then have had notice and they will not be liable for the assets of the said deceased or any part thereof so distributed to any person or persons of whose claims or demands they shall not then have had notice.— Dated this thirtieth day of June 1938.

M. A. JACOBS and SONS, 73-74 Jermyn Street, St. James's, London, S.W.I, Solicitors (025) for the said Executors.

# KENNETH JAMES SAUNDERS, Deceased.

Pursuant to the Trustee Act, 1925, Section 27.

NOTICE is hereby given that all creditors and other persons having any claims or demands against the estate of Kenneth James Saunders late of 41 Blackwater Road, Eastbourne, Sussex and also of High Acres, Berkeley in the county of Alameda California U.S.A. who died on the 19th day of November 1937 and to whose estate and effects letters of administration (with the Will annexed) were granted out of the Principal Probate Registry on the 18th day of May 1938 to William Thomas James as Attorney for American Trust Company, the executor named in the said Will, are hereby required to send particulars thereof in writing to us the undersigned on or before the 9th day of September 1938, after which date the administrator will proceed to distribute the assets of the said will proceed to distribute the assets of the said deceased amongst the persons entitled thereto, having regard only to the claims or demands then notified and will not be liable for the assets of the deceased or any part thereof so distributed to any persons of whose claims or demands he shall not then have notice.—Dated this 1st day of July 1938.

COLES and JAMES, Claremont Chambers, Eastbourne, Solicitors for the said (033) Administrator.

#### Re AGNES FOSBROOKE, Deceased.

ALL persons having claims against the estate of Agnes Fosbrooke of III Great Cheetham Street West Salford Lancashire (who died on 18th July 1937 and probate of whose Will was granted to Thomas Lord Holt and Walter John Humphries on 6th protate of whose will was granted to Inomas Lord Holt and Walter John Humphries on 6th September 1937) are required to send written particulars of their claims to us on or before the 5th September 1938 after which date the executors will proceed to distribute the assets having regard only to the claims then received.—Dated this 30th. June 1938.

GRUNDY, KERSHAW, FARRAR and CO., 31, Booth Street, Manchester 2, Solicitors for (039) the Executors.

Re RUTH SEDGWICK, Deceased, late of Woodhouse Manor Farm, Burnsall, in the West Riding of the county of York, who died on the 12th day of March, 1938.

NOTICE is hereby given that creditors and others having claims against the estate of the above named deceased should give notice thereof in writing to the undersigned the Solicitors to the executors of the Will of the said Ruth Sedgwick on or before the 12th day of September next after which time the executors intend to distribute the estate of the said deceased amongst the parties entitled thereto having regard only to the claims of which notice has then been received by the said Solicitors.—Dated this 30th day of June 1938.

NORMAN LEE, ARMISTEAD and ROBERTS, Lloyds Bank Chambers, Bradford, Solicitors (022) for the Executors.

#### Dr. THOMAS WILLIAM SCALE, Deceased. Pursuant to the Trustee Act, 1925.

ALL persons having claims against the estate of Dr. Thomas William Scale of Gloster House, Aberdare, in the county of Glamorgan, who died on the 17th day of March 1938 (and whose Will was duly proved in the Principal Probate Registry on the 10th day of June, 1938, by Lt.-Col. John Dymoke Scale, D.S.O., and Mrs. Muriel Mary Campbell, two of the executors therein named) are required to send particulars in writing to the undersigned on or before the sixth day of September 1938, after which date the executors will proceed to distribute the assets having regard only to the claims of which they shall then have had notice.—Dated this thirtieth day of June, 1938.

STANLEY RICHARDS and CO., Waycotts' Corner, Paignton, Devon, Solicitors for the said (023) Executors.

#### Re ADA MARY PETTIT, Deceased.

Pursuant to the Trustee Act, 1925.

ALL persons having claims against the estate of Ada Mary Pettit late of 33 Bethlehem Street in the county borough of Grimsby (who died on the 19th day of May 1938 and whose Will was proved by John Alec Gordon Pettit and Herbert Wilkinson Ayre the executors therein named on the 25th day of June 1938 in the Principal Probate Registry) are required to send particulars thereof in writing to the undersigned on or before the 19th day of September next after which date the executors will proceed to distribute the assets having regard only to the claims of which they shall then have had notice.—Dated this 30th day of June 1938.

GEO. A. WHITE and CO., 3, Town Hall Street, (112) Grimsby, Solicitors for the said Executors.

# Mrs. CONSTANCE MARGARET BENETT-DAMPIER, Deceased.

Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all persons having any claim against the estate of Mrs. Constance Margaret Benett-Dampier late of Glaston Hill House, Eversley, Hants, who died on the 3rd day of May 1938 are hereby required to send particulars thereof in writing to us, the undersigned, Solicitors to the executor on or before the roth day of September 1938 after which date the executor will proceed to distribute the said estate, having regard only to the claims then notified.—Dated this 29th day of June 1938.

POTTER CRUNDWELL and BRIDGE, Farnham, Surrey, Solicitors for the Executor. (020)

#### Re THOMAS CORNELIUS JONES, Deceased. Pursuant to the Trustee Act, 1925.

ALL persons having claims against the estate of Thomas Cornelius Jones late of 9 Wrexham Road, Lodge, Brymbo in the county of Denbigh who died on the 7th day of September 1935 and letters of administration to whose estate were granted on the 14th day of October 1937 out of the District Probate Registry at Bangor to John Jones are required to send particulars thereof in writing to the undersigned on or before the 3rd day of September next after which date the administrator will proceed to distribute the assets having regard only to the claims of which he shall then have had notice.—Dated this 30th day of June 1938.

CYRIL JONES, SON and WILLIAMS, 17, Egerton Street, Wrexham, Solicitors to the (038) said Administrator.

#### WILLIAM SLATER, Deceased.

#### Pursuant to the Trustee Act, 1925.

ALL persons having claims against the estate of William Slater late of 27 Linden Grove, Woodsmoor, Stockport, Cheshire, Retired Bank Manager, who died on the 10th May 1938 and probate of whose Will was granted to me by the Manchester District Probate Registry on 8th June 1938 are requested to send me written particulars thereof on or before 9th September 1938 after which day this deceased's assets will be distributed having regard only to claims of which I shall then have had notice.— Dated 5th July 1938.

WILLIAM SCOTT, 64, Cross Street, Manchester 2, Chartered Accountant. (193)

#### HARRIETT UNDERHILL WALKER, Deceased. Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all persons having any claim against the estate of Harriett Underhill Walker late of 132 Great Lister Street, Birmingham, Widow, who died on the 29th day of May 1938 are hereby required to send particulars thereof in writing to the Executor & Trustee Department, Lloyds Bank Limited, 4 Temple Row West, Birmingham, the executor of the Will of the said Harriett Underhill Walker or to the undersigned, the Solicitors to the executor, on or before the 7th day of September 1938, after which date the executor will proceed to distribute the said estate, having regard only to the claims then notified.—Dated this 30th day of June 1938.

JEFFERY WILD and LOVATT, 95, Colmore Row, Birmingham, Solicitors for the said Executor.

(194)

# Re LIONEL NICHOLSON HARDING, Deceased.

Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all persons having claims against the estate of Lionel Nicholson Harding late of South End South Nutfield Surrey who died on the 17th January 1938 intestate and to whose estate letters of administration were granted to Albane Louisa Emma Harding and Gertrude Alana Stenhouse out of the Principal Probate Registry on the 5th May 1938 are required to send particulars thereof to the undersigned before the 9th September 1938 after which date the administratrices will proceed to distribute the said estate having regard only to the claims then notified.—Dated this 4th day of July 1938.

COLBOURNE, BUSH and BARTLETT, 62, Old Steine, Brighton 1, Solicitors for the said (067) Administratices.

#### HARRY HASELER, Deceased.

#### Pursuant to the Trustee Act, 1925.

ALL persons having any claims against the estate of Harry Haseler late of 60, St. Peters Road, Handsworth in the city of Birmingham who died on the 12th day of March 1938 and whose Will was proved in the Birmingham District Probate Registry on the 17th day of June 1938 by Helen Mary Haseler, Nancy Welch and Helen Siddaway the executrices therein named are required to send particulars in writing of their claims to us, the undersigned, on or before the 7th day of September 1938 after which date the said executrices will proceed to distribute the estate amongst the persons entitled thereto, having regard only to the claims of which they shall then have had notice.—Dated this 1st day of July. 1938.

FROST and CO., 57, Colmore Row, Birmingham 3, Solicitors for the Executrices. (195) WALTER WILLIAM WELDON, Deceased (otherwise WILLIAM WALTER WELDON).

Pursuant to the Trustee Act, 1925 (Section 27).

NOTICE is hereby given that all creditors and other persons having any claims or demands against the estate of Walter William Weldon (otherwise William Walter Weldon) late of the White House, Burlington Road in the city of Nottingham who died on the seventh day of May 1938 and whose Will (with three Codicils thereto) was proved in the Nottingham District Registry of the Probate Division of His Majesty's High Court of Justice on the twentieth day of June 1938 by Barclays Bank. Limited and Frederick Harold Weldon the executors therein named are hereby required to send particulars thereof in writing to the Trustee Department Barclays Bank Limited 67 Colmore Row Birmingham or to the undersigned on or before the eighth day of September 1938 after which date the said executors will proceed to distribute the assets of the said deceased amongst the persons entitled thereto having regard only to the claims or demands then notified and will not be liable for the assets of the deceased or any part thereof so distributed to any persons to whose claims or demands they shall not then have had notice.—Dated this second day of July 1938.

BERNARD WRIGHT and CURSHAM, Castle House, Friar Lane, Nottingham, Solicitors (092) for the said Executors.

#### Re ELIZABETH SCHOFIELD, Deceased.

· Pursuant to Section 27 of the Trustee Act, 1925.

NOTICE is hereby given that all creditors and others having any claims against the estate of Elizabeth Schofield late of 33, Kelverlow Street, Oldham in the county of Lancaster, Widow (who died on the 24th day of April 1938 and whose Will was proved on the 10th day of June 1938 by Sarah Elizabeth Armitage and Hanson Buckley the executors therein named) are hereby required to send particulars in writing of their claims to the undersigned on or before the 6th day of September next after which date the executors will distribute the assets of the deceased having regard only to the claims of which they shall then have had notice.—Dated this 2nd day of July 1938.

J. ARNOLD BRIERLEY and ROBINSON, Priory Buildings, Union Street, Oldham. (206)

Re ELLEN RUTH BILLINGSLEY, Deceased, late • of 35, King Street, Darlaston, in the county of • Stafford, Widow (trading as "R. Billingsley & • Co.", Wholesale and Retail Tobacconists), who • died on the 13th January, 1938.

NOTICE is hereby given that creditors and others having claims against the estate of the above deceased letters of administration to whose estate were granted to Gertrude Mary Haynes and Gertrude Buxton Mathews out of the District Probate Registry at Birmingham on the 22nd April 1938, are hereby required to send written particulars of their claims and demands to either of us the undersigned on or before the 9th September 1938, after which date the administratrices will proceed to distribute the assets of the said deceased amongst the persons entitled thereto having regard only to the claims or demands of which notice shall then have been received.— Dated this 1st day of July 1938.

SLATER and CO., Darlaston.

J. F. PARTINGTON, 88-89, Darlington Street, Wolverhampton, Joint Solicitors for the Administratrices.

(205)

#### HENRY STANLEY JARVIS, Deceased.

Pursuant to the Trustee Act, 1925 (Section 27). NOTICE is hereby given that all creditors and other persons having any claims or demands against the estate of Henry Stanley Jarvis late of 25 Foxgrove Avenue Beckenham in the county of Kent formerly of "Hazelwood" 51 The Avenue Beckenham aforesaid who died on the 25th day of April 1938 and whose Will was proved in the Principal Registry of the Probate Division of His Majesty's High Court of Justice on the 18th day of June 1938

by Barclays Bank Limited the executor therein named are hereby required to send particulars thereof in writing to the Trustee Department Barclays Bank Limited 37 King William Street London E.C. or to the undersigned on or before the fifth day of September 1938 after which date the said executor will proceed to distribute the assets of the säid deceased amongst the persons entitled thereto having regard only to the claims or demands then notified and will not be liable for the assets of the deceased or any part thereof so distributed to any persons of whose claims or demands they shall not then have had notice.—Dated this 28th day of June, 1938.

MARCHANT and TIPPER, Westminster Bank Building, Deptford Broadway, ,S.E.8, Solicitors for the said Executor. (160)

## Re ALFRED DAVYE HAYES, Deceased. Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all creditors and others having any claims against the estate of Alfred Davye Hayes late of "Holmelea" Misterton in the county of Nottingham Retired Solicitor who died on the 9th day of February 1938 and whose Will with two Codicils thereto was proved in the Lincoln District Probate Registry on the 21st day of April 1938 by Wesley Francis Brewer, John James Stokes and Charles Frank Richmond the executors therein named are hereby required to send particulars thereof. in writing to the undersigned Solicitors on or before the 5th day of September 1938 after which date the said executors will proceed to distribute the assets of the deceased among the persons entitled thereto having regard only to the claims of which they shall then have had notice; and will not be liable for the assets of the deceased or any part thereof so distributed to any person or persons of whose claims or demands they shall not then have had notice.—Dated this 1st day of July, 1938.

HAYES, SON and RICHMOND, 9, Market Place, Gainsborough, Solicitors for the said (001) Executors.

#### WALTER CASS SMITH, Deceased. JANE SMITH, Deceased.

# Pursuant to the Trustee Act, 1925.

ALL persons having any claims against the estate of Walter Cass Smith who died on the 4th day of March 1938 or against the estate of Jane Smith (his Widow) who died on the 16th day of April 1938 both late of The Haven, Station Road, Scalby near Scarborough in the county of York and whose Wills were proved in the York District Probate Registry on the 12th day of April 1938 and the 28th day of January 1938 respectively are hereby required to send particulars thereof in writing to us, the undersigned, Solicitors for William Henry Nicholson the surviving executor under the said Wills on or before the 8th day of September 1938, after which date the said executor will proceed to distribute the said estates having regard only to the claims of which he shall then have had notice. —Dated this 2nd day of July 1938.

BIRDSALL and SNOWBALL, Bank Chambers, (034) Scarborough, Solicitors for the said Executor.

#### ARTHUR WEBSTER KEELEY, Deceased. Pursuant to the Trustee Act, 1925.

Pursuant to the Trustee Act, 1925. NOTICE is hereby given that all creditors and others having claims against the estate of the above deceased late of 37 Haymarket Lytham St. Annes and of Hutton Grammar School near Preston both in the county of Lancaster Schoolmaster formerly of 51 the Boulevard Lytham St. Annes who died on the 7th day of June 1938 should give notice thereof in writing to the undersigned the Solicitors to the executors of the Will of the said Arthur Webster Keeley on or before the 10th September 1938 after which date the executors will proceed to distribute the estate having regard only to the claims of which day of June, 1938.

T. H. and T. DODD and BUCKLEY, 47, Lune Street, Preston, Lancashire,

(036)

#### ANNIE RHODA GRIFFITHS, Deceased.

Pursuant to the Trustee Act, 1925.

Pursuant to the Trustee Act, 1925. NOTICE is hereby given that all creditors and other persons having any claims or demands against the estate of Annie Rhoda Griffiths late of The Croft, Sheen in the county of Stafford, Spinster, deceased who died on the 27th day of December 1937, and whose Will was proved on the 23rd day of February 1938 by one of the executors therein named, are hereby required to send the particulars in writing of their claims or demands to us the undersigned as Solicitors for the said executors on or before the 8th day of August 1938 after which date the said executors will proceed to distribute the assets of the said deceased amongst the persons entitled thereto having regard only to the debts entitled thereto having regard only to the debts claims and demands of which they shall then have had notice and they will not be liable for the assets of the said deceased or any part thereof so dis-tributed to any person or persons of whose claims or demands they shall not then have had notice.— Dated this 30th day of June 1938.

HINCKLEY BROWN and CRARER, Lichfield, Solicitors for the Executors.

(032)

#### Re JOHN FRANCIS KELLY, Deceased.

Pursuant to Section 27 of the Trustee Act, 1925. NOTICE is hereby given that all persons having any claims against the estate of John Francis Kelly late of 27 Mount Street Eccleshill in the city of Bradford Labourer deceased who died on the 30th day of March 1938 and whose Will was proved in the District Probate Registry at Wakefield on the 38th day of April 1008 by Arthur Hotchin the arcsu 28th day of April 1938 by Arthur Hotchin the execu-tor therein named are required to send particulars of their claims to us the undersigned Solicitors for the said executor on or before the 5th day of September 1938 after which date the said executor will proceed to distribute the assets of the said excettor win proceed the persons entitled thereto having regard only to the claims of which he shall then have had notice.— Dated this 30th day of June 1938.

JAMES A. LEE and PRIESTLEY, 77, Market Street, Bradford, Solicitors for the said (019) Executor.

# Re WALTER MORRITT, Deceased. Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all creditors and others having any claims against the estate of Walter Morritt late of 3, Shoe Market Pontefract in the county of York Confectioner who died on the 21st day of April 1938 and letters of administration to whose estate were granted on the 27th June 1938 out of the Principal Probate Registry to Samuel Morritt of the Principal Probate Registry to Samuel Mornit-are hereby required to send particulars thereof in writing to the undersigned Solicitors on or before the 30th day of September 1938 after which date the said administrator will proceed to distribute the assets of the deceased amongst the beneficiaries en-titled thereto having regard only to the claims of which he shall then have had notice and will not be liable for the assets of the deceased or any part thereof so distributed to any person or persons of whose claims or demands he shall not then have had notice.—Dated this 29th day of June 1938.

MASON and BARKER, Ropergate, Pontefract, Solicitors to the said Administrator. (197)

# GEORGE ALFRED COE, Deceased.

# Pursuant to the Trustee Act, 1925, Section 27 (as amended).

(as amended). , NOTICE is hereby given 1. That all creditors and persons having any claim upon or against the estate of George Alfred Coe of Herringswell Road Kent-ford Suffolk Labourer deceased who died on the zend day of May 1938 and whose Will was duly proved by William Henry Hutchinson the executor therein named on the 18th day of June 1938 in the District Probate Registry at Ipswich are hereby required to send in particulars of their claims to the undersigned Solicitors of the said executor on or before the 1st day of September 1938. z. That after the said 1st day of September 1938 the said

executor will proceed to convey and distribute the assets of the deceased to or among the persons entitled thereto having regard only to the claims of which the said executor shall then have had notice and that he will not be liable for the assets or any part thereof so conveyed or distributed to any persons of whose claim he shall not then have had notice.—Dated this 28th day of June 1938.

BENDALL and SONS, Mildenhall, Suffolk, Solicitors for the Executor. (024)

Re ELLEN BURNETT, Deceased.

#### Pursuant to the Trustee Act, 1925.

NOTICE is hereby given that all creditors and others having any claims against the estate of Ellen Burnett late of 37 St. Edward Street Leek in the county of Stafford Widow who died on the 11th day of May 1938 and whose Will was proved in the Exeter District Probate Registry on the 29th day of June 1938 by Thomas Somerville the executor therein named are hereby required to send par-ticulars thereof in writing to the undersigned Solicitor on or before the 9th day of September 1938 after which date the said executor will proceed to sonctor on or before the gth day of September 1938 after which date the said executor will proceed to distribute the assets of the deceased among the persons entitled thereto having regard only to the claims of which he shall then have had notice; and will not be liable for the assets of the deceased or any part thereof so distributed to any person or persons of whose claims or demands he shall not then have had notice.—Dated this 2nd day of July 1938.

S. HILTON, 15, Victoria Parade, Torquay, (138) Solicitor for the said Executor.

#### CLAIRE ROSALIE CURLIER, Deceased. Pursuant to the Trustee Act, 1925.

ALL persons having claims or demands against the estate of Claire Rosalie Curlier, nee Ducarre, Widow of Felix Joseph Curlier who resided at 21 Boulevard Beausejour Paris France and who died on the 22nd day of January 1937 and to whose estate letters of administration with Wills (as contained in a transadministration with Wills (as contained in a trans-lation thereof) annexed were granted on the 31st day of May 1938 out of the Principal Probate Registry to Joseph Edward Lambert and Gerald Sebastian White the lawful Attorneys of Achille Maurice Revel the person entitled to administer the estate of the said deceased by the law of the place at which deceased died domiciled are requested to send written particulars of such claims to me the under-signed on or before the 6th day of August 1938 after which date the said administrators will proceed to which date the said administrators will proceed to distribute the assets (in England) having regard only to the claims and demands then received.—Dated this rst day of July 1938.

J. E. LAMBERT, 23, Surrey Street, Strand, London, W.C.2, Solicitor for the said Adminis-(021) trators.

#### MARY MARTIN, Deceased.

#### Pursuant to the Trustee Act, 1925, Section 27 (as amended).

NOTICE is hereby given : - (1) That all creditors and persons having any claims upon or against the estate of Mary Martin, late of 27 Belmont Road, Harrogate in the county of York, Widow, who died on the 3rd day of May, 1938, and whose Will was proved by William Ewart Meikle, the executor therein named, on the fourteenth day of June, 1938, in the Newcastle-upon-Tyne District Probate Parity are hereby around the part of particular Registry, are hereby required to send in particulars registry, are hereby required to send in particulars of their claims to the undersigned. Solicitors for the said executor, on or before the 14th day of September, 1938. (2) That after that date the executor will proceed to convey and distribute the assets of the said deceased to or among the persons entitled thereto, having regard only to the claims of which the executor shall then have had notice; and that he will not be liable for the assets, or any part thereof, so conveyed or distributed, to any person of whose claim he shall not then have had notice.-Dated this 30th day of June, 1938.

REED RYDER and MEIKLE, 19, North-umberland Square, North Shields. (035)

#### WILLIAM JOHN SEDDON HUGHES, Deceased. Pursuant to Trustee Act, 1925.

Pursuant to Trustee Act, 1925. NOTICE is hereby given that all creditors and other persons having claims against the estate of William John Seddon Hughes, late of Elmstead, Buckingham Avenue, Shoreham-by-Sea, Brighton, and of Roden Lodge, Wem, Salop (who died on the 6th day of November 1937 and whose Will was proved in the Liverpool District Probate Registry on the 22nd day of January 1938 by Reginald Hamilton Hughes and Cyril Vaughan Hughes the executors named in the said Will), are hereby re-quired to send in particulars of their claims to us, the undersigned, on or before the minth day of September 1938, after which date the said execu-tors will proceed to distribute the assets of the said deceased amongst the persons entitled thereto, havdeceased amongst the persons entitled thereto, hav-ing regard only to the claims of which they shall then have had notice.—Dated this 29th day of June 1938.

BANKS, KENDALL, TAYLOR and GORST, 26, North John Street, Liverpool, Solicitors (209) for the Executors.

# JOHN JACKSON HILTON, Deceased.

Pursuant to the Trustee Act, 1925.

ALL persons having claims against the estate of the above deceased late of The Willows, Whitehaven in the county of Cumberland, who died there on the 25th day of March 1938 and whose Will was proved in the District Probate Registry at Carlisle on the 27th day of May 1938 by Joseph Isaac Hilton the surviving executor therein named are required to send in particulars thereof to us the undersigned within two months from the date hereof after which the executor will proceed to distribute the estate of the said deceased having regard only to claims then received.—Dated this 5th day of July 1938.

HOWSON, DICKINSON and MASON, 143 Queen Street, Whitehaven, Solicitors for the (204) said Executor.

#### Re RICHARD PROCTER, Deceased. Pursuant to the Trustee Act, 1925.

Pursuant to the Trustee Act, 1925. NOTICE is hereby given that all creditors and other persons having any debts claims or demands against the estate of Richard Procter late of 15 Southmount Terrace in the city of Leeds deceased (who died on the 28th day of March 1938) and to whose estate letters of administration were granted out of the Principal Registry of the Probate Division of His Majesty's High Court of Justice on the 4th day of June 1938 to Mrs. Flora Elizabeth Young, are hereby required to send in particulars of their debts, claims or demands, to us the undersigned, the Solicitors for the said administratrix on or before the third day of September 1938, after which date the said administratrix will proceed to dis-tribute the assets of the said deceased amongst the persons entitled thereto, having regard only to the claims and demands of which she shall then have had notice; she will not be liable for the assets of the said deceased or any part thereof so distributed the said deceased or any part thereof so distributed to any person or persons of whose debts claims and demands she shall not then have had notice.—Dated this 30th day of June 1938.

J. H. MILNER and SON, Albion Walk Cham-bers, Leeds 1, Solicitors for the said (203) Administratrix.

#### Re Miss MARY REBECCA LANDCASTLE, Deceased. Pursuant to the Trustee Act, 1925

NOTICE is hereby given that all creditors and other persons having any debts, claims or demands against the estate of Mary Rebecca Landcastle late of "Tregarth," 26 Stocks Lane, Chester in the county of Chester Spinster (who died on the 26th day of February 1938 and whose Will was proved in the Birmingham District Registry of the Probate Division of His Majesty's High Court of Justice on the 13th day of May 1938 by Thomas Leonard Kelly and Frederick Cawrse Landcastle the executors therein named) are hereby required to send in partherein named) are hereby required to send in particulars of their debts, claims or demands in writing to us the undersigned, Solicitors for the said execu-tors on or before the 6th day of September 1938,

after which date the said executors will proceed to distribute the assets of the said deceased amongst the persons entitled thereto, having regard only to the debts, claims and demands of which they shall then have had notice and they will not be liable for the assets of the said deceased or any part thereof so distributed, to any person or persons of whose claims or demands they shall not then have had notice.--Dated this 30th day of June 1938.

REECE, DAVIS and COX, 20A, Temple Street, Birmingham, Solicitors for the said Executors. (196)

#### Re ADA KATHLEEN WATTS, Deceased. Pursuant to the Trustee Act, 1925.

Pursuant to the Trustee Act, 1925. NOTICE is hereby given that all creditors and other persons having any debts claims or demands against the estate of Ada Kathleen Watts late of "Broomfields," 50, Broomfield Avenue Palmers Green in the county of Middlesex deceased Wife of Francis Watts of the same address who died on the 13th day of March 1938 and whose Will was proved in the Principal Probate Registry of His Majesty's High Court of Justice on the 20th day of June 1938 by the said Francis Watts and Horace Tom Best Reed of 18, Haven Drive Bishopstone Herne Bay in the county of Kent the executors therein named of their debts claims or demands to us the under-signed the Solicitors for the said executors on or before the 14th day of September 1938 after which date the said executors will proceed to distribute the assets of the said deceased amongst the persons en-titled thereto having regard only to the debts claims or demands of which they shall then have had notice and they will not be liable for the assets of the said deceased or any part thereof so distributed to any person or persons of whose debts claims or demands they shall not then have had notice.—Dated to any person or persons of whose debts claims or demands they shall not then have had notice.—Dated the said they shall not then have had notice.—Dated

JOHN H. MOTE and SON, 11, Gray's Inn Square, London, W.C.1, Solicitors for the said (003) Executors.

I, NATALIE GRAHAM of Ridgelands College The Ridgeway Wimbledon Surrey Spinster a naturalised British subject, heretofore called and known by the name of Natalia Ivanovna Grouchencoff (otherwise Grushenkova) hereby give notice that I have renounced and abandoned the nonce that I have renounced and abandoned the name of Natalia Ivanovna Grouchencoff (otherwise Grushenkova) and that I have assumed and intend henceforth on all occasions whatsoever and at all times to sign and use and to be called and known by the name of Natalie Graham in lieu of and in what is the name of Natalie Graham in lieu of and in substitution for my former name of Natalia Ivanovna Grouchencoff (otherwise Grushenkova) and I also hereby give notice that such change of name is formally declared and evidenced by a deed poll under my hand and seal dated the twenty-ninth day of June one thousand nine hundred and thirty-eight duly executed and attested, and that such deed poll was enrolled in the Central Office of the Supreme Court of Judicature on the thirtieth day of June court of Judicature on the thirtieth day of June one thousand nine hundred and thirty-eight.—Dated this thirtieth day of June this thirtieth day of June 1938.

N. GRAHAM, formerly N. I. Grouchencoff or (171) N. I. Grushenkova.

NOTICE is hereby given that by a deed poll dated the twentieth day of June one thousand nine dated the twentieth day of june one thousand nine hundred and thirty-eight and duly enrolled in the Supreme Court of Judicature on the twenty-eight day of June one thousand nine hundred and thirty-eight Gertrude Melville Melville of "The Ridgeway," Waxwell Lane, Pinner in the county of Middlesex Widow a natural born British subject and the Widow of a natural born British subject did for herself and her children Reginald Laurence Melville and Gertrude Constance Melville and remoter issue renounce and abandon the surname of Catt.—Dated the first day of July one thousand nine hundred and thirty-eight.

FITZHUGH WOOLLEY BAINES and CO., 3, Pavilion Parade, Brighton 1, Solicitors for the (103) said Gertrude Melville Melville.

I, ITHAMAR GOBEL COPE, hereby give notice that by a deed poll dated the 25th day of June 1938 enrolled in the Central Office of the Supreme Court of Judicature I renounced and abandoned my former surname of Cornish and adopted the surname of Cope by which name I intend at all times here-after to be known and described.—Dated this second day of July 1938.

ITHAMAR GOBEL COPE, formerly Ithamar (179) Gobel Cornish.

NOTICE is hereby given that by a deed poll dated the 22nd day of June 1938 and duly enrolled in the Supreme Court of Judicature on the first day of July 1938 BARBARA LYNTON of 183 Cavendish Road Bispham Blackpool Lancashire Spinster a natural born British subject renounced and abandoned the surname of Fielding.—Dated the first day of July 1938.

FIELD CUNNINGHAM and CO., 5, John Dalton Street, Manchester 2, Solicitors for the (049) said Barbara Lynton.

NOTICE is hereby given that by a deed poll dated the 28th day of June 1938 and duly enrolled in the Supreme Court of Judicature on the 29th day of June 1938 I, FRANK SACHS of 7, Montrose Court, Finchley Road in the county of London Company Director a natural born British subject renounced and abandoned the first names of Isaac Hermann.—Dated the 30th day of June 1938.

A. E. SAMUELS and CO., 30-38, New Broad Street, London, E.C.2, Solicitors for the said (026) Frank Sachs.

TO be Sold, pursuant to an Order of the High Court of Justice, Chancery Division, made in the Matter of DOROTHY ADA GROVES and In the Matter of the Judgments Act 1864, 1936 G. No. 1142 with the approbation of Mr. Justice Morton, by Mr. Frank Hayward, the person appointed by the said Judge, at the Grand Hotel, Colmore Row, Birmingham in the county of Warwick on the 21st day of July 1938 at 4 o'clock in the afternoon in One Lot, a certain freehold residence, No. 152 Brooklands Road, Hall Green, Birmingham in the county of Warwick and containing three bedrooms, bathroom, separate w.c. and the usual out-offices and garden in rear and space for garage at side. Particulars and conditions of sale may be obtained

Particulars and conditions of sale may be obtained of Messrs. Douglas Fearn & Co. of 3-4, Clement's Inn, London, W.C.2, Solicitors; of the Auctioneer at 39-40 Temple Street, Birmingham; and at the place of sale.

Dated this 4th day of July 1938.

A. H. HOLLAND, Master. (212)

In the High Court of Justice.-Chancery Division. Mr. Justice Crossman.

1936. G. 701.

Re GEORGES Limited.

THOMAS OSWALD WILLIAMS v. THE COMPANY.

By Order of the High Court dated the 14th October, 1936.

Notice of alteration of place and time of Sale.

THE place and time of Sale of Freehold Commercial Property being Nos. 1 and 2 Beauchamp Commercial Property being Nos. 1 and 2 Beauchamp Terrace Worcester Road Great Malvern consisting of a Shop and Bakery and Trade Fixtures advertised in issue of the London Gazette dated the 28th June 1938 as taking place at the Foley Arms Hotel Malvern Worcestershire on Thursday the 28th July 1938 at 7 p.m. has been altered and Messrs. Engall Cox & Co. are now instructed to submit the same property to auction at the Beauchamp Hotel, Graham Road, Malvern Worcestershire on Thursday the 28th July 1938 at 7.30 p.m. precisely. Particulars and Conditions of Sale may be obtained of Messrs. Winterbotham Guroy & Co. of 4 Rodney

Road Cheltenham in the county of Gloucester Solici-tors phone Gloucester 5147-8 Messrs. Engall Cox & Co. 21 Promenade Cheltenham in the county of Gloucester Auctioneers phone Cheltenham 26641 and of Mr. Christopher Vincent Smith of 15 Rodney Road Cheltenham in the county of Gloucester phone Cheltenham 2142 and at the new place of Sale.

M. G. WILLMOTT, Master. (183)

#### DAVID PATRICK WALSH, Deceased.

PURSUANT to an Order of the Court of Chancery PURSUANT to an Order of the Court of Chancery of the County Palatine of Lancaster, Manchester District dated the 2nd day of June 1938 made in an action David Joseph Jones against Marion Walsh (Widow) and Florence Mary Brown (Married Woman) 1938 Letter J. No. 70 the creditors of David Patrick Walsh late of "Wynton" Washway Road Sale in the county of Chester and 78 King Street Manchester in the county of Lancaster Chartered Accountant who died on the 12th day of February 1938 are on or before the 5th day of September 1938 reconnective who died on the 12th day of February 1938 are on or before the 5th day of September 1938 to send by post, prepaid, to Mr. Denis Hickey of Bank of England Chambers, Tib Lane Manchester in the county of Lancaster the Solicitor of the defendants Marion Walsh (Widow) and Florence Mary Brown (Married Woman) the administratrices of the deceased, their Christian and surnames, addressed and descriptions the full acting the of the deceased, their Christian and surnames, addresses and descriptions, the full particulars of their claims, a statement of their accounts, and the their claims, a statement of their accounts, and the nature of the securities (if any) held by them, or in default thereof, they will be peremptorily excluded from the benefit of the said estate. Every creditor holding any security is to produce the same before the Registrar of the Manchester District at his Chambers, situate at 4 Duchy Chambers Clarence Street Manchester aforesaid on Friday the 16th day of September 1938 at 11 o'clock in the forenoon, being the time appointed for adjudication on the claims. claims.

Dated this 28th day of June 1938.

HAROLD ROBERTS, Registrar.

BOOTE DUTTON and WHITTAKER, 12, 14 and 16, Booth Street, Manchester 2, Solicitors (040) for the Plaintiff.

In the High Court of Justice .- Chancery Division. Mr. Justice Simonds.

No. 00456 of 1938.

In the Matter of BRAZILIAN WARRANT AGENCY AND FINANCE COMPANY Limited, and in the Matter of the Companies Act, 1929.

Matter of the Companies Act, 1929. NOTICE is hereby given that by an Order dated the 30th day of June 1938, the Court has directed separate Meetings to be convened of the holders of the classes of shares of the above named Com-pany specified in the first column of the Schedule hereto for the purpose of considering and if thought fit approving (with or without modification) a Scheme of Arrangement proposed to be made be-tween the above named Company and the holders of such shares and that such Meetings will be held at Winchester House Old Broad Street London E.C.2, on Wednesday the 27th day of July 1938 at the respective times specified in the second column of the said Schedule hereto at which place and respective times all such holders are requested to attend. attend.

A copy of the said Scheme of Arrangement can be seen and forms of proxy applicable for the respec-tive Meetings may be obtained at the registered office of the Company situate at No. 20 King William Street London E.C.4, and at the office of the under-mentioned Solicitors at the address mentioned below

mentioned Solicitors at the address mentioned below during usual business hours on any week day prior to the day appointed for the said Meetings. Any such holder may attend the Meeting of any class of which he is a member and vote thereat either in person or by proxy but it is requested that all forms appointing proxies be lodged with the Secretary at the registered office of the Company

at No. 20 King William Street aforesaid not later than 12 o'clock (noon) on Tuesday the 26th day of July 1938, but if such forms are not so lodged they must be handed to the Chairman at the Meet-ing at which they are to be used. In the case of joint holders the vote of the senior who tenders a vote, whether in person or by proxy, will be accepted to the exclusion of the votes of the other joint holders, and for this purpose seniority will be determined by the order in which the names stand in the Register of Members of the Company. Company.

By the said Order the Court has appointed Arthur Whitworth or failing him Archibald Knightley Graham to act as Chairman of each of the said Meetings and has directed the Chairman to report the results of the said Meetings respectively to the

Court. The said Scheme of Arrangement will be subject to the subsequent approval of the Court.

THE SCHEDULE.

Particulars of Class Meet- ings ordered to be convened.	Time appointed for Meeting on Wednes- day, the 27th day of July, 1938.
(1) Holders of the Preference Shares.	12.15 o'clock in the afternoon or so soon thereafter as the Annual General Meeting shall have been concluded.
(2) Holders of the Ordinary Shares.	12.25 o'clock in the afternoon or so soon thereafter as the preceding Meeting shall have been con- cluded.

Dated this 4th day of July 1938.

SLAUGHTER and MAY, of No. 18, Austin Friars, London, E.C.2, Solicitors for the above (096) named Company.

#### BOROID Limited.

BOROID Limited. NOTICE is hereby given that a Meeting of the holders of the 6 per cent. Debentures to bearer of the above named Company will be held at 5, Bishopsgate, London, E.C.2, on Thursday, the 14th day of July, 1938 at 2.30 o'clock in the afternoon to receive from the Receiver appointed on behalf of such debenture holders on the 12th day of May, 1914 by the Trustees of the Trust Deed dated the 14th day of April, 1914 securing the said deben-tures a report of the conduct of the receivership and an account of his receipts and payments in connection therewith and to close the receivership. —Dated this 5th day of July, 1938.

L. C. BULLOCK, Receiver for the Debenture Holders.

Bishopsgate

(057) London, E.C.2.

#### FRANCIS JOHN CORNISH, Deceased.

INFORMATION is desired by the executor of the above deceased, who died at Penzance, Cornwall, on 15th February 1937, with regard to one Evelina Roberts, Daughter of William Roberts formerly of Trengothal, St. Levan, Cornwall, who subsequently married a person, whose name is unknown, a Telegraph Company Employee at Porthcurno, Cornwall, and who, it is believed, afterwards went to live at Ealing. The above named Evelina Roberts or any Ealing. The above named Evenua Autoria of her descendants are required to communicate with the undersigned.

BORLASE and VENNING, Solicitors, Penzance. (151)

#### H.M. LAND REGISTRY.

#### Freehold Title No. Z700Z.

Linden, 4, Beechcroft Gardens, Wembley, Middlesex.

IT is proposed to issue a new Land Certificate to William Rice of 25, Butler Avenue, Harrow, Middle-sex in place of one stated to have been lost. Any person possessing the missing Certificate or objecting to the issue of a new one should at once notify "H.M. Land Registry, London, W.C."

# SOUTH YORKSHIRE MINES DRAINAGE COMMITTEE.

NOTICE is hereby given that as from the 11th day of July 1938, the office of the above named Committee will be situated at "Westfield House," Westfield Road, Parkgate, Rotherham, Yorkshire, instead of at Imperial Buildings, Rotherham, York-shire.—Dated this 2nd day of July, 1938.

PARKER RHODES COCKBURN and CO., 14, Moorgate Street, Rotherham, Solicitors for the (198) said Committee.

# 3, Dean's Yard, Westminster, S.W.1, 5th July, 1938.

NOTICE is hereby given, pursuant to Charter of 3rd year of her late Majesty Queen Anne, that a General Court of the Governors of Queen Anne's Bounty will be held in their Board Room at the above address, on Thursday, the 21st July, at three o'clock, for the despatch of general business.

W. G. HANNAH, Secretary.

#### The NATIONAL BANK Limited,

13, Old Broad Street, London, E.C.2,

5th July, 1938.

NOTICE is hereby given that the next Half-yearly General Meeting of the Proprietors of the National Bank Limited will be held at the office of the Bank, No. 34 College Green, in the city of Dublin, on Thursday, the 28th day of July, 1938, at 12 noon of that day. At this Meeting the directors will submit their report and balance sheet for the half-year to 30th June, 1938.

# By Order of the Court of Directors, E. S. GOADBY, Secretary.

N.B.—The books of the Bank for the Registration of Transfer Deeds (both in London and Dublin) will be closed on Monday the 11th day of July, 1938, and will so remain until after Saturday, the 23rd day of the same month. (010)

.

In the Matter of a Deed of Assignment for the benefit of creditors executed on the 27th day of June, 1938, by Henry Thomas Stroud of 149, Plashet Grove, East Ham, in the county of Essex, carrying on business as "Jubilee Supply Com-pany," Credit Draper.

PERSONS having claims against the debtor who have not yet done so are required to send par-ticulars thereof to us the undersigned or to Messrs. Beecroft Sons & Nicholson of 12 Wood Street, Cheap-side, London, E.C.2 before the 3rd day of October 1938 otherwise they will be excluded from dividend under the said Assignment.—Dated the 1st day of July 1938.

N. MASON and CO., 41-46, Temple Chambers, Temple Avenue, London, E.C.4, (170) Solicitors for Mr. Thomas Taylor, the Trustee.

In the High Court of Justice.-In Bankruptcy. No. 322 of 1938.

In the Matter of a Bankruptcy Petition filed the 29th day of April, 1938.

To L. WIESENFELD (Male), of 175, High Road, Kilburn, in the county of London, Children's Outfitter.

TAKE notice, that a bankruptcy petition has been presented against you in this Court by M. & J. Toeman Limited of 3, Wood Street, Cheapside, E.C. Toeman Limited of 3, Wood Street, Cheapside, E.C. in the city of London, Hosiery Manufacturers and the Court has ordered that the publication of this notice in the London Gazette and in the Daily Telegraph and Morning Post newspaper shall be deemed to be service of the petition upon you; and further take notice that the said petition will be heard at this Court on the 14th day of July 1938 at 11 o'clock in the forenoon, on which day you are required to appear, and if you do not appear the Court may make a Receiving Order against you in your absence. The petition can be inspected by you on application at this Court.—Dated this 1st day of July 1938. (045) CYRIL L PARTON Registrar

(045) CYRIL J. PARTON, Registrar.

The Bankruptcy Acts, 1914 and 1926. In the Huddersfield County Court.-In Bankruptcy. No. 8 of 1937.

Re BENJAMIN HILTON and MARY JANE HILTON e BENJAMIN HILTON and MARTI JANE HILTON (his Wife), residing at 46, Bradley Street, Hud-dersfield, in the county of York, carrying on busi-ness in co-partnership at that address under the style or firm of "Hilton's," Grocers and General Dealers.

NOTICE is hereby given that there being in the hands of the Trustee in the above bankruptcy a surplus estimated at fr2 arising from the separate estate of Mary Jane Hilton one of the bankrupts, and there being no separate creditors of such bank-rupt, it is the intention of the Official Receiver, at the expiration of fourteen days from the appearance of this notice in the Gazette, to transfer such surplus to the credit of the joint estate in the said bank-rupty.—Dated this 30th day of June, 1938.

W. F. CRESSWELL, Hallfield Chambers, Manningham Lane, Bradford, Official Receiver and Trustee.

In the Liverpool County Court .-- In Bankruptcy. No. 62 of 1936.

Re HAROLD AUBREY SIMPSON and EVA DYSON (Married Woman), both residing at 13, Crescent Road, Birkdale, Southport, in the county of Lancaster, and trading in co-partnership to-gether at 18, Eastbank Street, Southport aforesaid, under the style of "Simpsons," Radio and Elec-tricol Engineer trical Engineers.

NOTICE is hereby given that there being in the hands of the Trustee in the above bankruptcy a

surplus estimated at  $f_{22}$  is. IId. arising from the separate estate of Harold Aubrey Simpson, one of the bankrupts, and there being no separate creditors of such bankrupt, it is the intention of such Trustee at the expiration of 14 days from the appearance of this notice in the London Gazette, to transfer such surplus to the credit of the joint estate in the said bankruptcy.—Dated this 1st day of July 1938.

W. G. LITHGOW, Bank Chambers, 413, Lord Street, Southport, Trustee.

#### The Bankruptcy Acts, 1914 and 1926.

In the Huddersfield County Court .- In Bankruptcy. No. 8 of 1937.

Re BENJAMIN HILTON and MARY JANE HILTON (his Wife), residing at 46, Bradley Street, Hud-dersfield, in the county of York, carrying on busi-ness in co-partnership at that address under the style or firm of "Hilton's," Grocers and General Decler Dealers.

NOTICE is hereby given that there being in the hands of the Trustee in the above bankruptcy a surplus estimated at  $f_3$  15s. arising from the separate estate of Benjamin Hilton one of the bankrupts, and estate of Benjamin Hilton one of the bankrupts, and there being no separate creditors of such bankrupt, it is the intention of the Official Receiver, at the expiration of fourteen days from the appearance of this notice in the Gazette, to transfer such surplus to the credit of the joint estate in the said bank-ruptcy.—Dated this 30th day of June, 1938.

W. F. CRESSWELL, Hallfield Chambers, 71, Manningham Lane, Bradford, Official Receiver and Trustee.

The Bankruptcy Acts, 1914 and 1926.

In the Bournemouth County Court.-In Bankruptcy. No. 32 of 1937.

Re EMMA WINTER GILLINGHAM, Wife of William Leonard Kay Gillingham; THOMAS HENRY HAZELDINE, Waiter, and DOROTHY MABEL HAZELDINE, Wife of Thomas Henry Hazeldine, all of "Kerri," Seacombe Road, Sand-banks, in the county of Dorset, and carrying on business in partnership at Bay View Stores, Banks Road, Sandbanks aforesaid, under the name or style of "Gillingham and Hazeldine," Grocers and Provision Merchants. EMMA WINTER GILLINGHAM, Wife of 'illiam Leonard Kay Gillingham; THOMAS Provision Merchants.

NOTICE is hereby given that there being in the hands of the Trustee in the above bankruptcy a surplus estimated at  $\frac{1}{2}$ 20 7s. rod. arising from the separate estate of Dorothy Mabel Hazeldine one of the bankrupts, and there being no separate creditors of such bankrupt, it is the intention of the Official Receiver, at the expiration of fourteen days from the appearance of this notice in the Gazette, to in the said bankruptcy.—Dated this and day of July 1938.

A. L. MEDCALF, ro, Rockstone Place, South-ampton, Official Receiver and Trustee.

# THE LONDON GAZETTE, 5 JULY, 1938

#### BANKRUPTCY ACTS. THE 1914 AND 1926.

•

**RECEIVING ORDERS.** 

- No. 2,085. BROZEN, Jack Edward, 11, Nevilles Court, London, N.W.2, and lately 30, Cres-pigny Road, London, N.W.4. INSURANCE pigny Road ASSESSOR

  - ASSESSOR. Court—HIGH COURT OF JUSTICE. Date of Filing Petition—July 1, 1938. No. of Matter—536 of 1938. Date of Receiving Order—July 1, 1938. No. of Receiving Order—July 1, 1938. No. of Receiving Order—330. Whether Debtor's or Creditor's Petition—Debtor's.
- No. 2,086. COHEN, L. (Male), carrying on business at 422, Liverpool Road, Holloway, London. SCRAP METAL MERCHANT. Court—HIGH COURT OF JUSTICE. Date of Filing Petition—June 3, 1938. No. of Matter—448 of 1938. Date of Receiving Order—July 1, 1938. No. of Receiving Order—July 1, 1938. No. of Receiving Order—331. Whether Debtor's or Creditor's Petition— Creditor's

  - Creditor's.
  - Act of Bankruptcy proved in Creditor's Petition-Section 1-1 (G.), Bankruptcy Act, 1914.
- 2,087. FITZGERALD, Desmond William, carrying on business at Goswell Works, Strat-ford, Essex, and residing at Stuart Private Hotel, Knole Road, Bexhill-on-Sea, Sussex, and lately residing at 6, Marina Court Avenue, Bexhill-on-Sea aforesaid. COMMERCIAL No. 2,087. AGENT.
  - Court—HIGH COURT OF JUSTICE.

  - Date of Filing Petition—June 10, 1938. No. of Matter—463 of 1938. Date of Receiving Order—July 1, 1938. No. of Receiving Order—332. Whether Debtor's or Creditor's Petition— Creditor's.
  - Act of Bankruptcy proved in Creditor's Petition-Section 1-1 (G.), Bankruptcy Act, 1914.
- No. 2,088. H. & S. BUILDING CO. (a firm), carry-ing on business at 47B, Redcliffe Square, Earls Court, London, W. BUILDERS. Court-HIGH COURT OF JUSTICE.

  - Date of Filing Petition—May 27, 1938. No. of Matter—428 of 1938. Date of Receiving Order—July 1, 1938. No. of Receiving Order—329. Whether Debtor's or Creditor's F Creditor's Petition-Creditor's.
  - Act of Bankruptcy proved in Creditor's Petition---Section 1-1 (G.), Bankruptcy Act, 1914.
- No. 2,089. PARNESS, Samuel, residing at 2, Courtleigh Bridge Lane, London, N.W.II. Description not known.
  Court—HIGH COURT OF JUSTICE.
  Date of Filing Petition—March 16, 1938.
  No. of Matter—216 of 1938.
  Date of Receiving Order—June 22, 1938.
  No. of Receiving Order—305.
  Whether Debtor's or Creditor's Petition— Creditor's.

  - Creditor's.
  - Act of Bankruptcy proved in Creditor's Petition-Section 1-1 (G.), Bankruptcy Act, 1914.
- No. 2,089. PASK, A. E. (Male), lately carrying on business at and residing at 172, Warren Drive, Chelsfield, Kent, present address unknown to Petitioning Creditors. DRAPER.
  Court—HIGH COURT OF JUSTICE.
  Date of Filing Petition—June 8, 1938.
  No. of Matter—453 of 1938.
  Date of Receiving Order—June 30, 1938.
  No. of Receiving Order—324.
  Whether Debtor's or Creditor's Petition— Creditor's.
  Act of Bankruptcy proved in Creditor's Petition—

  - Act of Bankruptcy proved in Creditor's Petition-Section 1-1 (D.), Bankruptcy Act, 1914.

- No. 2,090. SMITH, Coleman, of and residing at 27, Stockleigh Hall, Northgate, London, N.W.8. COMPANY DIRECTOR.
  Court—HIGH COURT OF JUSTICE.
  Date of Filing Petition—Dec. 13, 1937.
  No. of Matter—976 of 1937.
  Date of Receiving Order—July 1, 1938.
  No. of Receiving Order—July 1, 1938.
  Whether Debtor's or Creditor's Petition—Creditor's.
  Act of Bankruptcy proved in Creditor's Petition—

  - Act of Bankruptcy proved in Creditor's Petition-Section 1-1 (G.), Bankruptcy Act, 1914.
- No. 2,091. TUGWELL, Herbert Frederick Lewen, The Drill Hall, Church Street, Brighton, Sussex, BRIGADE MAJOR, and lately residing at Headquarters, Royal Artillery, The London Division, T.A., Finsbury Barracks, London, E.C.

  - Division, T.A., Finsbury Barracks, E.C. Court—HIGH COURT OF JUSTICE. Date of Filing Petition—May 13, 1938. No. of Matter—386 of 1938. Date of Receiving Order—June 30, 1938. No. of Receiving Order—325. Whether Debtor's or Creditor's P Creditor's. Act of Bankruptcy proved in Creditor's P Petition-
  - Act of Bankruptcy proved in Creditor's Petition-Section 1-1 (G.), Bankruptcy Act, 1914.
- No. 2,092. WINICK, Isaac, residing and carrying on business at 68-70, Christian Street, London, <u>E.I.</u><u>WHOLESALE</u> FURNITURE MANUon business at 08-70, Christian Street, E. I. WHOLESALE FURNITURE FACTURER. Court—HIGH COURT OF JUSTICE. Date of Filing Petition—May 31, 1938. No. of Matter—435 of 1938. Date of Receiving Order—June 30, 1938. No. of Receiving Order—June 30, 1938. No. of Receiving Order—426. Whether Debtor's or Creditor's P Creditor's.

  - Petition-
  - Act of Bankruptcy proved in Creditor's Petition— Section 1-1 (G.), Bankruptcy Act, 1914.
- No. 2,093. WOOD, Thomas Rigby, trading as T. R. WOOD & CO., ESTATE AGENT, at 41-42, Chandos House, Buckingham Gate, London, and residing at 27, Devon Road, Cheam, Surrey.
  Court—HIGH COURT OR JUSTICE.
  Date of Filing Petition—May 24, 1938.
  No. of Matter—415 of 1938.
  Date of Receiving Order—June 30, 1938.
  No. of Receiving Order—327.
  Whether Debtor's or Creditor's Petition— Creditor's.
  Act of Bankruptcy proved in Creditor's Petition—

  - Act of Bankruptcy proved in Creditor's Petition-Section 1-1 (G.), Bankruptcy Act, 1914.
- No. 2,094. BROADBENT, Mary Alice (Married Woman), residing at 71, Ladbrooke Road, Small-shaw, Ashton-under-Lyne, in the county of Lancaster, and formerly residing and carrying on business at 1, Silas Street, Hurst, Ashton-under-Lyne aforesaid, as a GROCER and MIXED GOODS DEALER.
  Court—ASHTON-UNDER-LYNE and STALY-BRIDGE.
  Date of Filing Petition—July 7, 2028

  - Date of Filing Petition—July 1, 1938. No. of Matter—5 of 1938. Date of Receiving Order—July 1, 1938. No. of Receiving Order—July 1, 1938. Whether Debtor's or Creditor's F Debtor's. Petition-
- No. 2,095. GRUNDY, Frank Neville, 210, Coventry Road, Coleshill, in the county of Warwick, Farm Worker, formerly residing and carrying on busi-ness at Blake Street Farm, Sutton Coldfield, in the county of Warwick. MARKET GARDENER and DAIRYMAN.
  Court—BIRMINGHAM.
  Date of Filing Petition—July 1, 1938.
  No. of Matter—33 of 1938.
  Date of Receiving Order—July 1, 1938.
  No. of Receiving Order—27.
  Whether Debtor's or Creditor's Petition— Debtor's.

- Debtor's.

- 2,096. WATKINS, Lydia, residing in rooms at 15, Second Avenue, Selly Park, Birming-ham, in the county of Warwick, and carrying on business at Guildhall Buildings, 11, Naviga-tion Street, and 37, Stephenson Street, Birming-ham, in the county of Warwick, as a FRUITERER and CONFECTIONER. No. 2,096. Court-BIRMINGHAM.

  - Date of Filing Petition—June 30, 1938. No. of Matter—32 of 1938. Date of Receiving Order—June 30, 1938. No. of Receiving Order—26. Whether Debtor's or Creditor's Petition— Debtor's.
- No. 2,097. PEACE, Clifford, residing at 4, Ferndene, Bingley, in the county of York, lately residing at 6, Trinity Place, Bingley aforesaid. TEXTILE OPERATIVE.
  Court-BRADFORD.
  Date of Filing Petition-June 30, 1938.
  No. of Matter-34 of 1938.
  Date of Receiving Order-June 30, 1938.
  No. of Receiving Order-31.
  Whether Debtor's or Creditor's Petition-Debtor's.

  - Debtor's.
- 2,098. HANWORTH SAND AND BALLAST SUPPLIES (a firm), Swan Yard, Hanworth, Middlesex. SAND and BALLAST MER-No. 2,098. CHANTS
  - Court-BRENTFORD.
  - Date of Filing Petition-May 25, 1938.

  - No. of Matter—11 of 1938. Date of Receiving Order—July 1, 1938. No. of Receiving Order—11. Whether Debtor's or Creditor's Petition-Creditor's.
  - Act of Bankruptcy proved in Creditor's Petition-Section 1-1 (G.), Bankruptcy Act, 1914.
- No. 2,099. CURTIS, Sidney, "Broadways," Lake Road, Henleaze, in the city and county of Bristol. BUILDER. Court-BRISTOL.

  - Court—BRISTOL. Date of Filing Petition—May 31, 1938. No. of Matter—24 of 1938. Date of Receiving Order—June 29, 1938. No. of Receiving Order—27. Whether Debtor's or Creditor's Petition— Creditor's.
  - Act of Bankruptcy proved in Creditor's Petition-Section 1-1 (A.), Bankruptcy Act, 1914.
- o. 2,100. WELLINGS, Dennis, residing at 11, Hickson Street, Barnton, near Northwich, in the county of Chester, formerly 18, Moss Road, Winnington, Northwich, and 37, Middlewich Road, Northwich, in the county of Chester. CHEMICAL LABOURER.
  Court—CREWE, NANTWICH and SANDBACH.
  Date of Filing Petition—June 30, 1938.
  No. of Matter—6 of 1938.
  Date of Receiving Order—June 30, 1938.
  No. of Receiving Order—6.
  Whether Debtor's or Creditor's Petition— Debtor's. No. 2,100.

  - Debtor's.
- No. 2,101. MINION, John Edward, 58, Church Street, in the county borough of Derby. CYCLE SHOP PROPRIETOR.
  Court—DERBY and LONG EATON.
  Date of Filing Petition—June 30, 1938.
  No. of Matter—18 of 1938.
  Date of Receiving Order—June 30, 1938.
  No. of Receiving Order—16.
  Whether Debtor's or Creditor's Petition—

  - Whether Debtor's or Debtor's. Creditor's Petition—
- No. 2,102. COATES, Henry, residing at 1, Copley, Butterknowle, in the county of Durham, and carrying on business at 1, Copley, Butterknowle aforesaid. COAL MERCHANT. Court-DURHAM.

- Date of Filing Petition—July 1, 1938. No. of Matter—4 of 1938. Date of Receiving Order—July 1, 1938. No. of Receiving Order—4. Whether Debtor's or Creditor's P -4. Creditor's Petition—
- Debtor's.
- No. 2,103. SANGWIN, Albert Edward, 28, Barley Mount, St. Thomas, Exeter, formerly residing and carrying on business at Roseland Dairy, Newcombe Terrace, Heavitree, Exeter aforesaid. Clerk, formerly DAIRYMAN. Court—EXETER.

  - Court—EARTER. Date of Filing Petition—June 20, 1938. No. of Matter—6 of 1938. Date of Receiving Order—July 1, 1938. No. of Receiving Order—7. Whether Debtor's or Creditor's Petition— Creditor's.
  - Act of Bankruptcy proved in Creditor's Petition-Section 1-1 (G.), Bankruptcy Act, 1914.
- No. 2,104. ANDERSON, Mervyn Kendall, The Mill Bakeries, Trusthorpe, Mablethorpe, Lincoln-shire, BAKER, and formerly of Seabank House, Gibraltar Road, Mablethorpe.
  Court—GREAT GRIMSBY.
  Date of Filing Petition—July 1, 1938.
  No. of Matter—19 of 1938.
  Date of Receiving Order—July 1, 1938.
  No. of Receiving Order—19.
  Whether Debtor's or Creditor's Petition— Debtor's.
- No. 2,105. HARRISON, Leonard, Travellers Rest Inn, Sowerby, Sowerby Bridge, in the county of York. LICENSED VICTUALLER and FARM LABOURER.

  - FARM LABOURER. Court—HALIFAX. Date of Filing Petition—July 1, 1938. No. of Matter—13 of 1938. Date of Receiving Order—July 1, 1938. No. of Receiving Order—11. Whether Debtor's or Creditor's Petition— Debtor's Debtor's.
- No. 2,106. BROWN, Eric Frederick, now residing at 20, Cheltenham Mount, Harrogate, in the county of York, and carrying on business at 15, Cambridge Street, Harrogate aforesaid, as a FRUITERER and FLORIST, under the style of "BROWN AND BEST." Court—HARROGATE. Data of Elina Battion June 20, 1008

  - court—HARROGATE. Date of Filing Petition—June 30, 1938. No. of Matter—9 of 1938 Date of Receiving Order—June 30, 1938. No. of Receiving Order—8. Whether Debtor's or Creditor's Pet Debtor's. Creditor's Petition-
- o. 2,107. CURTIS, Cecil, lately residing at The Imperial Hotel, Widemarsh Street, in the city of Hereford, and now residing in apartments at "St. Hyldas," Aylestone Hill, in the said city. COMMERCIAL TRAVELLER. Court—HEREFORD. No. 2,107.

  - Court—HEREFORD. Date of Filing Petition—July 2, 1938. No. of Matter—7 of 1938. Date of Receiving Order—July 2, 1938. No. of Receiving Order—7. Whether Debtor's or Creditor's Petition— Debtor's.
- No. 2,108. ROBERTSON, James Bruce, "Freshfield," Boyle Farm, Thames Ditton, Surrey, AGENT and CONTRACTOR, lately carrying on business at Willing House, 356-360, Gray's Inn Road, London, W.C.
  Court—KINGSTON (Surrey).
  Date of Filing Petition—May 28, 1938.
  No. of Matter—16 of 1938.
  Date of Receiving Order—June 29, 1938.
  No. of Receiving Order—14.
  Whether Debtor's or Creditor's Petition—Creditor's.

  - Act of Bankruptcy proved in Creditor's Petition-Section 1-1 (G.), Bankruptcy Act, 1914.

- Io. 2,109. BARKER, E. A., AND SONS (firm), Spen Farm, Headingley, Leeds I, in the county of York. MILK RETAILERS.
  Court—LEEDS.
  Date of Filing Petition—June 2, 1938.
  No. of Matter—29 of 1938.
  Date of Receiving Order—June 30, 1938.
  No. of Receiving Order—29.
  Whether Debtor's or Creditor's Petition— Creditor's. No. 2,109.

  - Creditor's.
  - Act of Bankruptcy proved in Creditor's Petition-Section 1-I (G.), Bankruptcy Act, 1914.
- a. 2,110. TAYLOR, Frank, now of Grantham Road, Bracebridge Heath, previously of Bran-ston, both in the county of Lincoln. BUTCHER. Court—LINCOLN and HORNCASTLE.
  Date of Filing Petition—July 2, 1938.
  No. of Matter—13 of 1938.
  Date of Receiving Order—July 2, 1938.
  No. of Receiving Order—12.
  Whether Debtor's or Creditor's Petition— Debtor's. No.
- No. 2,111. BUSHELL, Ernest, 58, Liverpool Road, Birkdale, lately residing at 19, Salford Road, Ainsdale, both in Southport, in the county of Lancaster. Of no occupation.
  Court—LIVERPOOL.
  Date of Filing Petition—July 1, 1938.
  No. of Matter—30 of 1938.
  Date of Receiving Order—July 1, 1938.
  No. of Receiving Order—27.
  Whether Debtor's or Creditor's Petition— Debtor's.

  - Debtor's.
- No. 2,112. WORMALD, Margaret (Widow), resid-ing and carrying on business at 165-7, Kensing-ton, Liverpool, in the county of Lancaster. TOBACCONIST, NEWSAGENT and CONFEC-TIONER. Court-LIVERPOOL
  - Date of Filing Petition-June 3, 1938.

  - No. of Matter—23 of 1938. Date of Receiving Order—June 30, 1938. No. of Receiving Order—26. Whether Debtor's or Creditor's Pe

  - Creditor's Petition-
  - Creditor's. Section I-I (A.), Bankruptcy Act, 1914.
- No. 2,113. LANGLEY, Edith (Widow), 11, Pang-bourne Street, Reading, in the county of Berks, formerly carrying on business at 754, Oxford Road, Reading, in the said county. COAL MERCHANT and GENERAL STORE KEEPER.
  - Court-READING

. .

- Date of Filing Petition-May 31, 1938.

- No. of Matter—5 of 1938. Date of Receiving Order—June 30, 1938. No. of Receiving Order—4. Whether Debtor's or Creditor's Petition-Creditor's.
- Act of Bankruptcy proved in Creditor's Petition— Section 1-1 (E.), Bankruptcy Acts, 1914 and 1926.
- 0. 2,114. SPENCELEY, Arthur, residing and carrying on business at 2, Franklin Road, Gillingham, in the county of Kent. GROCER and PROVISION MERCHANT. Court—ROCHESTER. No. 2,114.

  - Court—ROCHESTER. Date of Filing Petition—June 30, 1938. No. of Matter—16 of 1938. Date of Receiving Order—June 30, 1938. No. of Receiving Order—16. Whether Debtor's or Creditor's Pe Petition-Debtor's.

- 2,115. CLAPPISON, Henry, residing and carrying on business at Inn Farm, Settrington, near Malton, in the county of York. FARMER. No. 2,115.
- near Maiton, in the county of York. FA Court-SCARBOROUGH. Date of Filing Petition-June 30, 1938. No. of Matter-16 of 1938. Date of Receiving Order-June 30, 1938. No. of Receiving Order-15. Whether Debtor's or Creditor's Pe Debtor's. Petition-Debtor's.
- o. 2,116. MARSHALL, Albert Edward, 85, Clarkehouse Road, in the city of Sheffield, formerly residing at 2, Summerfield, and carry-ing on business at 18, Valley Road, 510, London Road, and 355, Glossop Road, all in Sheffield aforesaid, under the name or style of MARSHALL'S. PLASTERING CONTRACTOR and CABINET MAKERS SUPPLIER. Court-SHEFFIELD. Date of Filing Petition-June 30, 1038. No.

  - Date of Filing Petition—June 30, 1938. No. of Matter—46 of 1938. Date of Receiving Order—June 30, 1938. No. of Receiving Order—43. Whether Debtor's or Creditor's Petition-
- Petition-Debtor's.

- 2,117. CHILTON, Ernest, residing and carrying on business at New Buildings Farm, Gayton, Stafford, in the county of Stafford. FARMER No. FARMER
- Court—STAFFORD. Date of Filing Petition—July 1, 1938. No. of Matter—3 of 1938. Date of Receiving Order—July 1, 1938. No. of Receiving Order—3. Whether Debtor's or Creditor's Petition— Debtor's.
- Jo. 2,118. CARTER, William Philip Penquite, Blisland, Corhwall, and formerly of Grade Farm, The Lizard, Cornwall. Trapper, formerly FARMER.
  Court—TRURO and FALMOUTH.
  Date of Filing Petition—July 1, 1938.
  No. of Matter—15 of 1938.
  Date of Receiving Order—July 1, 1938.
  No. of Receiving Order—15.
  Whether Debtor's or Creditor's Petition— Debtor's. No. 2,118.

- No. 2,119. WILLIAMS, Leslie Lewis, residing at 316, Newhampton Road West, Wolverhampton, in the county of Stafford (no occupation), lately carrying on business at the same address, as a FRUITERER. Court-WOLVERHAMPTON.

  - Court—WOLVERHAMPTON. Date of Filing Petition—May 31, 1938. No. of Matter—10 of 1938. Date of Receiving Order—June 29, 1938. No. of Receiving Order—9. Whether Debtor's or Creditor's Petition-Creditor's.
  - Act of Bankruptcy proved in Creditor's Petition-Section 1-1 (G.), Bankruptcy Act, 1914.

#### FIRST MEETING.

- DAVIS, George Albert, deceased, late 29A, Friars Road, Coventry, in the county of Warwick, and lately carrying on business at 32, Hertford Street, Coventry, under the name or style of JACK DAVIS. GENTLEMEN'S OUT-JACK FITTER

  - Court—COVENTRY. No. of Matter—6 of 1938. Date of First Meeting—July 13, 1938. 3 p.m. Place—Official Receiver's Office, 9-11, High Street, Coventry.

#### FIRST MEETINGS AND PUBLIC EXAMINATIONS.

DZEN, Jack Edward, 11, Nevilles Court, London, N.W.2, and lately of 30, Crespigny Road, London, N.W.4. INSURANCE BROZEN. ASSESSOR.

ASSESSOR. Court—HIGH COURT OF JUSTICE. No. of Matter—536 of 1938. Date of First Meeting—July 14, 1938. 11.30 a.m. Place—Bankruptcy Buildings, Carey Street Place—Bankruptcy Buildings, Carey Street, London, W.C.2. Date of Public Examination—Oct. 4, 1938.

11 a.m.

- Place—Bankruptcy London, W.C.2. Buildings, Carey Street,
- COHEN, L. (Male), carrying on business at 422, Liverpool Road, Holloway, London. SCRAP METAL MERCHANT.

- METAL MERCHANI. Court—HIGH COURT OF JUSTICE. No. of Matter—448 of 1938. Date of First Meeting—July 15, 1938. 11.30 a.m. Place—Bankruptcy Buildings, Carey Street Place-Bankruptcy Buildings, Carey S London, W.C.2. Date of Public Examination-Oct. 4, Street,
- 1938. II a.m. Place-Bankruptcy London, W.C.2. Buildings, Carev Street.
- FITZGERALD, Desmond William, carrying on business at Goswell Works, Stratford, Essex, and residing at Stuart Private Hotel, Knole Road, residing at Stuart Private Hotel, Knole Road, Bexhill-on-Sea, Sussex, and lately residing at 6, Marina Court Avenue, Bexhill-on-Sea afore-said. COMMERCIAL AGENT. Court—HIGH COURT OF JUSTICE. No. of Matter—463 of 1938. Date of First Meeting—July 15, 1938. 11 a.m. Place—Bankruptcy Buildings, Carey Street, London, W.C.2.

  - Place—Bankruptcy Buildings, Carey Street, London, W.C.2.
     Date of Public Examination—Oct. 4, 1938.
  - II a.m.
  - Place-Bankruptcy London, W.C.2. Carey Buildings, Street.
- H. & S. BUILDING CO. (a firm), carrying on business at 47B, Redcliffe Square, Earl's Court, London, W. BUILDERS.
  Court—HIGH COURT OF JUSTICE.
  No. of Matter—428 of 1938.
  Date of First Meeting—July 13, 1938. 11 a.m.
  Place—Bankruptcy Buildings, Carey Street, London. W.C.2.

  - Place—Bankruptcy Buildings, Carey Street, London, W.C.2. Date of Public Examination—Sept. 20, 1938.

  - II a.m. Place—Bankruptcy London, W.C.2. Buildings, Carey Street,
- ASK, A. E. (Male), lately carrying on business at and residing at 172, Warren Drive, Chels-field, Kent, present address unknown to Petitioning Creditors. DRAPER. Court—HIGH COURT OF JUSTICE. No. of Matter—453 of 1938. Date of First Meeting—July 15, 1938. 11 a.m. Place—Bankruptcy Buildings, Carey Street, London W C.2. PASK, A.

- Place—Bankruptcy Buildings, Carey Street, London, W.C.2. Date of Public Examination—Sept. 27, 1938.
- II a.m. Buildings,
- Place—Bankruptcy London, W.C.2. Carey Street,
- SMITH, Coleman, of and residing at 27, Stockleigh Hall, Northgate, London, N.W.8. COMPANY DIRECTOR.

  - Court—HIGH COURT OF JUSTICE. No. of Matter—976 of 1937. Date of First Meeting—July 15, 1938. 11.30 a.m. Place—Bankruptcy Buildings, Carey' Street, Place—Bankruptcy Buildings, Carey Street, London, W.C.2. Date of Public Examination—Sept. 27, 1938.
  - II a.m.
  - Carey Place-Bankruptcy Buildings, Street, London, W.C.2.
    - No. 34528

- TUGWELL, Herbert Frederick Lewen, The Drill Hall, Church Street, Brighton, Sussex, BRIGADE MAJOR, and lately residing at Headquarters, Royal Artillery, The London Division T.A., Finsbury Barracks, London,

  - E.C. Court—HIGH COURT OF JUSTICE. No. of Matter—386 of 1938. Date of First Meeting—July 14, 1938. 12 noon. Place—Bankruptcy Buildings, Carey Street, Place—Bankruptcy Buildings, Carey Street, London, W.C.2. Date of Public Examination—Oct. 18, 1938.
  - II a.m.
  - Place—Bankruptcy London, W.C.2. Buildings, Carev Street.
- WINICK, Isaac, residing and carrying on business at 68-70, Christian Street, London, E.I. WHOLESALE FURNITURE MANUFAC-TURER.

  - Court—HIGH COURT OF JUSTICE. No. of Matter—435 of 1938. Date of First Meeting—July 15, 1938. 12 noon. Place—Bankruptcy Buildings, Carey Street, Place—Bankruptcy Buildings, Carey Street, London, W.C.2. Date of Public Examination—Sept. 27, 1938.

  - II a.m. Place—Bankruptcy London, W.C.2. Buildings, Carey Street,
- WOOD, Thomas Rigby, trading as T. R. WOOD & CO., ESTATE AGENT, at 41-42, Chandos House, Buckingham Gate, London, and residing at 27, Devon Road, Cheam, Surrey.
  Court—HIGH COURT OF JUSTICE.
  No. of Matter—415 of 1938.
  Date of First Meeting—July 14, 1938. 11.30 a.m.
  Place—Bankruptcy Buildings, Carey Street, London, W.C.2.

  - Place—Bankruptcy Buildings, Carey Street, London, W.C.2. Date of Public Examination—Oct. 18, 1938.
  - 11 a.m.
  - Place—Bankruptcy London, W.C.2. Buildings, Carey Street,
- PEACE, Clifford, residing at 4, Ferndene, Bingley, in the county of York, lately residing at 6, Trinity Place, Bingley aforesaid. TEXTILE OPERATIVE.

  - OPERATIVE. Court—BRADFORD. No. of Matter—34 of 1938. Date of First Meeting—July 12, 1938. 10.30 a.m. Place—The Official Receiver's Office, Hallfield Chambers, 71, Manningham Lane, Bradford. Date of Public Examination—Aug. 10, 1938.
  - 10.45 a.m. Place—The County Court, Manor Row, Bradford.
- VARLEY, George William, residing at 72, South Parade, Pudsey, in the county of York, and carrying on business at 160s, Swan Arcade, in the city of Bradford. WOOL MERCHANT. Court-BRADFORD.
  No. of Matter-30 of 1938.
  Date of First Meeting-July 12 1938. 11 a.m.
  Place-The Official Receiver's Office, Hallfield Chambers, 71, Manningham Lane, Bradford.
  Date of Public Examination-Oct. 4, 1938.

  - lace—The County Court, Manor Row, Bradford.
  - Place-
- CURTIS, Sidney, of "Broadways," Lake Road, Henleaze, in the city and county of Bristol. BUILDER.

  - BUILDER. Court—BRISTOL. No. of Matter—24 of 1938. Date of First Meeting—July 19, 1938. 12 noon. Place—Official Receiver's Office, 26, Baldwin Street, Bristol. Date of Public Examination—Oct. 28, 1938.
  - 10.30 a.m. lace—The Guildhall, Bristol. Place-

E

.HENSLEY, Percy Walter, 7, Victoria Avenue, Redfield, in the city and county of Bristol, and trading as F. HENSLEY & SON, at The Midland Saw Mills, Lawrence Hill, Bristol aforesaid. UPHOLSTERER and CHAIR FRAME MAKER. Court—BRISTOL. No. of Matter—27 of 1938. Date of First Meeting—July 19, 1938. 11.30 a.m. Place—Official Receiver's Office, 26, Baldwin Street. Bristol.

- Street, Bristol. Date of Public Examination—Sept. 30, 1938.
- 10.30 a.m.

Place-The Guildhall, Bristol.

- PRIOR, Philip William, 102, Sunnyfields Estate, Histon, in the county of Cambridge, Plasterer, lately residing and carrying on business at Green End, Fen Ditton, in the county of Cambridge, as a PLASTERING CON-TRACTOR Cambridge, as TRACTOR. Court-CAMBRIDGE.

  - Court—CAMBRIDGE. No. of Matter—9 of 1938. Date of First Meeting—July 15, 1938. 11.30 a.m. Place—The Official Receiver's Office, 41, Sidney Street, Cambridge. Date of Public Examination—July 27, 1938.
  - II a.m.
  - Place-The Guildhall, Cambridge.
- MALLINSON, John Gibson, residing at 2, London Road Terrace, in the city of Carlisle, and carrying on business at 42, South Street, in the said city of Carlisle, trading as "JAMES HEWITT & SON." JOINER and UNDER-TATER TAKER.

  - Court—CARLISLE. No. of Matter—5 of 1938. Date of First Meeting—July 13, 1938. 11.30 a.m. Place—Official Receiver's Office, 18, Bank Street, Carlisle. Date of Public Examination—July 25, 1938.

  - 2.30 p.m. Place—The Court House, English Street, Carlisle. Date of Order for Summary Administration— July 1, 1938.
- CARR, Molly Violet, the Wife of Leonard Thomas Carr, of Nantymwyn House, Rhandirmwyn, near Llandovery, in the county of Carmarthen.
  Court—CARMARTHEN.
  No. of Matter—6 of 1938.
  Date of First Meeting—July 12, 1938. 2.30 p.m.
  Place—4, Queen Street, Carmarthen.
  Date of Public Examination—Aug. 9, 1938.

  - 12 noon.

  - Place—The Guildhall, Carmarthen. Date of Order for Summary Administration— June 30, 1938.
- MINION, John Edward, 58, Church Street, in the county borough of Derby. CYCLE SHOP PROPRIETOR.
  - Court-DERBY and LONG EATON.

  - No. of Matter—18 of 1938. Date of First Meeting—July 12, 1938. 11 a.m. Place—The Official Receiver's Office, 22, Regent Street, Park Row, Nottingham. Date of Public Examination—July 19, 1938.

  - ro a.m. Place—The Court House, 20, St. Peter's Church-yard, Derby.
- GILES, Lewis Albert Clifford, residing and carrying on business at 18, Owen Street, Tipton, in the county of Stafford. BUTCHER and PROVI-SION MERCHANT.
  Court-DUDLEY.
  No. of Matter-5 of 1938.
  Date of First Meeting-July 12, 1938. 11.30 a.m.
  Place-Official Receiver's Offices, 1, Priory Street, Dudley.

  - Dudley.
  - Date of Public Examination-July 21, 1938. II a.m.
  - Place—The Court House, Priory Street, Dudley. Date of Order for Summary Administration— July 1, 1938.

- SMITH, William Norman, residing at 53, Oakham Road, Dudley, in the county of Worcester, and carrying on business at 18, King Street, Dudley aforesaid. WIRELESS DEALER. Court-DUDLEY.

  - No. of Matter—6 of 1938. Date of First Meeting—July 14, 1938. 12 noon. Place—Official Receiver's Offices, 1, Priory Street, Dudley.
  - Date of Public Examination-Aug. 4, 1938.

II a.m. Place—The Court House, Priory Street, Dudley. Date of Order for Summary Administration-July 1, 1938.

- BOLT, Charles Gilbert, 5, Gold Street, Ottery St. Mary, Devon. BUTCHER.
  Court—EXETER.
  No. of Matter—7 of 1938.
  Date of First Meeting—July 12, 1938. 11.30 a.m.
  Place—The Official Receiver's Offices, Exeter Bank Chambers, Broadgate, Exeter.
  Date of Public Examination—July 27, 1938.

  - 10.30 a.m. Place—The Castle, Exeter. Date of Order for Summary Administration— July 1, 1938.
- PREECE, Thomas William, lately carrying on business at and residing at Pencoyd Court, Pencoyd, in the county of Hereford. FARMER.
  Court—HEREFORD.
  No. of Matter—6 of 1938.
  Date of First Meeting—July 12, 1938. 12 noon.
  Place—Official Receiver's Office, 133, Saint Owen Stratt Hereford

  - Street, Hereford. Date of Public Examination—July 18, 1938. 2.30 p.m. Place—The Shirehall, Hereford.
- COULING, Norman, 43, Cannon Street, in the city of Lincoln. GENERAL DEALER. Court—LINCOLN and HORNCASTLE. No. of Matter—12 of 1938. Date of First Meeting—July 13, 1938. 12 noon. Place—Official Receiver's Office, 1, St. Swithin's Scaure Lincoln

  - Square, Lincoln.
  - Date of Public Examination-Aug. 4, 1938.

  - 3 p.m. Place—The Sessions House, Lincoln. Date of Order for Summary Administration— July 1, 1938.
- WORMALD, Margaret (Widow), residing and carry-ing on business at 165-7, Kensington, Liverpool, in the county of Lancaster. TOBACCONIST, NEWSAGENT and CONFECTIONER.

  - NEWSAGENT and CONFECTIONER. Court-LIVERPOOL. No. of Matter-23 of 1938. Date of First Meeting-July 13, 1938. 11.30 a.m. Place-At the offices of the Official Receiver, Government Buildings, Victoria Street, Liverpool. Date of Public Examination-Aug. 16, 1938.
  - Place—At the Court House, Government Buildings, Victoria Street, Liverpool. Date of Order for Summary Administration— July 2, 1938.
- FORD, John, and FORD, Frederick, both residing at 5, Mount Road, Alkrington, near Middleton, in the county of Lancaster, trading together in co-partnership as J. FORD and SON, at 35, Shudehill, Manchester, in the said county. PAPER MERCHANTS.
  Court—MANCHESTER.
  No. of Matter—36 of 1938.
  Date of First Meeting—July 14, 1938. 11 a.m.
  Place—Official Receiver's Offices, 20, Byrom Street, Manchester 3.

  - Street, Manchester 3. Date of Public Examination,—Aug. 5, 1938. 10 a.m.
  - Place-Court House, Quay Street, Manchester.

GREEN, Harry, residing at 68, Northumberland Street, Higher Broughton, Salford, and carrying on business in co-partnership with another at 54, Moreton Street, Strangeways, Manchester, both in the county of Lancaster. CAP MANU-FACTURER

- FACTORER.
  Court—MANCHESTER.
  No. of Matter—45 of 1938.
  Date of First Meeting—July 13, 1938. 11.30 a.m.
  Place—Official Receiver's Offices, 20, Byrom Street, Manchester 3.
  Date of Public Examination—Aug. 5, 1938.
- 10 a.m.

Place-Court House, Quay Street, Manchester.

- KELSALL, Thomas, residing and carrying on busi-ness at 35, York Road, Bowdon, in the county of Chester. GROCER and CONFECTIONER. Court--MANCHESTER.

  - Court—MANCHESIER. No. of Matter—48 of 1938. Date of First Meeting—July 13, 1938. 11 a.m. Place—Official Receiver's Offices, 20, Byrom Street, Manchester 3. Date of Public Examination—Aug. 5, 1938.
  - 10 a.m.
  - Place-Court House, Quay Street, Manchester.
- SUTTON, John Hewitt, residing at 97, Russell Street, Moss Side, Manchester, in the county of Lancaster, and carrying on business at 235, Moss Lane East, Moss Side, Manchester afore-coid CROCER said. GROCER.

Court-MANCHESTER.

- Date of First Meeting—July 12, 1938. 11 a.m. Place—Official Receiver's Office, 20, Byrom Street, Manchester.
- Date of Public Examination-July 29, 1938. 10 a.m.
- Place-Court House, Quay Street, Manchester.
- WOODSIDE, Charles Victor, residing at 60, Heald Grove, Rusholme, Manchester, in the county of Lancaster, and carrying on business at 94, Mosley Street, Manchester aforesaid. Mosley St. STATIONER.

- Court—MANCHESTER. No. of Matter—43 of 1938. Date of First Meeting—July 12, 1938. 11.30 a.m. Place—Official Receiver's Office, 20, Byrom Street, Manchester.
- Date of Public Examination—Aug. 5, 1938. 10 a.m.
- Place-Court House, Quay Street, Manchester.
- IOLTAWAY, Harriet (Married Woman), and HOLTAWAY, Edith Elizabeth Frances (Spinster), trading in co-partnership as H. & E. HOLTAWAY, at 15 and 17, St. Augustines Street, Norwich. FURNISHERS. Court—NORWICH. No. of Matter—22 of 1938. Date of First Meeting—July 14, 1938. 11 a.m. Place—The Official Receiver's Office, Castle Chambers, Opie Street, Norwich. Date of Public Examination—Sept. 20, 1938. 2.30 p.m. HOLTAWAY,

  - 2.30 p.m. Place—The Shirehall, Norwich.
- KIRK, George Henry, The Swan Inn, Gressenhall, Norfolk. LICENSED VICTUALLER.
  Court-NORWICH.
  No. of Matter-23 of 1938.
  Date of First Meeting-July 12, 1938. II a.m.
  Place-The Official Receiver's Office, Castle Chambers, Opie Street, Norwich.
  Date of Public Examination-July 19, 1938.
  2 30 p.m.

  - 2.30 p.m. Place—The Shirehall, Norwich. Date of Order for Summary Administration—

  - July 1, 1938.

- SPENCELEY, Arthur, residing at and carrying on business at 2, Franklin Road, Gillingham, in the county of Kent. GROCER and PRO-VISION MERCHANT.

  - VISION MERCHANI. Court-ROCHESTER. No. of Matter-16 of 1938. Date of First Meeting-July 15, 1938. 11.15 a.m. -Place-Official Receiver's Offices, 280A, High Street, Rochester, Kent. Date of Public Examination-Aug. 8, 1938.

  - 11.15 a.m. Place—The Court House, Eastgate, Rochester. Date of Order for Summary Administration— July 2, 1938.
- MARRIOTT, Arthur James, "Alpha," The Rise, Park Street, St. Albans, in the county of Hert-ford, lately residing and carrying on business at "The Gables," Watford Road, St. Stephens, St. Albans aforesaid. BUILDER. Court—ST. ALBANS. No. of Matter—14 of 1938.
  Date of First Meeting—July 12, 1938. 12 noon. Place—29, Russell Square, London, W.C.I. Date of Public Examination—Sept. 21, 1938. 10.30 a.m.
  - 10.30 a.m. Place—The Court House, St. Albans.
- SEARSTON, Ada (Married Woman), residing and carrying on business at 50, Norton Lees Road, in the city of Sheffield. GROCER. Court-SHEFFIELD.

  - No. of Matter-45 of 1938. Date of First Meeting-July 13, 1938. 11.30 a.m. Place-Official Receiver's Office, 55, Queen Street,
  - Sheffield I. Date of Public Examination—July 28, 1938. 2 p.m.
  - Place—County Sheffield 1. Hall, Court Bank Street.
  - Date of Order for Summary Administration-June 29, 1938.
- PERKS, Frank, 7, Belle Vue, Wordsley, in the county of Stafford, GLASS CUTTER, and formerly carrying on business in partnership with William Charles Griffin, under the name of The Alto Transport, at 7, Belle Vue, Words-

  - ley aforesaid. Court—STOURBRIDGE. No. of Matter—3 of 1938. Date of First Meeting—July 12, 1938. 12 noon. Place—Official Receiver's Offices, I, Priory Priory Street, Dudley. Date of Public Examination—Aug. 10, 1938.
  - 11 a.m.
  - Place--The Court House, Hagley Road, Stourbridge.
  - Date of Order for Summary Administration-July 1, 1938.
- PARRY, Wilfred Geoffrey, 15, Church Street, bury, in the county of Gloucester. BF bury, in LAYER BRICK-
  - Court-SWINDON.

  - No. of Matter—6 of 1938. Date of First Meeting—July 20, 1938. 11.30 a.m. Place—County Court Buildings, Clarence Street, Swindon.
  - Date of Public Examination-Sept. 21, 1938.
  - 2.15 p.m. ace—County Court Buildings, Clarence Street, Place Swindon.
  - Date of Order for Summary Administration-June 30, 1938.
- /ILLETT, Charles David Ballantyne, trading as CHARLES WILLETT, The Nursery, Yeading Lane, Hayes, in the county of Middlesex. NURSERYMAN.
  Court—WINDSOR.
  No. of Matter—18 of 1938.
  Date of First Meeting—July 13, 1938. 3 p.m.
  Place—The Official Receiver's Office, 8, Forbury, Reading WILLETT.

  - Reading. Date of Public Examination—July 29, 1938.
  - II a.m. Place-The Town Hall, Windsor.

E 2

<ul> <li>the county of Hereford, Farm Labourer, formerly a FARMER.</li> <li>Court—WORCESTER.</li> <li>No. of Matter—9 of 1938.</li> <li>Date of First Meeting—July 12, 1938. 3 p.m.</li> <li>Place—The Official Receiver's Office, 3, Sansome Place, Worcester.</li> <li>Date of Public Examination—July 26, 1938.</li> <li>-2 p.m.</li> </ul>	<ul> <li>MINION, John Edward, 58, Church Street, in the county borough of Derby. CYCLE SHOP PROPRIETOR.</li> <li>Court—DERBY and LONG EATON.</li> <li>No. of Matter—18 of 1938.</li> <li>Date of Order—June 30, 1938.</li> <li>Date of Filing Petition—June 30, 1938.</li> </ul>
Place—The Guildhall, Worceșter.	COATES, Henry, residing and carrying on business at 1, Copley, Butterknowle, S.O., in the county of Durham. COAL MERCHANT. Court-DURHAM. No. of Matter-4 of 1938. Date of Order-July 1, 1938. Date of Filing Petition-July 1, 1938.
ADJUDICATIONS.	and the second
<ul> <li>BROZEN, Jack Edward, 11, Nevilles Court, London, N.W.2, and lately of 30, Crespigny Road, London, N.W.4., INSURANCE ASSESSOR.</li> <li>Court—HIGH COURT OF JUSTICE.</li> <li>No. of Matter—536 of 1938.</li> <li>Date of Order—July 1, 1938.</li> <li>Date of Filing Petition—July 1, 1938.</li> </ul>	<ul> <li>ANDERSON, Mervyn Kendall, The Mill Bakeries, Trusthorpe, Mablethorpe, Lincolnshire, BAKER, and formerly of Seabank House, Gibraltar Road, Mablethorpe.</li> <li>Court—GREAT GRIMSBY.</li> <li>No. of Matter—19 of 1938.</li> <li>Date of Order—July 1, 1938.</li> <li>Date of Filing Petition—July 1, 1938.</li> </ul>
	TADDIGON Langed Travellow Boot Inp.
BROADBENT, Mary Alice (Married Woman), resid- ing at 71, Ladbrooke Road, Smallshaw, Ashton- under-Lyne, in the county of Lancaster, and formerly residing and carrying on business at 1, Silas Street, Hurst, Ashton-under-Lyne afore- said, as a GROCER and MIXED GOODS DEALER.	HARRISON, Leonard, Travellers Rest Inn, Sowerby, Sowerby Bridge, in the county of York. LICENSED VICTUALLER and FARM LABOURER. Court—HALIFAX. No. of Matter—13 of 1938. Date of Order—July 1, 1938. Date of Filing Petition—July 1, 1938.
Court-ASHTON-UNDER-LYNE and STALY-	Date of Fining Foldations July 1, 1930.
BRIDGE. No. of Matter—5 of 1938. Date of Order—July 1, 1938. Date of Filing Petition—July 1, 1938.	BROWN, Eric Frederick, now residing at 20, Cheltenham Mount, Harrogate, in the county of York, and carrying on business at 15, Cambridge Street, Harrogate aforesaid, as a FRUITERER and FLORIST, under the style of "BROWN
PEACE, Clifford, residing at 4, Ferndene, Bingley, in the county of York, lately residing at 6, Trinity Place, Bingley aforesaid. TEXTILE OPERATIVE.	AND BEST." Court—HARROGATE. No. of Matter—9 of 1938. Date of Order—June 30, 1938. Date of Filing Petition—June 30, 1938.
Court-BRADFORD.	Date of Fining Fertition—June 30, 1930.
No. of Matter—34 of 1938. Date of Order—June 30, 1938.	
Date of Filing Petition—June 30, 1938.	CURTIS, Cecil, lately residing at The Imperial Hotel, Widemarsh Street, in the city of Here- ford, and now residing in apartments at "St. Hyldas," Aylestone Hill, in the said city. COMMERCIAL TRAVELLER.
<ul> <li>WATKINS, Lydia, residing in rooms at 15, Second Avenue, Selly Park, Birmingham, in the county of Warwick, and carrying on business at Guild- hall Buildings, 11, Navigation Street, and 37, Stephenson Street, Birmingham, in the county of Warwick, as a FRUITERER and CONFEC- TIONER.</li> </ul>	Court—HEREFORD. No. of Matter—7 of 1938. Date of Order—July 2, 1938. Date of Filing Petition—July 2, 1938.
Court—BIRMINGHAM. No. of Matter—32 of 1938. Date of Order—June 30, 1938. Date of Filing Petition—June 30, 1938.	<ul> <li>TAYLOR, Frank, now of Grantham Road, Brace- bridge Heath, previously of Branston, both in the county of Lincoln. BUTCHER.</li> <li>Court—LINCOLN and HORNCASTLE.</li> <li>No. of Matter—13 of 1938.</li> <li>Date of Order—July 2, 1938.</li> <li>Date of Filing Petition—July 2, 1938.</li> </ul>
CURTIS, Sidney, "Broadways," Lake Road,	
Henleaze, in the city and county of Bristol. BUILDER. Court—BRISTOL. No. of Matter—24 of 1938. Date of Order—July 1, 1938. Date of Filing Petition—May 31, 1938.	BUSHELL, Ernest, 58, Liverpool Road, Birkdale, lately residing at 19, Salford Road, Ainsdale, both in Southport, in the county of Lancaster. Of no occupation. Court—LIVERPOOL. No. of Matter—30 of 1938.
	Date of Order-July 1, 1938. Date of Filing Petition-July 1, 1938.

- WELLINGS, Dennis, residing at 11, Hickson Street, Barnton, near Northwich, in the county of Chester, formerly 18, Moss Road, Winnington, Northwich, and 37, Middlewich Road, North-wich, in the county of Chester. CHEMICAL LABOURER.
  Court—CREWE, NANTWICH and SANDBACH.
  No. of Matter—6 of 1938.
  Date of Order—June 30, 1938.
  Date of Filing Pétition—June 30, 1938.

. .

- WORMALD, Margaret (Widow), residing and carrying on business at 165-7, Kensington, Liverpool, in the county of Lancaster. TOBACCONIST, NEWSAGENT and CONFECTIONER. Court-LIVERPOOL.
  No. of Matter-23 of 1938.
  Date of Order-June 30, 1938.
  Date of Filing Petition-June 3, 1938.

- GLADDISH, Alfred James, in the Receiving Order described as A. Gladdish (Male), 16, Calder Road, Ringlestone Estate, Maidstone, in the county of Kent, Butcher, formerly residing and carrying on business at Lambert House, East Malling, in the county of Kent, as a SAND and BALLAST MERCHANT, under the style or firm of LUNSFORD SAND and GRAVEL CO. čo.
  - . Court--MAIDSTONE.

  - Date of Order—July 1, 1938. Date of Filing Petition—May 31, 1938.
- FORD, John, and FORD, Frederick, both residing at 5, Mount Road, Alkrington, near Middleton, in the county of Lancaster, trading together in co-partnership as J. FORD & SON, at 35, Shudehill, Manchester, in the said county. PAPER MERCHANTS. Court-MANCHESTER. No. of Matter-26 of 1008
  - No. of Matter—36 of 1938. Date of Order—July 1, 1938. Date of Filing Petition—May 24, 1938.
- SPENCELEY, Arthur, residing and carrying on busi-ness at 2, Franklin Road, Gillingham, in the county of Kent. GROCER and PROVISION MERCHANT. No. of Matter—16 of 1938. Date of Order—June 30, 1938. Date of Filing Petition—June 30, 1938.
- CLAPPISON, Henry, residing and carrying on business at Inn Farm, Settrington, near Malton, in the county of York. FARMER.
  Court—SCARBOROUGH.
  No. of Matter—16 of 1938.
  Date of Order—June 30, 1938.
  Date of Filing Petition—June 30, 1938.
- MARSHALL, Albert Edward, 85, Clarkehouse Road, in the city of Sheffield, formerly residing at 2, Summerfield, and carrying on business at 18, Valley Road, 510, London Road, and 355, Glossop Road, all in Sheffield aforesaid, under the name or style of MARSHALL'S. PLASTER-ING CONTRACTOR and CABINET MAKERS' SUPPLIER.
  Court—SHEFFIELD.
  No. of Matter—46 of 1938.
  Date of Order—June 30, 1938.
  Date of Filing Petition—June 30, 1938.

- CHILTON, Ernest, residing and carrying on busi-ness at New Buildings Farm, Gayton, Stafford, in the county of Stafford. FARMER. Court-STAFFORD. No. of Matter—3 of 1938. Date of Order—July 1, 1938. Date of Filing Petition—July 1, 1938.
- CARTER, William Philip, Penquite, Blisland, Cornwall, and formerly of Grade Farm, The Lizard, Cornwall. Trapper, formerly FARMER.
  Court—TRURO and FALMOUTH.
  No. of Matter—15 of 1938.
  Date of Order—July 1, 1938.
  Date of Filing Petition—July 1. 1938.
- GARNESS, William, The Oaks Farm, Bringsty, in the county of Hereford. Farm Labourer, formerly a FARMER. Court-WORCESTER.

  - No. of Matter—9 of 1938. Date of Order—July 2, 1938. Date of Filing Petition—May 24, 1938.

# APPLICATIONS FOR DISCHARGE.

- BARLOW, Ashworth, Flat 70, Witley Court, Coram Street, London, W.C., and lately residing at 66, Highview Gardens, Edgware, London, W., and carrying on business at 2, Duncan Avenue, London, W.C.I, under the style of "PARAGON ENGRAVING COMPANY." ENGRAVER. Court-HIGH COURT OF JUSTICE.

  - Court—HIGH COOKI OF Journal No. of Matter—522 of 1937. Day Fixed for Hearing—July 27, 1938. 11 a.m. Place—Bankruptcy Buildings, Carey Street, London, W.C.2.
- DA COSTA, Alfred Gomez, commonly known and described in the Receiving Order as Alfred da Costa, residing at 177 and formerly at 159, Maida Vale, London, W.9, lately carrying on business as DA COSTA & CO., at 32, Lee High Road, London, S.E.13. SALESMAN and DIS-TRIBUTOR of WASHING MACHINES.
  Court--HIGH COURT OF JUSTICE.
  No. of Matter-308 of 1938.
  Day Fixed for Hearing-July 27, 1938. 11 a.m. Place-Bankruptcy Buildings, Carey Street, London, W.C.2.
- FARRANT, Robert Donovan, 196, Archway Road, Highgate, London, and lately carrying on busi-ness at rear of 543, Holloway Road, London. RADIO DEALER.
  Court—HIGH COURT OF JUSTICE.
  No. of Matter—55 of 1938.
  Day Fixed for Hearing—July 27, 1938. 11 a.m.
  Place—Bankruptcy Buildings, Carey Street, London, W.C.2.
- JACOBS, Morris, 49, Craven Walk, Stamford Hill, N.16, lately carrying on business under the style of "GENERAL TRANSPORT SER-VICE" and "GENERAL TRANSPORT SUPPLY," at 32, Brushfield Street, Spitalfields, E.I. MOTOR HAULAGE CONTRACTOR.
  Court-HIGH COURT OF JUSTICE.
  No. of Matter-262 of 1931.
  Day Fixed for Hearing-July 28, 1938. 11 a.m.
  Place-Bankruptcy Buildings, Carey Street, London, W.C.2.

  - Place—Bankruptcy London, W.C.2.
- JOHNSON, John Paley, residing at 13, Rosary Gardens, London, S.W.7, described in the Re-ceiving Order as lately carrying on business at 9, Basinghall Street, London, E.C., and residing at Fieldhead, Sunningdale, Berks, a domiciled Englishman, domiciled in England. DEALER in STOCKS and SHARES.
  Court-HIGH COURT OF JUSTICE. No. of Matter—803 of 1937.
  Day Fixed for Hearing—July 28, 1938. 11 a.m. Place—Bankruptcy Buildings, Carey Street, London, W.C.2.
- KNIGHT, Henry Alexander, 46, Malvern Road, Hackney, London, E.8, lately carrying on busi-ness as H. H. HOOPER & CO., at 32, Fortescue Avenue, Hackney, London, E.8. SHOE MANU-FACTURER.

  - Court—HIGH COURT OF JUSTICE. No. of Matter—78 of 1938. Day Fixed for Hearing—July 28, 1938. 11 a.m. Place—Bankruptcy Buildings, Carey Street, Place—Bankruptcy London, W.C.2.
- MACKAY, Violet Maud, 24, D'Oyley Street, Sloane Square, London, S.W.I, Widow (described in the Receiving Order as Violet Mackay). Occupation unknown.
- patton unknown. Court—HIGH COURT OF JUSTICE. No. of Matter—201 of 1937. Day Fixed for Hearing—July 29, 1938. 11 a.m. Place—Bankruptcy Buildings, Carey Street, London, W.C.2.

MELLOR, Ernest Rogers, 8, Queen Street, London, E.C., SOLICITOR, and lately carrying on busi-ness at 8, Great Winchester Street, London, E.C.

-HIGH COURT OF JUSTICE. Court-

- Day Fixed for Hearing—July 29, 1938. 11 a.m. Place—Bankruptcy Buildings, Carey Street Place—Bankruptcy London, W.C.2. Street,
- TER, Florence Bessie (Spinster) (trading as CARTERS), 319, London Road, Mitcham, in the county of Surrey. ART NEEDLEWORK CARTER,

- DEALER. Court—CROYDON. No. of Matter—37 of 1935. Day Fixed for Hearing—July 27, 1938. 10.30 a.m. Place—The County Court, Scarbrook Road, Croydon.
- ARLEY, Sidney, trading as. BETCHWORTH SPORTS CLUB, residing and carrying on busi-ness at Brockham Lane, Betchworth, Surrey. SPORTS CLUB PROPRIETOR. Court—CROYDON. No. of Matter—46 of 1937. Day Fixed for Hearing—July 27, 1938. 10.30 a.m. Place—The County Court, Scarbrook Road, Croydon. FARLEY

  - Croydon.
- RICHARDS, John Martin, The Manor Inn, Bishops-teignton, Devon. LICENSED VICTUALLER. Court—EXETER. Court-No. of Matter-46 of 1927. Day Fixed for Hearing-Sept. 16, 1938. 10.30 a.m. lace—The Castle, Exeter. Place
- PARNELL, Thomas, "Weir End," Sunbury-on-Thames, in the county of Middlesex. RETIRED ARMY OFFICER. Court—KINGSTON (Surrey). No. of Matter—23 of 1937 Day Fixed for Hearing—July 26, 1938.
  - 10.30 a.m. Place—The Guildhall, Kingston, Surrey.
- WALLACE, Charles Claude, Feltham House, East Molesey, Surrey, Gentleman, lately residing at Grasmere, Thames Ditton, in the same county. Court—KINGSTON (Surrey).
  No. of Matter—ro of 1898.
  Day Fixed for Hearing—July 26, 1938.
  - 10.30 a.m. lace—The Guildhall, Kingston, Surrey.
  - Place
- CATLING, Thomas, the Elder, and CATLING, Thomas, the Younger, both of Finedon, in the Thomas, the Younger, both or Finedon, in the county of Northampton, carrying on business under the style or firm of Catling and Son. Boot Manufacturers. (Separate application of Thomas Catling, the Younger.) Court—NORTHAMPTON. No. of Matter—20 of 1922. Day Fixed for Hearing—Aug. 5, 1938. 10.45 a.m.

10.45 a.m. lace—County Hall, Northampton. Place-

- MOSS, Thomas William, 20, Brook Street, Williams-town, in the county of Glamorgan, DISPENS-ING CHEMIST, carrying on business at 20, Brook Street, Williamstown aforesaid. Court-PONTYPRIDD, YSTRADYFODWG and PORTH.

28 of 1925 No. of Matter-

- Day Fixed for Hearing—Aug. 10, 1938. 11:30 a.m. Place—The Court House, Courthouse Street,
- Pontypridd.

- WILLIAMS. David John, 5, Lanelay Crescent, Maesycoed, Pontypridd, in the county of Glamorgan, carrying on business at 45, Mill Street, Pontypridd aforesaid, and previously at 18, High Street, Pontypridd aforesaid. OUT-FITTER.
  - Court-PONTYPRIDD, YSTRADYFODWG and PORTH. No. of Matter—17 of 1931. Day Fixed for Hearing—Aug.

  - 1938. 10, 11.30 a.m. Place—The

Court House, Courthouse Street, Pontypridd.

- MORRELL, Percy Mont, 21, Charlotte Street, in the city and county of Kingston-upon-Hull, and lately residing at 30, Bass Terrace, Thorne Road, Doncaster, in the county of York. MOTOR PROPRIETOR.
  Court—SHEFFIELD.
  No. of Matter—33 of 1930.
  Day Fixed for Hearing—July 28, 1938. 2 p.m.
  Place—County Court Hall, Bank Street, Sheffeld t

  - Sheffield 1.

#### ORDERS MADE ON APPLICATIONS FOR DISCHARGE.

- HUGHES, Herbert, "Trelawney," Malletts Row, Barnstaple, Devonshire. LODGINGHOUSE KEEPER.
  - Court-BARNSTAPLE.

  - Court.—BARNSTAPLE.
    No. of Matter—6 of 1935.
    Date of Order—May 31, 1938.
    Nature of Order made—Bankrupt's discharge suspended for seven days, and that he be discharged as from June 8, 1938.
    Grounds named in Order for refusing an absolute Order of Discharge—Proof of Fact mentioned in Section 26, sub-section 3 (B.), Bankruptcy Act, 1914, as amended by Section 1 of the Bankruptcy (Amendment) Act, 1926.
- WRIGHT, Julia Anne (Spinster), Ranelagh House, I, Ranelagh Avenue, Barnes Common, S.W.13, in the county of Surrey. SCHOOL PRO-PRIETRESS.

  - PRIETRESS. Court—WANDSWORTH. No. of Matter—21 of 1937. Date of Order—June 2, 1938. Nature of Order made—That the Bankrupt's dis-charge be suspended for 18 months, and that she be discharged from Dec. 2, 1939. Grounds named in Order for refusing an absolute Order of Discharge -Broofe of Froth monthered
  - Order of Discharge—Proofs of Facts mentioned in Section 26, sub-section 3 (A. and B.), Bank-ruptcy Act, 1914, as amended by Section 1 of the Bankruptcy (Amendment) Act, 1926.

# RELEASE OF TRUSTEES.

- SAACS, Max, 44, Kingsley Court, Willesden Green, Middlesex, BOOT and SHOE RETAILER, and carrying on business at 338, Edgware Road, Paddington, London, and 87, The Broadway, Southall, Middlesex.
  Court—HIGH COURT OF JUSTICE.
  No. of Matter—349 of 1937.
  Trustee's Name, Address and Description— Goldney, Graham Gabriel Prior, 57, Grace-church Street, London, E.C.3, Chartered Accountant. ISAACS

  - Goldney, Grammer, Grammer, Grammer, Goldney, Grammer, G Accountant.
  - Date of Release-June 10, 1938.
- SCHLIFKA, Jacob, otherwise Lipsick (known as Phillips John), of and lately carrying on busi-ness at 319, Whitechapel Road, London, E.I. MANTLE MANUFACTURER.
  Court—HIGH COURT OF JUSTICE.
  No. of Matter—484 of 1934.
  Trustee's Name, Address and Description— Sacks, Harry Benjamin, 62-63, Fenchurch Street, London, E.C.3, Chartered Accountant.
  Date of Release—June 10, 1938.

- SEDDON, Peter Martin, 48, Holborn Viaduct House, 12, Holborn Viaduct, London, E.C. POULTRY IMPORTER.
  Court—HIGH COURT OF JUSTICE.
  No. of Matter—837 of 1935.
  Trustee's Name, Address and Description— Cork, William Henry, 19, Eastcheap, London, E.C.3, Accountant and Auditor.
  Date of Release—June. 8, 1938. E.C.3, Accountant and Audit Date of Release—June 8, 1938.
- SOLOMONS, Fanny, known as Fanny Saunders (Married Woman), practising as COLLETS, who resides and carries on business at 54, Station Road, North Harrow, Middlesex, as AUCTIONEERS and ESTATE AGENTS. Court—HIGH COURT OF JUSTICE.
  No. of Matter—825 of 1934.
  Trustee's Name, Address and Description— Inns, Frederick William, Finsbury Court, Fins-bury Pavement, London, E.C.2, Accountant. Date of Release—May 18, 1938.

- GUNSTONE, Arthur Frederick (deceased), late 26, The Market Place, Westbury, in the county of Wilts. WIRELESS DEALER. Court-BATH. No. of Matter-2 of 1937. Trustee's Name, Address and Description-Wheeler, Harold, 26, Baldwin Street, Bristol, Official Receiver. Date of Release-June 22, 1028
- Date of Release—June 23, 1938.

.

- STOKES, Stanley Charles, of "Glenthorn," Salt-ford, Somerset, and carrying on business at that address and at the Railway Station Yard, Saltford aforesaid. COAL and CORN CORN

۰. • •

- that address and at the Railway Station Yard, Saltford aforesaid. COAL and CORN MERCHANT. Court—BATH. No. of Matter—8 of 1935. Trustee's Name, Address and Description— Taylor, John Godfrey, 28, Baldwin Street, Bristol, Chartered Accountant. Date of Release—June 21, 1938.
- WHITE, Ronald Eric, 233, Bristol Street, Birming-ham. CHEMIST. Court-BIRMINGHAM.

- No. of Matter—4 of 1937. Trustee's Name, Address and Description— Booth, Parkin Stanley, 2, Bixteth Street, Liver-pool, Certified Accountant. Date of Release—June 21, 1938.
- FREELEY, Francis Bernard, residing at 9, Grenfell Drive, Bradford Moor, in the city of Brad-ford and carrying on business at 34, Manchester Road, Bradford aforesaid, under the style of "MIDLAND FACILITIES." MANUFAC-TURER'S AGENT. Court-BRADFORD.

  - Court—Bradrond.
     No. of Matter—41 of 1937. 0
     Trustee's Name, Address and Description— Cresswell, William Foy, Hallfield Chambers, 71, Manningham Lane, Bradford, Official Bradford, Official 71, Ma Receiver.
  - Date of Release-June 23, 1938.
- LARGE, Harry, residing and carrying on business at Hodgson Lane Farm, Drighlington, near the city of Bradford. CATTLE DEALER. Court-BRADFORD.

  - No. of Matter—46 of 1937. Trustee's Name, Address and Description— Cresswell, William Foy, Hallfield Chambers 71, Manningham Lane, Bradford, Officia Bradford, Official Receiver.
  - Date of Release-June 23, 1938.

- PHILLIPS, David, 51, Princes Road, Feltham, Middlesex, and lately carrying on business at R.A.S.C. Depot, Feltham, in the county of Middlesex, as a MASTER TAILOR... Court-BRENTFORD.

  - No. of Matter—33 of 1936. Trustee's Name, Address and Description— Poland, John Langman, 29, Russell Square, London, W.C.r. London, W.C.r. Date of Release—June 22, 1938.
- TAYLOR, Frederick Killick, 4, Pownall Gardens, Hounslow, Middlesex, lately residing and carry-ing on business at "The Hope and Anchor", Upper Sutton Lane, Heston, Middlesex. LICENSED VICTUALLER.

  - LICENSED VICTUALLER. Court—BRENTFORD. No. of Matter—II of 1937. Trustee's Name, Address and Description— Poland, John Langman, 29, Russell Square, London, W.C.I, Official Receiver. Date of Release—June 22, 1938.
    - . . . . . .
- LINDSELL, Harry Percy, Manor House, High Street, Upper Beeding, in the county of Sussex. DRAPER and GENERAL DEALER. Court-BRIGHTON and LEWES (at Brighton).
  - No. of Matter—22 of 1937. Trustee's Name, Address and Description— Orbell, Alfred Everard, 6 and 7, Old Steine, Brighton 1, Accountant and Auditor. Date of Release-June 21, 1938.
- HOLMES, Ruth (Married Woman), The Oakfield and St. Anne's Nursery College, Henbury Court, Henbury, near Bristol.
   Court—BRISTOL.
   No. of Matter—30 of 1936.

  - Trustee's Name, Address and Description-Wheeler, Harold, 26, Baldwin Street, Bristol, Official Receiver. Date of Release-June 23, 1938.
- WHITTARD, Howard Anthony, 127A, Hampton Road, Redland, in the city and county of Bristol. CONFECTIONER and TOBAC-

  - Bristol. CONFECTIONER and TOBAC-CONIST. Court-BRISTOL. No. of Matter-33 of 1937. Trustee's Name, Address and Description-Wheeler, Harold, 26, Baldwin Street, Bristol, Official Receiver.
  - Date of Release-June 23, 1938.
- BULL, Frederick (trading as WEBSTER & CO.), residing at "Sunnyside," Midway, and carry-ing on business as WEBSTER & CO., at High Street, Woodville, both in the county of Derby. GROCER.

  - Court-BURTON-ON-TRENT. No. of Matter-11 of 1937. Trustee's Name, Address
  - Row, Nottingham, Official Receiver. Date of Release—June 23, 1938.
- HAWKSWORTH, Francis Arthur (in the petition described as F. Hawksworth (Male)), III, Northgate Street, in the city of Canterbury, and lately carrying on business at High Street, Harlow, in the county of Essex. FISH SALES-MAN MAN.

- MAN. Court—CANTERBURY. No. of Matter—38 of 1937. Trustee's Name, Address and Description— Wells, Frank Charles, I, The Parade, Canter-bury, Official Receiver. Date of Release—June 22, 1938.

- JARVIS, Harry Edward, 70, Victoria Avenue, Westgate-on-Sea, in the county of Kent, Lorry Driver, lately carrying on business at 74, Station Road, Birchington-on-Sea, in the said county of Kent, as a COAL MERCHANT. Court-CANTERBURY.
- Court-CANTERBURY. No. of Matter-37 of 1937. Trustee's Name, Address and Description-Wells, Frank Charles, 1, The Parade, Canter-bury, Kent, Official Receiver. Date of Release-June 22, 1938.
- STILLMAN, Thomas, residing at 22, Crow Hill, Broadstairs, and carrying on business at 51, Albion Street, Broadstairs, Kent. GREEN-GROCER and FRUITERER.
- · Court-CANTERBURY.
- Court—CANTERDORI. No. of Matter—27 of 1937. Trustee's Name, Address and Description-Lopes, Leonard Dennis, 120, High Stree Broadstairs, Corporate Accountant. Date of Release—June 23, 1938. Street,
- STAFFORTH.
- STAFFORTH, Anthony Harbord, 3, Western Avenue, Gidea Park, in the county of Essex. BUILDER and CONTRACTOR.
  Court—CHELMSFORD.
  No. of Matter—43 of 1935. Trustee's Name, Address and Description— Phillips, Percy, 14, Langham Street, London, W.I, Certified Accountant. Date of Release—June 10, 1028
- Date of Release-June 10, 1938.
- PRESSWOOD, Edwin, residing at 163, Boythorpe Road, Chesterfield, in the county of Derby, carrying on business at 10, Lordsmill Street, Chesterfield aforesaid. CYCLE DEALER. Court-CHESTERFIELD.
- Court-GHESTERFIELD No. of Matter-7 of 1937. Trustee's Name, Address and Description-Rogers, Alfred Joseph. 22, Regent Street, Park Row, Nottingham, Official Receiver. Date of Release-June 23, 1938.

- ROBINSON, Maurice Allan, "Binet," Peaks Hill, Purley, Surrey. FINANCIER.
  Court—CROYDON.
  No. of Matter—67 of 1933.
  Trustee's Name, Address and Description— Chapman, William Albert, 133, Aldersgate Street, E.C.1, Incorporated Accountant.
  Date of Release—June 21, 1938.
- YEATES, Percival William, 36, Lancing Road, Orpington, Kent, carrying on business at 3, Lower Road, St. Mary Cray, Kent. Lower Ro BUTCHER.
- BUICHER. Court—CROYDON. No. of Matter—2 of 1937. Trustee's Name, Address and Description— Waterer, Clarence Roy, 29, Russell Square, London, W.C.1, Official Receiver. Date of Release—June 22, 1938.
- BONFIELD, Robert Arthur Scott, High Street, Roydon, in the county of Essex. COAL MERCHANT. Court—EDMONTON. COAL

  - Cont.-EDMONTON. No. of Matter-25 of 1937. Trustee's Name, Address and Description-Poland, John Langman, 29, Russell Square, London, W.C.1, Official Receiver. Date of Release-June 22, 1938.
- EEVES, Sturton Wakefield, trading as S. W. REEVES & SON, residing and carrying on busi-ness at 26, Highbridge Street, Waltham Abbey, Essex. WIRELESS and CYCLE DEALER.
  Court-EDMONTON.
  No. of Matter-8 of 1937.
  Trustee's Name, Address and Description-Poland, John Langman, 29, Russell Square, London, W.C.1, Official Receiver.
  Date of Release-June 22, 1938. REEVES,

- VERNON, Richard Robert Arthur, 6, Park Road, Forest Hill, London. FILM and STAGE Forest Hill, PRODUCER.

  - PRODUCER. Court-GREENWICH. No. of Matter-7 of 1936. Trustee's Name, Address and Description-Waterer, Clarence Roy, 29, Russell Square, London, W.C.1, Official Receiver. Date of Release-June 23, 1938.
- STEER, William Henry, trading as H. STEER & SON, who resides and carries on business at Birch House, Horsell, in the county of Surrey. BUILDER.

  - BUILDER. Court—GUILDFORD. No. of Matter—21 of 1936. Trustee's Name, Address and Description— Quaife, Albert Edward, 104, High Holborn, London, W.C.1, Incorporated Accountant. Date of Release—June 21, 1938.
- KEIGHLEY, Luther Melancthon, and KEIGHLEY, George, both 81, Beckett Road, Doncaster, carrying on business together in Wheatley Lane, Doncaster, under the style of "HARLAND ROLANDS & CO., FORWARD HATCHERIES" and both lately residing at 51, Kliffen Place, Halifax, all Yorks. POULTRY BREEDERS and MERCHANTS.

  - Court—HALIFAX. No. of Matter—12 of 1937. Trustee's Name, Address and Description— Lund, Walter, Mornington Chambers, 235, Manningham Lane, Bradford, Accountant. Date of Release-June 21, 1938.
- THOMPSON, Cecil Herbert Allfree, now residing at 23. Queens Gate, Kensington, London, S.W.7, formerly of 83. Warwick Road, Earls Court, London, S.W., and Abbeydale Court, Haverstock Hill, Hampstead, Middlesex, and lately residing and carrying on business at Chadwell House, Hertford Road, Ware, in the county of Herts, HOTEL PROPRIETOR (de-scribed in the Receiving Order as of Aberdare, Haverstock Hill, Hampstead, Middlesex, and lately carrying on business at Chadwell House, Hertford Road, Ware aforesaid).
  Court—HERTFORD.
  No. of Matter—4 of 1936.
  Trustee's Name, Address and Description— Hole, John Sidney Bradley, 240, High Holborn, London, W.C.I, Certified Accountant.
  Date of Release—June 10,1938.
- BRAND, James Edgar, residing at 113, Copgrove Road, in the city of Leeds, lately carrying on business at 4A, Dyer Street, Leeds aforesaid, and now carrying on business at 63, George Street, Kirkgate Market, Leeds aforesaid, WHOLESALE FRUIT and POTATO MERCHANT, trading as "BRAND & CO." Court—LEEDS. No. of Matter—26. of 1927

- No. of Matter—26, of 1937. Trustee's Name, Address and Description— Bowling, Harry Clifford, 24, Bond Street, Leeds, Official Receiver. Leeds, Official Receiver. Date of Release—June 22, 1938.
- BRIGGS, Annie (Widow), 45, Luxor View, in the city of Leeds, and carrying on business at 2, Harehills Road, Leeds aforesaid. FURNITURE DEALER.

Court-LEEDS.

- No. of Matter-35 of 1937. Trustee's Name, Address and Description----Bowling, Harry Clifford, 24, Bond Street, Leeds, Official Receiver.
- Date of Release-June 22, 1938.

- REEVE, Leonard Walter, residing and carrying on business at 55A, Barrack Street, in the city of Leeds. BOOT REPAIRER.
  - Court-LEEDS. No. of Matter-51 of 1937.
- Trustee's Name, Address and Description-Bowling, Harry Clifford, 24, Bond Street, Leeds, Official Receiver. Date of Release-June 22, 1938.

.

PROCTOR, Thomas, Castle Inn, Torksey, in the county of Lincoln. LICENSED VICTUALLER and FARMER.

- Lincoln, Official Receiver.
- Date of Release-June 23, 1938.
- WILLOUGHBY, Edna Jessie, Married Woman, The Guest House and Cafe, London Road, East Retford, in the county of Nottingham. GUEST HOUSE and CAFE PROPRIETRESS. Court—LINCOLN and HORNCASTLE. No. of Matter—26 of 1937. Trustee's Name Address and Description—

  - Trustee's Name, Address and Description-Midgley, Edwin Craven, 1, St. Swithin's Square, Lincoln, Official Receiver. Date of Release-June 23, 1938.
- ABRAHAM, William, residing and carrying on busi-ness at Yew Tree Farm, Old Boston, Haydock, nr. St. Helens, in the county of Lancaster. DAIRYMAN.

  - Court—LIVERPOOL. No. of Matter—19 of 1937. Trustee's Name, Address and Description— Airey, John, 8, Victoria Street, Liverpool, Incorporated Accountant.

Date of Release-June 21, 1938.

FROST, Ernest Earl (in the Receiving Order described as E. E. Frost (Male), Little Danes, Harrietsham, in the county of Kent, Builder), Lower Danes, Ashford Road, Harrietsham, in the county of Kent, Builder, formerly residing and carrying on business at "San Remo," Yeomans Lane, Bearsted, in the county of Kent. HAULAGE CONTRACTOR. Court—MAIDSTONE.
No. of Matter—6 of 1937. Trustee's Name, Address and Description—Wells, Frank Charles, 280A, High Street, Rochester, Official Receiver.
Date of Release—June 22, 1938.

Date of Release—June 22, 1938.

- GOLDSMITH, Beatrice Elizabeth, Married Woman, of the Paddocks, Sparsholt, Wantage, in the county of Berks. Court—OXFORD.

No. of Matter-14 of 1935.

- MARSHALL, Edwin John, The Laurels, Beenham, in the county of Berks, and MARSHALL, Charles Edward, The Rose Cottage, Beenham aforesaid, trading together as MARSHALL and SON, of Beenham aforesaid. BUILDERS and EADWERS FARMERS.

- Court—READING. No. of Matter—4 of 1937. Trustee's Name, Address and Description— Simmons, Joseph Bruce, 8, Forbury, Reading, Official Receiver.
- Date of Release-June 22, 1938.

- HAGUE, Herbert, of Storrs Green, Storrs, Stannington, near Sheffield, in the county of York, lately residing and carrying on business at Storrs House Farm, Storrs, Stannington, near Sheffield aforesaid. MILK RETAILER.
  Court—SHEFFIELD.
  No. of Matter—58 of 1937.
  Trustee's Name, Address and Description—Smith, Ernest, Queens Buildings, Queen Street, Sheffield 1, Official Receiver.
  Date of Release—June 23, 1938.

- SHERRIFF, Clarence King, residing at Anderida, St. David's Road, Laindon Hills, in the county of Essex, carrying on business at Manor Road Corner, High Road, Laindon, Essex aforesaid. BUILDER.
  Court—SOUTHEND: No. of Matter—3 of 1937.
  Trustee's Name, Address and Description— Poland, John Langman, 29, Russell Square, London, W.C.I, Official Receiver.
  Date of Release—June 22, 1938.
- HARTLEY, William Foster, residing and carrying on business at Thomas Street, Darlington, Durham. BUILDER and CONTRACTOR.
  Court-STOCKTON-ON-TEES.
  No. of Matter-2 of 1936.
  Trustee's Name, Address and Description-Sansom, Harry, 36, Priestgate, Darlington, Chartered Accountant.
  Data of Belasse-June of 1928.

  - Date of Release-June 21, 1938.
- ROBERTS, William Stephen Hill, residing at Hen-wood Lane, Tettenhall, Wolverhampton, in the county of Stafford, and carrying on business at Temple Chambers, 65, Lichfield Street, Wolverhampton aforesaid. INCORPORATED ACCOUNTANT.

  - ACCOUNTANT. Court—WOLVERHAMPTON. No. of Matter—26 of 1936. Trustee's Name, Address and Description— Cripwell, Albert, 12, Cherry Street, Birming-ham, Incorporated Accountant. Date of Release—June 21, 1938.
- WILSON, John Vernon, residing at Dust House Lane, Tardebigge, near Bromsgrove, Worcester-shire, and carrying on business at 20, High Street, Bromsgrove aforesaid. ELECTRICAL CONTRACTOR.

  - CONTRACTOR. Court—WORCESTER. No. of Matter—6 of 1936. Trustee's Name, Address and Description— Bendall, Frederick Ernest, Midland Chambers, Warwick Passage, Corporation Street, Birming-ham 2, Certified Accountant. Date of Release—June 21, 1938.

# INTENDED DIVIDENDS.

CLARK, Claude Stanley, residing at 35, The Embankment, Bedford, Bedfordshire. COM-PANY DIRECTOR. Court-BEDFORD. No. of Matter-1 of 1938. Last Day for Receiving Proofs-July 21, 1938. Name of Trustee and Address-Salaman, Frederick Source and Address-Salaman, Frederick

- Seymour, 1-2, Bucklersbury, London, E.C.4.
- YO, Reginald Burton, 14, Waterford Road, Birkenhead, in the county of Chester, and also residing in apartments at 52, Dawson Road, Kingston-on-Thames, in the county of Surrey. Departmental Store Official, formerly BANK MAYO, MANAGER.

Court-BIRKENHEAD

No. of Matter—12 of 1937. Last Day for Receiving Proofs—July 19, 1938. Name of Trustee and Address—Williams, Fran Lloyd, 11-13, Victoria Street, Liverpook. Frank WOODHALL, Herbert Charnley, "Quarme," Signal Hayes Road, Walmley, Sutton Coldfield, in the county of Warwick, lately residing at 30, Cecil Road, Erdington, Birmingham, in the county of Warwick, and carrying on business at 20, Villa Road, Handsworth, Birmingham aforesaid. ADVERTISING AGENT.
Court—BIRMINGHAM.
No. of Matter—46 of 1932.
Last Day for Receiving Proofs—July 19, 1938.
Name of Trustee and Address—Clark, Rudolf Kynoch, Empire House, 159, Great Charles Street, Birmingham, Official Receiver.

- GILLINGHAM, Emma Winter, Wife of William Leonard Kay Gillingham, HAZELDINE, Thomas Henry, Waiter, and HAZELDINE, Dorothy Mabel, Wife of Thomas Henry Hazeldine, all of "Kerri," Seacombe Road, Sandbanks, in the county of Dorset, and lately carrying on business in partnership at Bay View Stores, Banks Road, Sandbanks aforesaid, under the name or style of "GILLINGHAM AND HAZELDINE." GROCERS and PROVISION MERCHANTS. (Joint Estate.) Court—BOURNEMOUTH. No. of Matter—32 of 1937. Last Day for Receiving Proofs—July 21, 1938. Name of Trustee and Address—Medcalf, Alexander Lauder, 10, Rockstone Place, Southampton, Official Receiver.

- NASH, Ernest, residing at 98, Sterte Road, Poole, and carrying on business at 25, Hill Street, Poole, both in the county of Dorset. GROCER. Court—BOURNEMOUTH.
  No. of Matter—2 of 1938.
  Last Day for Receiving Proofs—July 19, 1938.
  Name of Trustee and Address—Medcalf, Alexander Lauder, 10, Rockstone Place, Southampton, Official Receiver.

- ARMITAGE, David, residing at 1, Twining Avenue, Twickenham, Middlesex, and BOWMAN, Twickenham, Middlesex, and BOWMAN, Horace, residing at 45, Clonmel Road, Tedding-ton, Middlesex, carrying on business under the style or firm of ARMITAGE & BOWMAN, BUILDERS and DECORATORS, at 1, Twining

- BUILDERS and DECORATORS, at 1, Twining Avenue, Twickenham aforesaid. Court-BRENTFORD. No. of Matter-35 of 1937. Last Day for Receiving Proofs-July 20, 1938. Name of Trustee and Address-Poland, John Langman, 29, Russell Square, London, W.C.1, Official Receiver Official Receiver.
- OCKE, Roy Francis, trading as The BALLAST AND SAND SUPPLIES CO., Council Yard, Petersfield Road, Acton, W.3, Middlesex, and lately residing at 45, Cumberland Park, Acton aforesaid. SAND and BALLAST MERCHANT. Court—BRENTFORD.
  No. of Matter—43 of 1937.
  Last Day for Receiving Proofs—July 20, 1938.
  Name of Trustee and Address—Poland, John Langman, 29, Russell Square, London, W.C.I, Official Receiver. LOCKE

- YOUNG, Constance Frances Morton, Flat 2, 130, Marine Parade, Brighton, Sussex. Widow.
  Court-BRIGHTON and LEWES (at Brighton).
  No. of Matter—10 of 1938.
  Last Day for Receiving Proofs—July 22, 1938.
  Name of Trustee and Address—James, Ernest, St. Lawrence House, 1, Trump Street, King Street, London E.C.2

  - London, E.C.2.
- KAUFMAN, Marie Vital (described in the Receiving Order as V. Kaufman, Married Woman), trading as "SUSANS," 1385, London Road, Norbury, in the county of Surrey. COSTUMIER. Court—CROYDON.

- No. of Matter—1 of 1938. Last Day for Receiving Proofs—July 19, 1938. Name of Trustee and Address—Waterer, Clarence Roy, 29, Russell Square, London, W.C.1, Official Receiver.
- WOLF, Marguerite Russell (Spinster), described in the Receiving Order as Margaret Wolf (Spinster), 110, North End, Croydon, in the county of
- 110, Ito. Surrey. Court—CROYDON. No. of Matter—66 of 1937. Last Day for Receiving Proofs—July 19, 1938. Name of Trustee and Address—Waterer, Clarence Roy, 29, Russell Square, London, W.C.1, Official Receiver.

- HARRISON, Frederick James, Hill Top Farm, Stanley, in the county of Derby. FARMER.
  Court—DERBY AND LONG EATON.
  No. of Matter—39 of 1937.
  Last Day for Receiving Proofs—July 20, 1938.
  Name of Trustee and Address—Rogers, Alfred Joseph, 22, Regent Street, Park Row, Notting-ham, Official Receiver.
- KEETLEY, Mary, now 6, Sunny Grove, Chaddesden, and formerly 26, Market Street, Heanor, both in the county of Derby. (Wife of Arthur Harry Keetley.) DRAPER.
  Court—DERBY AND LONG EATON.
  No. of Matter—31 of 1937.
  Last Day for Receiving Proofs—July 20, 1938..
  Name of Trustee and Address—Rogers. Alfred Joseph, 22, Regent Street, Park Row, Notting-ham, Official Receiver.
- KING, Harold James, residing and carrying on business at 18, Derby Road, Sandiacre, in the county of Derby. BAKER and CONFECTIONER.
  Court—DERBY AND LONG EATON.
  No. of Matter—8 of 1938.
  Last Day for Receiving Proofs—July 20, 1938.
  Name of Trustee and Address—Rogers, Alfred Joseph, 22, Regent Street, Park Row, Nottingham Official Receiver

  - Joseph, 22, Regent Str ham, Official Receiver.
- CEY, Albert, 159, Northgate Street, Great Yarmouth, Norfolk, carrying on business under the name of "STACEY'S." GENERAL STACEY, Albert, DRAPER.

  - Court-GREAT YARMOUTH. No. of Matter-2 of 1938. Last Day for Receiving Proofs-July 19, 1938. Name of Trustee and Address-Walter, F. R. D., Castle Chambers, Opie Street, Norwich, Official Receiver Receiver.
- IREDALE, James Henry, 20, Green Park Road, Halifax, Yorks, and PRIESTLEY, Edith (Spinster), 10, Cromwell Terrace, Halifax, trading in co-partnership as F. YATES & CO., 43A, Stannary Street, Halifax. BLOUSE and GOWN MANUFACTURERS. (Joint Estate.) Court—HALIFAX.
  No. of Matter—25 of 1937.
  Last Day for Receiving Proofs—July 22, 1938.
  Name of Trustee and Address—Horrocks-Taylor, Frank, National Provincial Bank Chambers, Halifax.

  - Halifax.
- PRIESTLEY, Edith. (Separate Estate.) Court—HALIFAX. No. of Matter—25 of 1937. Last Day for Receiving Proofs—July 22, 1938. Name of Trustee and Address—Horrocks-Taylor, Frank, National Provincial Bank Chambers, Halifar Halifax. . .

VATERHOUSE, William Evelyn, trading as WATERHOUSE & CO., Lidgett Lane Cabinet Works, Roundhay, in the city of Leeds, JOINER and CABINET MAKER, and resid-ing at 10, Orchard Grove, Crossgates, in the county of York.
Court-LEEDS.
No. of Matter-I of 1924.
Last Day for Receiving Proofs-July 20, 1938.
Name of Trustee and Address-Bowling, Harry Clifford, 24, Bond Street, Leeds 1, Official Receiver. WATERHOUSE.

- FELDMAN, Isidore, 48, Heywood Drive, Luton, in the county of Bedford, lately residing at 53, Wellington Street, Luton aforesaid, and carrying on business at 53, Wellington Street, Luton aforesaid, and at 41, Mill Street, Luton aforesaid. WATCHMAKER and JEWELLER. Court—LUTON.
  No. of Matter—7 of 1938.
  Last Day for Receiving Proofs—July 19, 1938.
  Name of Trustee and Address—Bengough, Thomas, 6. The Parade. Northampton. Official Receiver.

- 6, The Parade, Northampton, Official Receiver.
- JONES, John Rhys, residing at 76, Dunraven Street, Tonypandy, Glam., and lately carrying on busi-ness at 76, Dunraven Street, and 94, Dunraven Street, both in Tonypandy, Glam. GENT'S OUTFITTER.
  - Court-PONTYPRIDD, YSTRADYFODWG and PORTH.

  - No. of Matter—9 of 1937. Last Day for Receiving Proofs—July 20, 1938. Name of Trustee and Address—Betts, Reginald, 34, Park Place, Cardiff, Official Receiver.
- TUNGATE, John Harold, Bredgar Cottage, All-hallows, near Rochester, Kent. GENERAL hallows, near R SHOPKEEPER.

- Court—ROCHESTER. No. of Matter—27 of 1937. Last Day for Receiving Proofs—July 21, 1938. Name of Trustee and Address—Mills, Walter Thomas, 95-7, High Street, Chatham.
- LINSELL, George William (described in the Receiv-ing Order as G. W. Linsell), Tippendale Lane, Park Street, near St. Albans, in the county of Park Street, hear St. Albans, in the county of Hertford, and lately carrying on business at 26, London Road, St. Albans aforesaid. RADIO and ELECTRICAL DEALER.
  Court—ST. ALBANS.
  No. of Matter—6 of 1937.
  Last Day for Receiving Proofs—July 19, 1938.
  Name of Trustee and Address—Bendall, Frederick Exerct and Address—Bendall, Frederick

  - Ernest, 4, Charterhouse Square, London, E.C.I.
- MATHER, Joseph George, residing at 95, Yarburgh Street, Moss Side, in the city of Manchester, and carrying on business at 87, Great Western Street, Moss Side aforesaid, as a RETAIL TIMBER MERCHANT, under the style of "MOSS SIDE SAWMILLS." Court-SALFORD.

  - No. of Matter—r of 1938. Last Day for Receiving Proofs—July 20, 1938. Name of Trustee and Address—Langmaid, Frederick Harold, Byrom Street, Manchester 3, Spring Official Barciana Senior Official Receiver.
- SELBY, Alexander Prideaux, deceased, lately a STOCKBROKER; and member of the firm of Selby & Co., Stockbrokers, carrying on business at Kings Chambers, Sheffield, in the county of Verity York. Court-SHEFFIELD.

- No. of Matter—121 of 1932. Last Day for Receiving Proofs—July 20, 1938. Name of Trustee and Address—Salaman, Frederick Seymour, 1-2, Bucklersbury, London, E.C.4.
- BLACKSHAW, Charles Arthur, residing at 36, Belmont Road, Penn, Wolverhampton, in the county of Stafford, and carrying on business at Napier Road, Wolverhampton. BUILDER. Court-WOLVERHAMPTON.

  - Court—WOLVERTIANT FORM No. of Matter—61 of 1923. Last Day for Receiving Proofs—July 19, 1938. Name of Trustee and Address—Clark, Rudolf Kynoch, Empire House, 159, Great Charles Street, Birmingham, Official Receiver.

#### DIVIDENDS.

- DENNIS, George Walton Tyndall, lately trading as G. WALTON, and described in the Receiving Order as George Walton Dennis, sued as George Dennis (Male), 46 Alric Avenue, Malden, Surrey, FRUITERER, and lately carrying on business at 19, Fouberts Place, Regent Street, 25, York Road, Lambeth, and The Subway, Waterloo Station, all London.

  - Waterioo Station, all London. Court—HIGH COURT OF JUSTICE. No of Matter—1,006 of 1935. Amount per £—38. 4<sup>1</sup>/<sub>4</sub>d. First or Final, or otherwise—Supplemental. When Payable—Any day (except Saturday), between the hours of 11 and 2. Where Payable—Bankruptcy Buildings, Carey Street London, W C 2
  - Street, London, W.C.2.
- FUDGE, Frederick James, residing at 49, Webber Street, Blackfriars, London, S.E.1, described in Street, Blackfriars, London, S.E.I, described in the Receiving Order and lately carrying on busi-ness under the style of and sole proprietor of F. FUDGE & CO. (a firm), GENERAL EN-GINEERS and MILLWRIGHTS, 122, New Kent Road, London, S.E.I. Court—HIGH COURT OF JUSTICE. No. of Matter—669 of 1937. Amount per  $f_{-}$ -4d. First or Final, or otherwise—First and Final. When Payable—Any day (except Saturday), between the hours of 11 and 2. Where Payable—Bankruptcy Buildings, Carey Street, London, W.C.2.
- HEATHFIELD, Daniel, "Kinross," Coombe Lane, Malden, Surrey, and carrying on business at 200A. Coldharbour Lane, Camberwell, London, S.E.5. WINE and SPIRIT MERCHANT. Court—HIGH COURT OF JUSTICE. No. of Matter—580 of 1937. Amount per £—18. 7<sup>2</sup>4d.
  First or Final, or otherwise—Second and Final. When Payable—July 19, 1938.
  Where Payable—I4, Langham Street, Portland Place, London, W.I.
- YOUNG, Trevelyan Hartland, residing at 43, Balmoral Road, Accrington, in the county of Lancaster, and carrying on business at I and 2, Post Office Arcade, Church Street, Accrington aforesaid. BOOT and SHOE RETAILER. Court—BLACKBURN and CLITHEROE. No. of Matter—4 of 1008

  - Court-BLACKBURN and CLITHEROE. No. of Matter-4 of 1938. Amount per £-61d. First or Final, or otherwise-First and Final. When Payable-July 12, 1938. Where Payable-The Official Receiver's Office, 7, Lord Street West, Blackburn.

.

- PENNINGTON, Cyril, 71, Lumley Avenue, Skegness, Lincolnshire. BUILDER and CONTRACTOR.HICKS, William Atkins, Bacton Road, North<br/>Walsham, Norfolk. BUILDER.<br/>Court—BOSTON.No. of Matter—14 of 1937.No. of Matter—14 of 1937.Amount per f—2s. 5gd.First or Final, or otherwise—First and Final.First or Final, or otherwise—First and Final.When Payable—July 11, 1938.Where Payable—Old Post Office Chambers, Skegness.Where Payable—Old Post Office Chambers, Skegness.

.

- TOR. Court—BOSTON. No. of Matter—14 of 1937. Amount per f.—2s. 5kd. First or Final, or otherwise—First and Final. When Payable—July 11, 1938. Where Payable—Old Post Office Chambers, Skeg-

.

- ness.
- ROBINSON, Harry, residing and carrying on busi-ness at 240, Leeds Road, Nelson, in the county of Lancaster. HOUSEHOLD STORES DEALER. of Lanc DEALER.

  - DEALER. Court—BURNLEY. No. of Matter—17 of 1937. Amount per 4—9s. 3d. First or Final, or otherwise—First and Final. When Payable—July 13, 1938. Where Payable—The Official Receiver's Office, 7, Lord Street West, Blackburn.
- JONES, John George, residing and carrying on business at 13, Carlisle Street, Cardiff. GROCER.
  Court—CARDIFF and BARRY.
  No. of Matter—12 of 1937.
  Amount per f.—13. 6d.
  First or Final, or otherwise—Second and Final.
  When Payable—July 11, 1938.
  Where Payable—At the offices of E. T. Collins & Son. 28 Baldwin Street Briefol

- Son, 28, Baldwin Street, Bristol.
- ILLING, Arthur Henry Gwynne, residing and carrying on business at "Bank House Farm," Kelsall, in the county of Chester. HAULAGE CONTRACTOR and SMALLHOLDER.
  Court—CHESTER.
  No. of Matter—6 of 1932.
  Amount per 4-8d.
  First or Final, or otherwise—Fifth.
  When Payable—July 15, 1938.
  Where Payable—Official Receiver's Office, St. Peter's Churchward. The Cross. Chester. TILLING,

- Peter's Churchyard, The Cross, Chester.
- BEL, Charlie, trading as NON BETTA PRODUCTS, 150, Broughton Street, Cheetham, Manchester, in the county of Lancaster, WHOLESALE CONFECTIONER, lately carry-ing on business at 216, City Road, Hulme, Manchester aforesaid.
  Court—MANCHESTER.
  No. of Matter—24 of 1937.
  Amount per f.—10d.
  First or Final, or otherwise—First and Final.
  When Payable—July 11, 1938.
  Where Payable—15, Eastcheap, London, E.C.3. ABEL
- COHEN, Maurice Leonard Harry, otherwise known as "L. H. Cowan," 11, Wilderton Road, Stamford Hill, London, lately residing at 156, West Road, and carrying on business at 6, Handysides Buildings, Percy Street, 5, Benton Road, 158, West Road, and 34, Percy Street, all in the city and county of Newcastle-upon-Tyne, Library Proprietor and Tobacconist, and formerly carrying on business under the name of "CHARLES COLLINS," at 19, Lawton Street, Westmorland Road, in the said city and county of Newcastle-upon-Tyne, as a TAXI CAB PROPRIETOR.
  Court—NEWCASTLE-UPON-TYNE. No. of Matter—31 of 1937.

  - Court—NEWCASILE-OFON-ITTE. No. of Matter—31 of 1937. Amount per  $\xi$ —8s. 4d. First or Final, or otherwise—First and Final. When Payable—July 18, 1938. Where Payable—Trustee's Offices, 72, Grey Street, Newcastle-upon-Tyne I.

.

- BODFIELD, Lewis Henry, 189, Old Road East, Gravesend, in the county of Kent, and BODFIELD, Arthur William, residing at 22, Park Place, Gravesend, in the county of Kent, carrying on business in co-partnership under the firm name or style of OAK HAULAGE CO., at Bath Street, Gravesend aforesaid. HAULAGE CONTRACTORS and BUILDERS' MERCHANTS. Court—ROCHESTER. No. of Matter—7 of 1938 (under Order for Con-solidation of Proceedings). Amount per  $f_{--5}^{+}d$ . First or Final, or otherwise—First and Final. When Payable—July 12 1938. Where Payable—The Official Receiver's Office, 280A, High Street, Rochester.
- HOUGHTON, Albert Edward (deceased), Dykes Lane, in the city of York. CI COMPANY BRANCH MANAGER. Court-YORK. ed), 5, Green CLOTHING

  - Court—YORK. No. of Matter—5 of 1938. Amount per £—58. 9d. First or Final, or otherwise—First and Final. When Payable—July 20, 1938. Where Payable—Official Receiver's Office, Lantern Tower Chambers, Coppergate, York.

Pursuant to the Act and Rules, notices to the above effect have been received by the Board of Trade.

> D. WILLIAMS, Inspector-General in Bankruptcy.

# THE COMPANIES ACTS.

#### FIRST MEETINGS.

- Name of Company-CAR SALES (CHISWICK) Limited.

- Address of Registered Office—51, Sutton Lane, Chiswick, London, W.4. Court—HIGH COURT OF JUSTICE. No. of Matter—00371 of 1938. Creditors—Date, July 15, 1938; Hour, 2.30 p.m.; Place, 33, Carey Street, Lincoln's Inn, London, WC.2. W.C.2.
- Contributories—Date, July 15, 1938; Hour, 3 p.m.; Place, 33, Carey Street, Lincoln's Inn, London, W.C.2.

- Name of Company-PLAN Limited. Address of Registered Office-Carlton House, Regent Street, London, S.W.I. Court-HIGH COURT OF JUSTICE. No. of Matter-00368 of 1938. Creditors-Date, July 13, 1938; Hour, 11.30 a.m.; Place, 33, Carey Street, Lincoln's Inn, London, W.C.2.
  - Contributories—Date, July 13, 1938; Hour, 12 noon; Place, 33, Carey Street, Lincoln's Inn, London, W.C.2.

#### GENERAL MEETINGS OF CREDITORS AND CONTRIBUTORIES.

ame of Company—METROPOLITAN CHAIN STORES Limited. Address of Registered Office—134, Cheapside, in the city of London, E.C.2. Court—HIGH COURT OF JUSTICE. Name

- Contributories—Date, July 15, 1938; Hour, 11.30 a.m.; Place, The Drawing Room, Connaught Rooms, Great Queen Street, London, W.C.2. Contributories—Date, July 15, 1938; Hour, 12.15 p.m.; Place, The Drawing Room, Conneught Rooms, Great Queen Street London
- Connaught Rooms, Great Queen Street, London, W.C.2.

#### INTENDED DIVIDEND.

- Name

- Jame of Company—TAYLOR, AND COMPANY (BOXES) Limited.
  Address of Registered Office—32-36, Rathbone Street, Liverpool.
  Court—LIVERPOOL.
  No. of Matter—5 of 1937.
  Last Day for Receiving Proofs—July 20, 1938.
  Names and Addresses of Joint Liquidators—Arthur Tyldesley Eaves, 47, Mosley Street, Man-chester, and Lawrence Bailey, 8, Harrington Street, Liverpool. chester, and Lav Street, Liverpool.

#### DIVIDEND.

- Name of Company-PANY Limited. -CAVENDISH SPINNING COM-

  - PANY Limited. Address of Registered Office—Cavendish Mill, Cavendish Street, Ashton-under-Lyne. Court—CHANCERY OF THE COUNTY PALA-TINE OF LANCASTER, Manchester District. No. of Matter—1934 Letter C. No. 119. Amount per  $\pounds$ —3d. First and Final, or otherwise—Supplementary. When Payable—July 13, 1938. Where Payable—At the office of the Liquidator, 24, Clegg Street, Oldham.

Pursuant to the Companies (Winding-up) Act, 1890, and the Companies (Consolidation) Act, 1908, and the Companies Act, 1929, and the Rules thereunder, notices to the above effect have been received by the Board of Trade.

. •

۰.

.

E. H. MARKER, Comptroller of the Companies Department.

.;

•• 1. . . .

.:

. :

· · · · · · . . . . 1203

3

.. [

٠:

••

# THE LONDON GAZETTE, 5 JULY, 1938

# NOTICE—All Notices and Advertisements are published in The London Gazette at the risk of the Advertiser.

All Notices and Advertisements by Private Advertisers may be tendered at or sent direct by post to the Office of The London Gazette, H.M. Stationery Office, Princes Street, Westminster, S.W.I, for insertion at the authorised rates of payment. The office hours are from 10 a.m. to 5 p.m., closing at one o'clock on Saturdays.

All Notices and Advertisements must be prepaid. To save delay, Notices and Advertisements sent direct by post should be accompanied by Postal Orders, made payable to The London Gazette. Notices and Advertisements paid for by cheque will not be inserted until such cheques have been cleared.

Notices of Dissolution of Partnership will not be inserted unless signed by the Partners named therein, or by their legal representatives; and the signature or representative character of the signatory must be verified by Statutory Declaration made by a Solicitor of the Supreme Court.

A Notice of Dissolution of Partnership not signed by all the Partners, or their legal representatives, must be accompanied by a Statutory Declaration made by a Solicitor of the Supreme Court to the effect that such notice is given in pursuance of the terms of the partnership to which it relates.

Advertisements purporting to be issued in pursuance of Statutes or under Orders of Court will not be inserted unless signed or attested by a Solicitor of the Supreme Court.

Advertisements relating to Bills before Parliament will not be inserted unless signed by a Parliamentary Agent or a Solicitor of the Supreme Court.

Authorised Scale of Charges for Notices and Advertisements.

(a) Notices under the Bankruptcy Act 1914 and 1926-7s. 6d.

(b) Notices under the Companies (Winding-up) Act, 1890, the Companies (Consolidation) Act, 1908, and the Companies Act, 1929, as prescribed by the Rules under the Acts-75. 6d. Other Companies Winding-up Notices at the rates given under (e).

(c) Notices under the County Courts Act, 1934, when received from the Registrar of County Court Judgments—15s.

(d) Friendly Societies Notices-7s. 6d.

(e) All other Notices or Advertisements, including Applications to Parliament and Scotch Sequestrations, will be charged by the number of lines appearing as plain matter in the type of the Gazette :---

If not exceeding 10 lines of printed matter—15s. For each additional 5 lines or under—7s. 6d.

Table or tabular matter will be charged at the rate of £1 10s. per quarter page or part thereof.

(f) In Notices of Dissolution of Partnership, the signatures of the partners will not be charged for.

The London Gazette is published every Tuesday and Friday.

All Notices and Advertisements should reach the Office of The London Gazette, H.M. Stationery Office, Princes Street, Westminster, S.W.I, before 2 p.m. on the day previous to publication. Notices and Advertisements received after that time will be inserted, if circumstances permit, on payment of a late fee for each Notice or Advertisement at the following rates :--

Up to 5 p.m. on the day previous to put	olicatio	n	 •••	5s.
Up to noon on the day of publication			•••	IOS.
Up to 2 p.m. on the day of publication	,	•••	 	20S.

All communications on the business of The London Gazette should be addressed to The London Gazette, H.M. Stationery Office, Princes Street, Westminster, S.W.I.

# THE LONDON GAZETTE, 5 JULY, 1938

#### SUBSCRIPTION TERMS FOR PARLIAMENTARY PUBLICATIONS

#### House of Lords Papers, &c.

Minutes of Proceedings. £2 (£2 15s.). Bills. £3 (£4). Papers and Bills, and Command Papers. £11 (£12 5s.). Complete Set, excluding Debates. £12 10s. (£14).

## House of Commons Papers, &c.

Votes and Proceedings. £8 10s. (£9 10s.). Bills. £2 10s. (£3 10s.). Reports and Papers, Bills and Command Papers. £17 (£18 5s.). Trade and Navigation Returns. (£2 17s.). Complete Set, excluding Debates. £24 (£25 10s.).

#### Papers, &c., of both Houses

Minutes of Lords and Votes and Proceedings of Commons. £10 10s. (£11 15s.). Bills of both Houses. £5 10s. (£6 15s.).

Papers and Bills of both Houses, and Command Papers, excluding Minutes and Votes. £21 (£22 IOS.).

Papers of both Houses and Command Papers (excluding Minutes, Votes and Bills). £16 (£17 5s.).

Complete Set of all Papers, &c., excluding Debates. £28 (£30).

Complete Set of all Papers, &c., except Bills and Debates. £24 (£25 10s.).

#### Parliamentary Debates

House of Lords, daily parts 6d. each.  $f_2$  ( $f_2$  15s.). House of Commons, daily parts 6d. each.  $f_2$  10s. ( $f_3$  10s.). Standing orders can be accepted for bound volumes of the Debates.

> Delivery will be made within a radius of three miles from Charing Cross at the prices shown in brackets, which also cover postage charges inland and abroad.

## LONDON AND EDINBURGH GAZETTES

# London Gazette

Tuesdays and Fridays. Single numbers 2s. (Annual Subscription post free  $f_{10}$  10s.).

Delivery by hand on evening of publication  $f_{12}$  12s.

#### Edinburgh Gazette

Tuesdays and Fridays (Published and obtainable at the Exchequer Chambers, Parliament Square, Edinburgh). Single numbers 1s. 6d. Annual Subscription £6. (These rates include postage).

The above rates are strictly net and payable in advance. Orders should be sent to The Director of Publications, H.M. Stationery Office, Adastral House, Kingsway, London, W.C.2.

# **The London Gazette**

# TUESDAY, 5 JULY, 1938

. **.** .

# CONTENTS

PAGE	PAGE
State Intelligence 4325	Boroid Ltd.—Notice 4376
Post Office—Parcel Post Notice 4337	H.M. Land Registry-Notice 4376
Public Health Act, 1936-Notices 4337	South Yorkshire Mines Drainage Com-
Leeds Corporation Light Railways (Ex-	mittee—Notice 4376
tension) Order, 1935—Notiec 4337	Queen Anne's Bounty—Notice 4376
Road Traffic Act, 1930-Notices 4337	The National Bank, Ltd.—Notice 4376
The Bankruptcy Acts, 1914 and 1926-	Deeds of Arrangement Act-Notices 4376
Returns 4338	Bankruptcy Acts, 1914 and 1926—
British Corn — Statements showing Quarterly and Weekly Returns 4340	Notices 4377
Bullion and Specie-Notice 4342	Panlameters Acta Tot and Tooh
Cereals, Average Prices of 4344	Bankruptcy Acts, 1914 and 1926—
H.M. Land Registry 4346	Receiving Orders 4378
Diseases of Animals Acts, 1894 & 1936 Returns 4347	First Meetings and Public Examina- tions 4380
Places Registered for Solemnizing	Adjudications 4384
Marriages-Notices 4351	Applications for Discharge 4385
Companies (Consolidation) Act, 1908, and the Companies Act, 1929—Notices 4351	Orders made on Applications for Discharge 4386
Partnerships Dissolved 4361	Release of Trustees 4386
Trustee Act, 1925, Section 27-Notices to Creditors 4363	Intended Dividends 4389
Change of Name by Deed—Notices 4374	Dividends Declared 4391
Auction Sale by Order of the Court— Notices 4375	Companies Acts
Chancery Division of the High Court of	First Meetings 4392
Justice—Notice 4375	Intended Dividend 4393
Brazilian Warrant Agency and Finance Co., LtdNotice 4375	Dividend Declared 4393

•**,** • •

LONDON PRINTED AND PUBLISHED BY HIS MAJESTY'S STATIONERY OFFICE To be purchased directly from H.M. STATIONERY OFFICE at the following addresses: Adastral House, Kingsway, London, W.C.2; 120 George Street, Edinburgh 2; 26 York Street, Manchester 1; 1 St. Andrew's Crescent, Cardiff; 80. Chichester Street, Belfast; or through any bookseller

2.5

::

1938

Price Two Shillings net